SECURITIES AND EXCHANGE COMMISSION

FORM SC 13D

Schedule filed to report acquisition of beneficial ownership of 5% or more of a class of equity securities

Filing Date: **2017-01-03 SEC Accession No.** 0001144204-16-141978

(HTML Version on secdatabase.com)

SUBJECT COMPANY

Dipexium Pharmaceuticals, Inc.

CIK:1497504| IRS No.: 271707962 | State of Incorp.:DE | Fiscal Year End: 1231

Type: SC 13D | Act: 34 | File No.: 005-88072 | Film No.: 162078210

SIC: 2834 Pharmaceutical preparations

Mailing Address 14 WALL ST. SUITE 3D NEW YORK NY 10005 Business Address 14 WALL ST. SUITE 3D NEW YORK NY 10005 212-269-2834

FILED BY

PLX PHARMA INC.

CIK:1291520| IRS No.: 000000000 | State of Incorp.:DE | Fiscal Year End: 1231

Type: SC 13D

SIC: 2834 Pharmaceutical preparations

Mailing Address 8285 EL RIO STREET SUITE 130 HOUSTON TX 77054

Business Address 8285 EL RIO STREET SUITE 130 HOUSTON TX 77054 713-842-1249

SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

SCHEDULE 13D

(Rule 13d-101)

INFORMATION TO BE INCLUDED IN STATEMENTS FILED PURSUANT TO RULE 13d-1(a) AND AMENDMENTS THERETO FILED PURSUANT TO RULE 13d-2(a)

Dipexium Pharmaceuticals, Inc.

(Name of Issuer)

Common Stock

(Title of Class of Securities)

25456J104

(CUSIP Number)

PLx Pharma Inc. 8285 El Rio St., Suite 130

Houston, Texas 77054 713.842.1249

With a copy to: Michael L. Laussade Jackson Walker L.L.P. 2323 Ross Ave. Suite 600 Dallas, TX 75201 214.953.5805

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

December 22, 2016

(Date of Event which Requires Filing of this Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition which is the subject of this Schedule 13D, and is filing this schedule because of $\S\S$ 240.13d-1(e), 240.13d-1(f) or 240.13d-1(g), check the following box \square .

The information required on the remainder of this cover page shall not be deemed to be "filed" for the purpose of Section 18 of the Securities Exchange Act of 1934 (the "Act") or otherwise subject to the liabilities of that section of the Act but shall be subject to all other provisions of the Act (however, see the Notes).

Cusip No. 25456J104 Page 2 of 8 Pages

1	NAMES OF REPORTING PERSONS						
_		PLx Pharma Inc.					
2	CHECK THE APPROPRIATE BOX IF A MEMBER OF A GROUP			(A) □ (B) ♣			
3	SEC USE ONLY						
4	SOURCE OF FUNDS OO						
5	CHECK IF DISCLOSURE OF LEGAL PROCEEDINGS IS REQUIRED PURSUANT TO ITEMS 2(D) OR 2(E) □						
6	CITIZENSHIP OR PLACE OF ORGANIZATION DE						
NUMBER OF SHARES BENEFICIALLY OWNED BY EACH REPORTING PERSON WITH		7	SOLE VOTING POWER 0				
		8	SHARED VOTING POWER 4,414,270 (see Introduction and Item 5)				
		9	SOLE DISPOSITIVE POWER 0				
		10	SHARED DISPOSITIVE POWER 4,414,270 (see Introduction and Item 5)				
11	AGGREGATE AMOUNT BENEFICIALLY OWNED BY EACH REPORTING PERSON 4,414,270 (see Introduction and Item 5)						
12	CHECK BOX IF THE AGGREGATE AMOUNT IN ROW (11) EXCLUDES CERTAIN SHARES □						
13	PERCENT OF CLASS REPRESENTED BY AMOUNT IN ROW (11)						
	39.63%*						
14	TYPE OF REPORTING PERSON						
	СО						

^{*} Based on 10,401,122 shares of common stock of the issuer reported outstanding on the issuer's Quarterly Report on Form 10-Q for the period ended September 30, 2016.

Cusip No. 25456J104 Page 3 of 8 Pages

INTRODUCTORY NOTE

This statement on Schedule 13D (this "Statement") relates to the Agreement and Plan of Merger and Reorganization, dated as of December 22, 2016 (the "Merger Agreement"), by and among PLx Pharma Inc., a Delaware corporation ("PLx"), Dipexium Acquisition Corp., Inc., a Delaware corporation ("AcquireCo"), and Dipexium Pharmaceuticals, Inc., a Delaware corporation (the "Company"), and the transactions contemplated thereby. The Merger Agreement contemplates that, subject to the terms and conditions of the Merger Agreement, AcquireCo will be merged with and into PLx (the "Merger"), with PLx continuing after the Merger as the surviving corporation. Pursuant to the Merger Agreement, at the effective time of the Merger, each issued and outstanding share of PLx's common stock, par value \$0.001 per share will be converted into the right to receive shares (the "Shares") of the Company's common stock, par value \$0.001 per share (the "Common Stock"), subject to an equity exchange ratio determined and adjusted in accordance with the Merger Agreement.

David Luci and Robert DeLuccia (together, the "*Shareholders*") are affiliates of the Company who beneficially own 2,208,535 Shares and 2,205,735 Shares, respectively, for a combined total of 4,414,270 Shares or approximately 39.63% of the outstanding Common Stock. In connection with the execution of the Merger Agreement, each of Mr. Luci and Mr. DeLuccia (as well as certain trusts controlled by each of them) entered into a separate voting agreement with PLx (the "*Voting Agreements*"), pursuant to which they have agreed (i) to grant to PLx an irrevocable proxy to vote their Shares in favor of the Merger and the Merger Agreement and against any other acquisition proposals, and (ii) to certain restrictions on the transfer of their Shares.

The description of the Merger Agreement and the Voting Agreements is qualified in its entirety by the terms and conditions of the Merger Agreement and the Voting Agreements, which are filed as Exhibits 99.1 through 99.5 hereto, and are incorporated herein by reference.

ITEM 1. SECURITY AND ISSUER.

The class of equity security to which this statement relates is the Common Stock of the Company. The address of the principal executive offices of the Company is 14 Wall St., Suite 3D, New York, NY.

ITEM 2. IDENTITY AND BACKGROUND.

(a) - (c), (f) This statement is being filed by PLx.

PLx is a corporation formed under the laws of the State of Delaware, and is a late-stage pharmaceutical company. The principal business and office address for PLx is 8285 El Rio St., Suite 130, Houston, Texas 77054.

(d) – (e) During the last five years, PLx has not, and to the best knowledge of PLx, none of the persons listed on Schedule A hereto has, been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors) or was a party to a civil proceeding of a judicial or administrative body of competent jurisdiction that resulted in such person being subject to a judgment, decree, or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

ITEM 3. SOURCE AND AMOUNT OF FUNDS OR OTHER CONSIDERATION.

As more fully described in Item 4 hereof, as an inducement for PLx to enter into the Merger Agreement, and in consideration thereof, the Shareholders entered into the Voting Agreements with PLx pursuant to which each Shareholder agreed to vote all of its Shares and any other shares of capital stock of the Company owned, beneficially or of record, as set forth in Schedule 1 of each Voting Agreement (its "Owned Securities") in favor of the approval of the Merger, pursuant to which the shares of PLx will be converted into the right to receive common shares of the Company. The Shares to which this Statement relates have not been purchased by PLx, and thus no funds have been used for such purpose. Other than the consideration to be delivered by PLx pursuant to the Merger Agreement in accordance with the terms and conditions of the Merger Agreement, PLx has paid no funds or other consideration in connection with the execution and delivery of the Voting Agreements. For a description of the Voting Agreements and the Merger Agreement, see Item 4 below, which description is incorporated by reference in response to this Item 3.

Cusip No. 25456J104 Page 4 of 8 Pages

ITEM 4. PURPOSE OF TRANSACTION.

(a) - (j)

Merger Agreement

As set forth in the Introductory Note, on December 22, 2016, PLx, the Company, and AcquireCo entered into the Merger Agreement. The Merger Agreement provides for the merger of AcquireCo with and into PLx, with PLx surviving the Merger as a wholly owned subsidiary of the Company, on the terms and subject to the conditions set forth in the Merger Agreement. Upon consummation of the Merger, the shares of PLx's common stock will be converted into Shares based on the Equity Exchange Ratio (as defined in the Merger Agreement).

The Merger Agreement contains certain termination rights for both the Company and PLx, as well as customary representations, warranties and covenants made by the Company and PLx, including as to the indemnification of directors and officers, and the Company's and PLx's conduct of their respective businesses between the date of signing the Merger Agreement and the closing of the Merger. Pursuant to the terms of the Merger Agreement, the Company will make a bridge loan available to PLx in the principal amount of up to \$2 million by January 15, 2017.

The Company's and PLx's obligations to consummate the Merger are subject to the satisfaction or waiver of customary closing conditions, including, among others, obtaining the requisite approvals of the stockholders of the Company and PLx, the Company's having a minimum level of cash of \$12 million at the time of closing, and the effectiveness of a registration statement on Form S-4 relating to the shares of Common Stock to be issued to PLx stockholders pursuant to the Merger Agreement.

The above description of the Merger and the Merger Agreement is qualified in its entirety by the text of the Merger Agreement, which is attached hereto as Exhibit 99.1 and incorporated herein by reference.

Voting Agreements

As an inducement for PLx to enter into the Merger Agreement, and in consideration thereof, the Shareholders entered into the Voting Agreements with PLx. Pursuant to the Voting Agreements, each of the Shareholders has agreed, at any meeting of the shareholders of the Company, or at any adjournment or postponement thereof, or in connection with any written consent of the shareholders of the Company, to vote (or cause to be voted) all of such Shareholder's Owned Securities: (i) in favor of (A) the adoption of the Merger Agreement and the approval of the Merger, and (B) the DPRX Charter Amendment (as defined in the Merger Agreement); (ii) against any action, proposal, transaction or agreement that, to the knowledge of such Shareholder, would reasonably be expected to result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of the Company under the Merger Agreement or that would reasonably be expected to result in any of the conditions to closing under the Merger Agreement not being fulfilled; and (iii) against any DPRX Acquisition Proposal (as defined in the Merger Agreement), or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement.

Each Shareholder also agreed, among other things, subject to certain exceptions, not to, directly or indirectly: (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of any Owned Securities, (b) deposit any Owned Securities into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than pursuant to the Voting Agreement), or (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of any Owned Securities.

Cusip No. 25456J104 Page 5 of 8 Pages

The Voting Agreements and the proxies granted thereunder terminate upon the earlier to occur of (a) the Merger Effective Time (as defined in the Merger Agreement), (b) such date and time as the Merger Agreement shall be terminated pursuant to <u>Article VII</u> thereof or otherwise, or (c) upon mutual written agreement of the parties to a given Voting Agreement.

The cover page of this Statement refers to 4,414,270 Shares beneficially owned by PLx with shared voting power and shared dispositive power, representing approximately 39.63% of the total outstanding Shares based on 10,401,122 Shares reported outstanding as of December 22, 2016 (as represented by the Company in the Merger Agreement). However, such percentage and number of Shares may change at or prior to the time of the applicable vote.

The references to, and descriptions of, the Merger Agreement and the Voting Agreements are not intended to be complete and are qualified in their entirety by reference to the Merger Agreement and the Voting Agreements, copies of which are filed as Exhibits hereto and which are incorporated herein by reference.

Except as set forth in this Statement, the Voting Agreements or the Merger Agreement, neither PLx nor, to the knowledge of PLx, any of the persons listed on Schedule A hereto, has any present plans or proposals which relate to or would result in any of the transactions described in subparagraphs (a) through (j) of Item 4 of Schedule 13D.

ITEM 5. INTEREST IN SECURITIES OF THE ISSUER.

- (a) Under the definition of "beneficial ownership" as set forth in Rule 13d-3 under the Securities Exchange Act of 1934, as a result of entering into the Voting Agreements, PLx may be deemed to have beneficial ownership of 4,414,270 Shares, constituting approximately 39.63% of the 10,401,122 Shares outstanding as of December 22, 2016 (as represented by the Company in the Merger Agreement). Neither the filing of this Statement nor any of its contents shall be deemed to constitute an admission by PLx that it is the beneficial owner of any of the Shares referred to herein for purposes of the Act, or for any other purpose, and such beneficial ownership is expressly disclaimed. To PLx's knowledge, no Shares are beneficially owned by any of the persons listed on Schedule A hereto.
- (b) As a result of entering into the Voting Agreements, PLx may be deemed to have shared power to vote 4,414,270 Shares held by the Shareholders.
- (c) Except for the Voting Agreements, the Merger Agreement, and the transactions contemplated by those agreements, neither PLx nor, to PLx's knowledge, any person named on Schedule A has effected any transaction in the Shares during the past 60 days.
- (d) To PLx's knowledge, no person has the right to receive or power to direct the receipt of dividends from, or the proceeds from the sale of, the Shares held by the Shareholders, other than the Shareholders.
- (e) Not applicable.

CONTRACTS, ARRANGEMENTS, UNDERSTANDINGS OR RELATIONSHIPS WITH RESPECT TO SECURITIES OF THE ISSUER.

The information set forth in Items 1, 3, and 4 above is incorporated herein by reference.

Cusip No. 25456J104 Page 6 of 8 Pages

ITEM 7. MATERIAL TO BE FILED AS EXHIBITS.

99.1	Agreement and Plan of Merger and Reorganization dated as of December 22, 2016 by and among the Company, PLx and Dipexium Acquisition Corp.
99.2	Voting Agreement, dated December 22, 2016, by and between the Company, PLx, and Robert J. DeLuccia 2016 Annuity Trust
99.3	Voting Agreement, dated December 22, 2016, by and between the Company, PLx, and Robert J. DeLuccia
99.4	Voting Agreement, dated December 22, 2016, by and between the Company, PLx, and David P. Luci 2016 Annuity Trust
99.5	Voting Agreement, dated December 22, 2016, by and between the Company, PLx, and David P. Luci

Cusip No. 25456J104 Page 7 of 8 Pages

SIGNATURES

After reasonable inquiry and to the best of the knowledge and belief of the undersigned, the undersigned certify that the information set forth in this statement is true, complete, and correct.

Date:	Decem	ber	30,	2016
		~ ~	,	

PLX PHARMA INC.

By: /s/ Natasha Giordano

Natasha Giordano, Chief Executive Officer

Cusip No. 25456J104 Page 8 of 8 Pages

SCHEDULE A

Name	Present Principal Occupation or Employment	Citizenship
Board of Directors		
Michael Valentino	Executive Chairman of the Board of PLx Pharma Inc.	United States
Natasha Giordano President and Chief Executive Officer of PLx Pharma Inc.; Director of Aceto Corporation		United States
Gary Balkema	Director of Brady Corporation	United States
Robert Casale	Director of First Aid Shot Therapy; Director of Common Sense LTD	United States
Kirk Calhoun	Director of NantHealth, Inc.; Director of Great Basin Scientific, Inc.; and Director of Ryerson Holding Corporation	United States
John W. Hadden II	Chief Executive Officer of IRX Therapeutics, Inc.	United States
Executive Officers		
David Jorden	Acting Chief Financial Officer of PLx Pharma Inc.; Chief Executive Officer and Director of Nanospectra Biosciences, Inc.; and Chief Executive Officer and Chief Financial Officer, and Director of Nuo Therapeutics, Inc.	United States
Gary Mossman	Chief Operating Officer, PLx Pharma Inc.	United States

AGREEMENT AND PLAN OF MERGER AND REORGANIZATION

among

PLX PHARMA INC.

and

DIPEXIUM PHARMACEUTICALS, INC.

and

DIPEXIUM ACQUISITION CORP.

December 22, 2016

TABLE OF CONTENTS

		<u> Page</u>
	ARTICLE I	
	INTERPRETATION	
1.1	Definitions	4
1.2	Currency	16
1.3	Interpretation Not Affected by Headings	16
1.4	Knowledge and Disclosure	16
	ARTICLE II	
	THE MERGER	
2.1	The Merger	16
2.2	The Closing	22
2.3	Preparation of Joint Proxy Statement and Registration Statements	22
2.4	Shareholder Meetings	23
2.5	Calculation of DPRX Closing Cash	25
	Culturation of 21 fair closing custs	
	ARTICLE III	
	REPRESENTATIONS AND WARRANTIES	
3.1	Representations and Warranties of DPRX	25
3.2	Representations and Warranties of PLx	43
3.3	Survival of Representations and Warranties	58
	A DOTAL TO THE	
	ARTICLE IV	
	COVENANTS REGARDING THE CONDUCT OF BUSINESS	
4.1	Covenants of DPRX	59
4.2	Covenants of PLx	62
1.2	COVERGING OF LEA	02
	ARTICLE V	
	ADDITIONAL COVENANTS	
5.1	Access to Information	65
5.2	Consents and Approvals	65
5.3	Covenants of DPRX Regarding the Transaction	67
5.4	Covenants of PLx Regarding the Transaction	67
5.5	[Reserved]	67
5.6	Indemnification and Insurance	68
5.7	Rule 16b-3 Actions	69
5.8	Stock Exchange Listing	69
5.9	Takeover Statutes	69
5.10	Board of Directors and Officers	70
5.11	Lock-Up Agreements	70
	1	
	1	

	ARTICLE VI	
	ACQUISITION PROPOSALS	
6.1	DPRX Non-Solicitation	70
6.2	DPRX Superior Proposal or Intervening Event	72
6.3	Securities Law Compliance	72
6.4	PLx Non-Solicitation	72
	ARTICLE VII	
	TERMINATION	
7.1	Termination	73
7.2	Termination Fee Events	74
	ARTICLE VIII	
	CONDITIONS PRECEDENT	
8.1	Mutual Conditions Precedent	75
8.2	Additional Conditions Precedent to the Obligations of DPRX	76
8.3	Additional Conditions Precedent to the Obligations of PLx	77
8.4	Notice and Cure Provisions	78
	ARTICLE IX	
	GENERAL	
9.1	Notices	78
9.2	Expenses	79
9.3	No Assignment	79
9.4	Benefit of Agreement	79
9.5	Public Announcements	79
9.6	Governing Law; Attornment; Service of Process; Waiver of Jury	80
9.7	Entire Agreement	80
9.8	Third Party Beneficiaries	80
9.9	Amendment	81
9.10	Waiver and Modifications.	81
9.11	Severability	81
9.12	Further Assurances	81
9.13	Injunctive Relief	81
9.14	Counterparts	81

 $Schedule\ I-List\ of\ PLx\ and\ DPRX\ executive\ officers,\ directors\ and\ holders\ of\ PLx\ Shares\ and\ DPRX\ Shares\ signing\ PLx\ Lock-Up\ Agreement\ and\ DPRX\ Lock-Up\ Agreement$

Schedule 5.10 – Officers of DPRX after the Merger Effective Time

Exhibit A -Form of DPRX Voting Agreement

Exhibit B –Form of PLx Voting Agreement

Exhibit C – Form of PLx Lock-Up Agreement

Exhibit D – Form of DPRX Lock-Up Agreement

Exhibit E – Form of Promissory Note

AGREEMENT AND PLAN OF MERGER AND REORGANIZATION

THIS AGREEMENT is made as of December 22, 2016, among PLx Pharma Inc. a corporation incorporated under the laws of the State of Delaware ("PLx"), Dipexium Pharmaceuticals, Inc., a corporation incorporated under the laws of the State of Delaware ("DPRX"), and Dipexium Acquisition Corp., a corporation incorporated under the laws of the State of Delaware and a wholly-owned subsidiary of DPRX ("AcquireCo").

WHEREAS, the Parties intend that AcquireCo be merged with and into PLx with PLx surviving such merger on the terms and conditions of this Agreement (the "**Merger**") whereby the stockholders of PLx will receive DPRX Shares (as defined herein) in exchange for their PLx Shares (as defined herein);

WHEREAS, the Parties intend, by approving resolutions authorizing this Agreement, to adopt this Agreement as a "plan of reorganization" within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Code") and the Treasury Regulations promulgated thereunder, to cause the Merger to qualify as a "reorganization" under the provisions of Section 368(a) of the Code, and that DPRX, PLx and AcquireCo will each be a "party to the reorganization" within the meaning of Section 368(b) of the Code;

WHEREAS, the board of directors of DPRX (i) has determined that the Merger is fair to, and in the best interests of, DPRX and its stockholders, (ii) has approved, adopted and declared advisable this Agreement, the Merger, the issuance of DPRX Shares to the stockholders of PLx pursuant to the terms of this Agreement, the change of control of DPRX, and the other actions contemplated by this Agreement and (iii) has determined to recommend that the stockholders of DPRX vote to approve the DPRX Stockholder Resolution (as defined herein);

WHEREAS, the board of directors of AcquireCo (i) has determined that the Merger is fair to, and in the best interests of, AcquireCo and its sole stockholder, (ii) has approved, adopted and declared advisable this Agreement, the Merger, and the other actions contemplated by this Agreement and (iii) has determined to recommend that its sole stockholder vote to adopt this Agreement;

WHEREAS, the board of directors of PLx (i) has determined that the Merger is fair to, and in the best interests of, PLx and its stockholders, (ii) has approved, adopted and declared advisable this Agreement, the Merger and the other transactions contemplated by this Agreement and (iii) has determined to recommend that the stockholders of PLx vote to approve the Merger;

WHEREAS, in order to induce PLx to enter into this Agreement and to cause the Merger to be consummated, certain directors, officers and stockholders of DPRX (collectively, the "DPRX Specified Stockholders") intend to enter into voting agreements, in substantially the form attached hereto as Exhibit A, concurrently with the execution of this Agreement, providing that, among other things, the DPRX Specified Stockholders will support the DPRX Stockholder Resolution and the other transactions contemplated by this Agreement (the "DPRX Voting Agreements").

WHEREAS, in order to induce DPRX and AcquireCo to enter into this Agreement and to cause the Merger to be consummated, the directors, officers and certain stockholders of PLx (collectively, the "PLx Specified Stockholders") intend to enter into voting agreements, in substantially the form attached hereto as Exhibit B, concurrently with the execution of this Agreement, providing that, among other things, the PLx Specified Stockholders will support the Merger and the other transactions contemplated by this Agreement (the "PLx Voting Agreements" and, together with the DPRX Voting Agreements, the "Voting Agreements")).

WHEREAS, in order to induce DPRX and AcquireCo to cause the Merger to be consummated, each of PLx's executive officers, directors and holders of PLx Shares listed on Schedule I will execute lock-up agreements in favor of DPRX prior to the Closing relating to sales and certain other dispositions of DPRX Shares or certain other securities in substantially the form attached hereto as Exhibit C (the "PLx Lock-up Agreements").

WHEREAS, in order to induce PLx to cause the Merger to be consummated, the individuals listed on Schedule I will execute lock-up agreements in favor of PLx prior to the Closing relating to sales and certain other dispositions of PLx Shares or certain other securities in substantially the form attached hereto as Exhibit D (the "DPRX Lock-up Agreements").

NOW THEREFORE in consideration of the premises and the covenants and agreements contained herein, the Parties agree as follows:

ARTICLE I INTERPRETATION

- 1.1 <u>Definitions</u>. In this Agreement, unless otherwise defined or expressly stated herein or something in the subject matter or the context is clearly inconsistent therewith:
 - "AcquireCo" shall have the meaning ascribed to it in the Recitals.
 - "Affiliate" shall have the meaning ascribed to it in Rule 405 promulgated under the Securities Act.
- "Agreement" means this agreement and plan of merger and reorganization (including the Schedules attached hereto) as the same may be amended, supplemented, restated or otherwise modified from time to time in accordance with the terms hereof.
- "Business Day" means a day other than a Saturday, a Sunday or any other day on which major commercial banking institutions in New York, New York are closed for business.
 - "Certificate" shall have the meaning ascribed to it in Section 2.1(f)(iii).
 - "Certificate of Merger" means the certificate of merger relating to the Merger.
 - "Chancery Court" shall have the meaning ascribed to it in Section 9.6(a).
 - "Closing" shall have the meaning ascribed to it in Section 2.2.
 - "Closing Date" shall have the meaning ascribed to it in Section 2.2.
 - "Code" shall have the meaning ascribed to it in the Recitals.
- "Contract" means any legally binding contract, agreement, indenture, note, instrument, license, franchise, lease, arrangement, commitment, understanding or other right or obligation (whether written or oral) to which DPRX or any of its Subsidiaries, on the one hand, or PLx or any PLx Subsidiary, on the other hand, is a party or by which DPRX or any of its Subsidiaries, on the one hand, or PLx or any of its Subsidiaries, on the other hand, is bound or affected or to which any of their respective properties or assets is subject.

"DGCL" means the General Corporation Law of the State of Delaware.

"**DPRX**" shall have the meaning ascribed to it in the Recitals.

"DPRX Acquisition Agreement" shall have the meaning ascribed to it in Section 6.1(a)(iv).

"DPRX Acquisition Proposal" means, at any time, whether or not in writing, any proposal or offer (including any modification or proposed modification thereto), with respect to:

- (a) the acquisition or purchase by any Person or group of Persons acting jointly or in concert of any capital stock or other voting securities, or securities convertible into or exercisable or exchangeable for any DPRX Shares or other voting securities of DPRX or any of its Subsidiaries representing 20% or more of the outstanding voting securities of DPRX or such Subsidiary; or
- (b) the acquisition or purchase by any Person or group of Persons acting jointly or in concert of any assets of DPRX and/or one or more of its Subsidiaries (including equity interests of any Subsidiary of DPRX) which assets individually or in the aggregate contribute 20% or more of the consolidated revenue or represent 20% or more of the total asset value of DPRX and its Subsidiaries taken as a whole (in each case based on the consolidated financial statements of DPRX most recently filed prior to such time as part of the DPRX Public Disclosure Record) (or any lease, license, royalty, long- term supply agreement or other arrangement having a similar economic effect); or
- (c) a merger, amalgamation, recapitalization, reorganization, or other business combination (including by way of plan of arrangement) involving DPRX or any of its Subsidiaries whether in a single transaction or a series of related transactions (i) in which DPRX (or its Subsidiaries) is a constituent corporation, (ii) in which a Person or "group" (as defined in the Exchange Act and the rules promulgated thereunder) of Persons directly or indirectly acquires beneficial or record ownership of securities representing more than 20% of the outstanding securities of any class of voting securities of DPRX (or its Subsidiaries) issues securities representing more than 20% of the outstanding securities of any class of voting securities of any such entity (other than as contemplated under this Agreement),

in each case excluding the Transaction and excluding any transaction between only DPRX and/or one or more of its Subsidiaries.

"DPRX Annual Financial Statements" means the audited consolidated financial statements of DPRX as of and for the years ending December 31, 2015 and 2014, together with the notes thereto.

"DPRX Board of Directors" means the board of directors of DPRX.

"DPRX Change of Recommendation" means any of the following:

(a) the DPRX Board of Directors fails to publicly make the DPRX Recommendation or withholds, withdraws, modifies, changes or qualifies in a manner adverse to PLx its approval of the Merger or the DPRX Recommendation;

- (b) PLx requests in writing that the DPRX Board of Directors publicly reaffirm the DPRX Recommendation and/ or publicly reject any DPRX Acquisition Proposal and the DPRX Board of Directors, in each case, shall not have done so within five Business Days following receipt of such request;
 - (c) the DPRX Board of Directors accepts, approves, endorses or recommends any DPRX Acquisition Proposal;
- (d) DPRX enters into a DPRX Acquisition Agreement related to, or that is intended to or is reasonably expected to lead to, any DPRX Acquisition Proposal; or
- (e) DPRX or the DPRX Board of Directors publicly proposes or announces its intention to do any of the foregoing,

it being understood that publicly taking a neutral position or no position with respect to any DPRX Acquisition Proposal until five Business Days following the public announcement of such DPRX Acquisition Proposal shall not be considered a DPRX Change of Recommendation (it being further understood that after five Business Days following the public announcement of such DPRX Acquisition Proposal, continuing to take no position or a neutral position will be deemed to be a DPRX Change of Recommendation).

"DPRX Charter Amendments" means amendments to the Certificate of Incorporation of DPRX that (a) increase the number of authorized DPRX Shares in a manner sufficient to allow for the issuance of all DPRX Shares issuable (i) as Merger Consideration, and (ii) upon exercise of the Replacement PLx Options, (b) changes the name of DPRX to "PLx Pharma Inc.", and (c) if deemed reasonably necessary by PLx, effect the Reverse Stock Split.

"DPRX Closing Cash" means, as of the Closing Date (a) the total cash, cash equivalents and short-term investments of DPRX, plus (b) security deposits of DPRX of up to \$56,630, plus (c) provided that it is paid in accordance with Section 5.5, the Loan Amount, plus (d) up to \$350,000 in documented legal fees, accounting fees, NASDAQ listing fees, and printing costs paid or payable by DPRX in connection with the Transaction.

"DPRX Closing Cash Schedule" shall have the meaning ascribed to that term in Section 2.5.

"DPRX Disclosure Letter" means the disclosure letter dated the date hereof regarding this Agreement that has been executed by DPRX and delivered to PLx concurrently with the execution of this Agreement.

"DPRX Employment Agreement" shall have the meaning ascribed to it in Section 3.1(q)(i).

"DPRX Fairness Opinion" means the opinion of DPRX's financial advisor to the effect that, as of the date of this Agreement and based on and subject to the assumptions, qualifications and limitations set forth therein, the Equity Exchange Ratio is fair, from a financial point of view, to DPRX.

"DPRX Financial Statements" means the DPRX Annual Financial Statements and the DPRX Interim Financial Statements.

"DPRX Indemnified Party" shall have the meaning ascribed to that term in Section 5.6(a).

- "DPRX Intellectual Property" shall have the meaning ascribed to that term in Section 3.1(s)(i).
- "DPRX Interim Financial Statements" means the unaudited interim consolidated financial statements of DPRX for the nine months ended September 30, 2016, together with the notes thereto.
 - "DPRX Lock-up Agreements" shall have the meaning ascribed to it in the Recitals.
 - "DPRX Material Contract" shall have the meaning ascribed to that term in Section 3.1(o)(i).
- "**DPRX Meeting**" means the special meeting of the DPRX Stockholders, including any adjournment or postponement thereof, to be called and held in accordance with this Agreement for the purpose of considering and approving the DPRX Stockholder Resolution.
 - "DPRX Option" means an option to purchase DPRX Shares granted under the DPRX Stock Plan.
 - "DPRX Parties" means collectively DPRX and AcquireCo and "DPRX Party" means either of them.
- "DPRX Percentage" means (A) 23.25%, if DPRX Closing Cash is at least \$12,500,000, and (B) 22.5% if DPRX Closing Cash is at least \$12,000,000 but less than \$12,500,000. For purposes of the definition of "Merger Shares," the "DPRX Percentage" shall be expressed as (i) 0.2325 in the case of clause (A) of this definition of "DPRX Percentage" or (ii) 0.225 in the case of clause (B) of this definition of "DPRX Percentage."
 - "**DPRX Plan**" shall have the meaning ascribed to that term in Section 3.1(r)(i).
 - "DPRX Product" shall have the meaning ascribed to it in Section 3.1(t)(viii).
 - "DPRX Public Disclosure Record" means all documents filed by or on behalf of DPRX on EDGAR since December 31, 2013.
- "DPRX Recommendation" means the recommendation of the DPRX Board of Directors that the DPRX Stockholders vote in favor of the DPRX Stockholder Resolution.
 - "DPRX Senior Management" means the individuals set forth in Section 1.4 of the DPRX Disclosure Letter.
 - "DPRX Shares" means the shares of common stock, par value \$0.001 per share, of DPRX.
 - "DPRX Specified Stockholders" shall have the meaning ascribed to it in the Recitals.
 - "DPRX Stockholder" means a holder of one or more DPRX Shares.
- "DPRX Stockholder Approval" means the affirmative vote of a majority of the votes cast on the DPRX Stockholder Resolution by the DPRX Stockholders present in person or represented by proxy at the DPRX Meeting.
- "DPRX Stockholder Resolution" means the resolution of DPRX Stockholders approving (i) the relevant provisions of the DPRX Charter Amendments that require DPRX Stockholder Approval, and (ii) the issuance of DPRX Shares pursuant to the Merger to be considered and, if thought fit, passed with or without variation at the DPRX Meeting.
 - "DPRX Stock Plan" means the DPRX 2013 Equity Incentive Plan as amended.

"DPRX Subsidiary" means a Subsidiary of DPRX.

"DPRX Superior Proposal" means an unsolicited *bona fide* written DPRX Acquisition Proposal (*provided*, *however*, that, for the purposes of this definition, all references to "20%" in the definition of "DPRX Acquisition Proposal" as it relates to securities of DPRX shall be changed to "100%" and references to "20%", as regards the assets of DPRX, shall be changed to "all or substantially all") made by a Person or Persons acting jointly or in concert (other than PLx and any of its Affiliates) and which, or in respect of which:

- (a) the DPRX Board of Directors has determined in good faith, after consultation with its financial advisors and outside legal counsel:
 - (i) would, if consummated taking into account all of the terms and conditions of such DPRX Acquisition Proposal (but not assuming away any risk of non-completion), result in a transaction which is more favorable to the DPRX Stockholders from a financial point of view than the Transaction (including any adjustment to the terms and conditions of the Transaction proposed by PLx pursuant to Section 6.2);
 - (ii) is reasonably capable of being completed in accordance with its terms, without undue delay, taking into account all legal, financial, regulatory and other aspects of such DPRX Acquisition Proposal and the Person or Persons making such DPRX Acquisition Proposal; and
 - (iii) that funds, securities or other consideration necessary for the DPRX Acquisition Proposal are or are reasonably likely to be available; and
- (b) in the case of a DPRX Acquisition Proposal involving the DPRX Shares, is made available to all of the DPRX Stockholders on the same terms and conditions.

"DPRX Termination Fee" shall have the meaning ascribed to it in Section 7.2(a).

"DPRX Termination Fee Event" shall have the meaning ascribed to it in Section 7.2(b).

"**DPRX Warrant**" means a warrant to purchase DPRX Shares.

"DPRX Voting Agreements" shall have the meaning ascribed to it in the Recitals.

"EDGAR" means the Electronic Data-Gathering, Analysis and Retrieval system maintained by the SEC.

"Equity Exchange Ratio" shall equal the quotient obtained by dividing (A) the Merger Shares by (B) the Outstanding PLx Shares.

"ERISA" shall have the meaning ascribed to it in Section 3.1(r)(vi).

"Exchange Act" means the United States Securities Exchange Act of 1934.

"Exchange Agent" shall have the meaning ascribed to it in Section 2.1(g)(i).

"FDA" means the United States Food and Drug Administration or any successor entity.

"FDA Regulations" shall have the meaning ascribed to it in Section 3.1(t)(iii).

"FDCA" shall have the meaning ascribed to it in Section 3.1(t)(i).

"Form S-4" shall have the meaning ascribed to it in Section 2.3(a).

"Form S-8" shall have the meaning ascribed to it in Section 2.3(i).

"Fraud Policy" shall have the meaning ascribed to it in Section 3.1(t)(iv).

"Governmental Authority" means any international, multinational, federal, provincial, territorial, state, regional, municipal, local or other government or governmental body and any ministry, department, division, bureau, agent, official, agency, commission, board or authority of any government, governmental body, quasi-governmental or private body (including NASDAQ or any other stock exchange), domestic or foreign, exercising any statutory, regulatory, expropriation or taxing authority under the authority of any of the foregoing and any domestic, foreign or international judicial, quasi-judicial or administrative court, tribunal, commission, board, panel, arbitrator or arbitral body acting under the authority of any of the foregoing.

"HIPAA" shall have the meaning ascribed to it in Section 3.1(t)(i).

"HSR Act" means the United States Hart-Scott-Rodino Antitrust Improvements Act of 1976.

"Indemnified Party" and "Indemnified Parties" have the meanings ascribed thereto in Section 5.6(a).

"Intellectual Property" means all intellectual property and industrial property rights and rights in confidential information of every kind and description throughout the world, including all United States, Canadian and foreign (i) patents, patent applications, invention disclosures, and all related continuations, continuations-in-part, divisionals, reissues, re-examinations, substitutions, and extensions thereof ("Patents"), (ii) registered or unregistered trademarks, service marks, names, corporate names, trade names, domain names, logos, slogans, trade dress, design rights, and other similar designations of source or origin, together with the goodwill symbolized by any of the foregoing ("Trademarks"), (iii) copyrights and copyrightable subject matter ("Copyrights"), (iv) rights in computer programs (whether in source code, object code, or other form), algorithms, databases, compilations and data, technology supporting the foregoing, and all documentation, including user manuals and training materials, related to any of the foregoing ("Software"), (v) trade secrets and all other confidential information, ideas, know- how, inventions, proprietary processes, formulae, models, and methodologies, (vi) rights of publicity, privacy, and rights to personal information, (vii) moral rights and rights of attribution and integrity, (viii) all rights in the foregoing and in other similar intangible assets and (ix) all applications and registrations for the foregoing.

"Intervening Event" means any event, change, effect, development, condition or occurrence that (a) does not relate to any DPRX Acquisition Proposal and (b) is not known and was not reasonably foreseeable to the DPRX Board of Directors as of the date hereof.

"Joint Proxy Statement" shall have the meaning ascribed to it in Section 2.3(a).

"Laws" means any and all laws, statutes, codes, ordinances (including zoning), approvals, rules, regulations, instruments, bylaws, notices, policies, protocols, guidelines, guidance, manuals, treaties or other requirements of any Governmental Authority having the force of law and any legal requirements arising under the common law or principles of law or equity.

"Letter of Transmittal" shall have the meaning ascribed to it in Section 2.1(g)(ii).

"Liens" means any pledge, claim, lien, charge, option, hypothec, mortgage, security interest, restriction, adverse right, prior assignment, lease, sublease, license, sublicense, right to possession or any other encumbrance, right or restriction of any kind or nature whatsoever, whether contingent or absolute, or any agreement, option, right or privilege (whether by Law, contract or otherwise) capable of becoming any of the foregoing.

"Loan Amount" shall have the meaning ascribed to it in Section 5.5.

"Material Adverse Effect", when used in connection with PLx or DPRX, means any result, fact, change, effect, event, circumstance, occurrence or development that, individually or in the aggregate with all other adverse results, facts, changes, effects, events, circumstances, occurrences or developments, has or would reasonably be expected to have, a material and adverse effect on (i) the business, operations, results of operations or condition (whether financial or otherwise) of such Party and its Subsidiaries, taken as a whole or (ii) the ability of PLx, DPRX or either Party's Subsidiaries to perform their covenants or obligations under this Agreement or to consummate the Transaction; *provided*, *however*, that any result, fact, change, effect, event, circumstance, occurrence or development shall not be deemed to constitute, and shall not be taken into account in determining whether there has been, a Material Adverse Effect to the extent that such result, fact, change, effect, event, circumstance, occurrence or development arises out of or results from:

- (a) changes, developments or conditions in or relating to general international, political, economic or financial or capital market conditions, or political, economic or financial or capital market conditions in any jurisdiction in which such Party or any of its Subsidiaries operates or carries on business;
- (b) changes, developments or conditions resulting from any act of sabotage or terrorism or any outbreak of hostilities or declared or undeclared war, or any escalation or worsening of such acts of sabotage, terrorism, hostilities or war;
 - (c) any natural disaster;
 - (d) changes or developments in or relating to currency exchange or interest rates;
 - (e) changes or developments affecting the pharmaceutical industry in general;
 - (f) any change in applicable Laws (other than Orders against a Party or a Subsidiary thereof) or U.S. GAAP;
- (g) except for purposes of Sections 3.1(c), 3.1(d), 3.2(c) and 3.2(d), the announcement of the execution of this Agreement or the Transaction;
 - (h) any actions taken (or omitted to be taken) by DPRX or PLx upon the express written request of the other;
 - (i) with respect to DPRX, any of the matters described on Section 1.1 of the DPRX Disclosure Letter;
- (j) (A) any changes in the share price or trading volume of DPRX Shares or the credit rating or in any analyst's recommendation with respect to DPRX, or (B) any failure of DPRX to meet projections, guidance, milestones, forecasts or published financial or operating predictions or measures (it being agreed that the facts and circumstances giving rise to any of the foregoing events or failures, unless expressly excluded by another clause of this definition, may constitute and/or may be taken into account in determining whether a Material Adverse Effect has occurred or is reasonably likely to occur); or

- (k) with respect to PLx, any of the matters described on Section 1.1 of the PLx Disclosure Letter.
- "Merger" shall have the meaning ascribed to it in the Recitals.
- "Merger Consideration" shall have the meaning ascribed to it in Section 2.1(f)(iii).
- "Merger Effective Time" means the time at which the Merger becomes effective in accordance with Section 2.1(b) and the DGCL.
- "Merger Shares" means the total number of DPRX shares to be issued in the Merger to PLx Stockholders pursuant to Section 2.1(f)(iii), determined as follows: (A) the quotient of (i) the Outstanding DPRX Shares divided by (ii) the DPRX Percentage, minus (B) the Outstanding DPRX Shares.
 - "NASDAQ" means the NASDAQ Capital Market.
- "**Non-Disclosure Agreement**" means the non-disclosure agreement dated as of November 10, 2016 between DPRX and PLx, as it may be amended, restated, supplemented or otherwise modified from time to time.
- "Order" means all judicial, arbitral, administrative, ministerial, departmental or regulatory judgments, injunctions, orders, decisions, rulings, determinations, awards, decrees or similar actions taken by, or applied by, any Governmental Authority (in each case, whether temporary, preliminary or permanent).
- "ordinary course of business", or any similar reference, means, with respect to an action taken or to be taken by any Person, that such action is consistent with the past practices of such Person (including with respect to amount and frequency) and is taken in the ordinary course of the normal day-to-day business and operations of such Person.
 - "**Orphan Act**" shall have the meaning ascribed to it in Section 3.1(t)(i).
- "Outside Date" means April 30, 2017, or such later date as may be (i) established pursuant to Section 7.1(b)(1), or (ii) otherwise agreed to in writing by the Parties.
- "Outstanding DPRX Shares" means the total number of DPRX Shares issued and outstanding immediately prior to the Merger Effective Time.
- "Outstanding PLx Shares" shall mean the total number of PLx Shares issued and outstanding immediately prior to the Merger Effective Time (which, for the avoidance of doubt, shall include all PLx Shares issued immediately prior to the Merger Effective Time upon the conversion of PLx Convertible Notes).
 - "Parties" means the parties to this Agreement and "Party" means any one of them.
- "Permit" means any lease, license, permit, certificate, consent, order, grant, approval, classification, registration or other authorization of or from any Governmental Authority.

"Permitted Liens" means, for DPRX or any of its Subsidiaries, or PLx or any of its Subsidiaries, as the context requires: (i) any Liens for Taxes not yet due and payable or which are being contested in good faith by appropriate proceedings and for which adequate reserves have been provided in conformity with U.S. GAAP, as applicable; (ii) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other similar Liens; (iii) pledges or deposits in connection with workers' compensation, unemployment insurance, and other social security legislation; (iv) easements, rights-of-way, covenants, restrictions and other encumbrances incurred in the ordinary course of business that, in the aggregate, are not material in amount and that do not, in any case, materially detract from the value or the use of the property subject thereto; (v) statutory landlords' Liens and Liens granted to landlords under any lease, (vi) licenses of non-material Intellectual Property in the ordinary course of business; (vii) any purchase money security interests, equipment leases or similar financing arrangements; (viii) any Liens which are disclosed on the most recent consolidated balance sheet of DPRX or PLx, as applicable, or the notes thereto; and (ix) any Liens that are not material to DPRX, its Subsidiaries or their businesses, taken as a whole, as applicable.

"Person" includes an individual, sole proprietorship, corporation, body corporate, incorporated or unincorporated association, syndicate or organization, partnership, limited partnership, limited liability company, unlimited liability company, joint venture, joint stock company, trust, natural person in his or her capacity as trustee, executor, administrator or other legal representative, a government or Governmental Authority or other entity, whether or not having legal status.

"PHSA" shall have the meaning ascribed to it in Section 3.1(t)(i).

"PLx" shall have the meaning ascribed to it in the Recitals.

"PLx Acquisition Proposal" means, at any time, whether or not in writing, any proposal or offer (including any modification or proposed modification thereto), with respect to:

- (a) the acquisition or purchase by any Person or group of Persons acting jointly or in concert of any capital stock or other voting securities, or securities convertible into or exercisable or exchangeable for any PLx Shares or other voting securities of PLx or any of its Subsidiaries representing 20% or more of the outstanding voting securities of PLx or such Subsidiary; or
- (b) the acquisition or purchase by any Person or group of Persons acting jointly or in concert of any assets of PLx and/or one or more of its Subsidiaries (including equity interests of any Subsidiary of PLx) which assets individually or in the aggregate contribute 20% or more of the consolidated revenue or represent 20% or more of the total asset value of PLx and its Subsidiaries taken as a whole (in each case based on the consolidated financial statements of PLx most recently filed prior to such time) (or any lease, license, royalty, long-term supply agreement or other arrangement having a similar economic effect); or
- (c) a merger, amalgamation, recapitalization, reorganization, or other business combination (including by way of plan of arrangement) involving PLx or any of its Subsidiaries whether in a single transaction or a series of related transactions (i) in which PLx (or its Subsidiaries) is a constituent corporation, (ii) in which a Person or "group" (as defined in the Exchange Act and the rules promulgated thereunder) of Persons directly or indirectly acquires beneficial or record ownership of securities representing more than 20% of the outstanding securities of any class of voting securities of PLx (or its Subsidiaries), or (iii) in which PLx (or its Subsidiaries) issues securities representing more than 20% of the outstanding securities of any class of voting securities of any such entity (other than as contemplated under this Agreement),

in each case excluding the Transaction and excluding any transaction between only PLx and/or one or more of its Subsidiaries.

- "PLx Annual Financial Statements" means the audited consolidated financial statements of PLx as of and for the years ending December 31, 2015 and 2014, together with the notes thereto.
 - "PLx Board of Directors" means the board of directors of PLx.
- "PLx Convertible Notes" means those certain convertible promissory notes issued by PLx prior to the Closing Date and convertible into an aggregate maximum of 405,000 PLx Shares.
- "PLx Disclosure Letter" means the disclosure letter dated the date hereof regarding this Agreement that has been executed by PLx and delivered to DPRX prior to the execution of this Agreement.
 - "PLx Employment Agreement" shall have the meaning ascribed to it in Section 3.2(q)(i).
- "PLx Fairness Opinion" means the opinion of PLx's financial advisors to the effect that, based upon and subject to the assumptions, limitations, qualifications and conditions set forth therein, as of the date of such opinion, the Equity Exchange Ratio was fair, from a financial point of view, to the PLx Stockholders.
 - "PLx Financial Statements" means PLx Annual Financial Statements and PLx Interim Financial Statements.
 - "PLx Indemnified Party" shall have the meaning ascribed to that term in Section 5.6(a).
- "PLx Interim Financial Statements" means the unaudited consolidated financial statement as of and for the nine months ended September 30, 2016, together with the notes thereto.
 - "PLx Lock-up Agreements" shall have the meaning ascribed to it in the Recitals.
 - "PLx Material Contracts" shall have the meaning ascribed to that term in Section 3.2(o)(i).
- "PLx Meeting" means the special meeting of PLx Stockholders, including any adjournment or postponement thereof to be called and held in accordance with this Agreement for the purpose of obtaining the PLx Stockholder Approval.
 - "PLx Option" means an option issued by PLx to purchase PLx Shares.
 - "PLx Product" shall have the meaning ascribed to it in Section 3.2(r)(viii).
- "PLx Recommendation" means the recommendation of the PLx Board of Directors that PLx Stockholders adopt this Agreement.
 - "PLx Senior Management" means the individuals set forth in Section 1.4 of the PLx Disclosure Letter.
 - "PLx Share" means a share of common stock, par value \$0.001 per share, of PLx.
 - "PLx Specified Stockholders" shall have the meaning ascribed to it in the Recitals.
 - "PLx Stock Plan" means the 2015 Omnibus Incentive Plan of PLx, as amended.
 - "PLx Stockholder" means a holder of one or more PLx Shares.

- "PLx Stockholder Approval" means adoption of this Agreement by affirmative vote or consent of PLx Stockholders holding a majority of the outstanding shares of PLx Shares.
 - "PLx Subsidiary" means a Subsidiary of PLx.
 - "PLx Voting Agreements" shall have the meaning ascribed to it in the Recitals.
- "**Proceeding**" means a court, administrative, regulatory or similar proceeding (whether civil, quasi-criminal or criminal), arbitration or other dispute settlement procedure, investigation or inquiry before or by any Governmental Authority, or any claim, action, suit, demand, arbitration, charge, indictment, hearing or other similar civil, quasi-criminal or criminal, administrative or investigative matter or proceeding.
- "Regulatory Authority" means the FDA and any other federal, state, provincial, local or foreign Governmental Authority with jurisdiction over the authorization, approval, marketing, advertising, sale, pricing, storage, distribution, use, handling and control, safety, efficacy, reliability or manufacturing of pharmaceutical products, including but not limited to human drugs, biologics, and drug combination products.
- "Regulatory Authorization" means any registration, authorization, approval, clearance, license, permit, certificate or exemption issued by any Regulatory Authority or Governmental Authority (including new drug applications, new drug submissions, investigational new drug applications, clinical trial applications, manufacturing approvals and authorizations, pricing and reimbursement approvals, labeling approvals, registration notifications or their foreign equivalent) that are required for the research, development, manufacture, distribution, marketing, storage, transportation, use and sale of the products of DPRX or PLx and their respective Subsidiaries.
- "Regulatory Guidelines" means applicable rules, guidance, manuals, protocols, codes, guidelines, treaties, policies, notices, directions, decrees, judgments, awards or requirements, in each case of any Regulatory Authority to the extent that the foregoing do not have the force of law.
 - "Relevant Laws" shall have the meaning ascribed to it in Section 5.2(b).
- "Replacement PLx Options" means the options to acquire DPRX Shares to be issued in exchange for PLx Options pursuant to this Agreement.
- "Representatives" means, collectively, with respect to a Person, any officers, directors, employees, consultants, advisors, agents or other representatives (including legal counsel, accountants, investment bankers and financial advisors) of that Person or any Subsidiary of that Person.
 - "Restraint" shall have the meaning ascribed to it in Section 5.2(e).
- "Returns" means all reports, forms, elections, designations, schedules, statements, estimates, declarations of estimated tax, information statements and returns relating to, or required to be filed with any Governmental Authority in connection with, any Taxes and including any other filings relating to Taxes, including all returns in respect of Taxes and other material reports and information under the Code or any foreign country or political subdivision thereof in which the relevant Person carries on business or to a jurisdiction of which it is otherwise subject, any sales or excise tax legislation of any state or any foreign country, or political subdivision thereof or legislation affecting any other Taxes, applicable to such Person pursuant to which it is liable or required to pay or remit Taxes (including any schedules or attachments thereto or amendments thereof).

"Reverse Stock Split" means a reverse stock split of the DPRX Shares in connection with the Merger.

"SEC" means the United States Securities and Exchange Commission or any successor entity.

"Securities Act" means the United States Securities Act of 1933.

"Social Security Act" shall have the meaning ascribed to it in Section 3.1(t)(vii).

"Subsidiary" means, with respect to a specified entity, any:

- (a) corporation of which issued and outstanding voting securities of such corporation to which are attached more than 50% of the votes that may be cast to elect directors of the corporation (whether or not shares of any other class or classes will or might be entitled to vote upon the happening of any event or contingency) are at all times owned by such specified entity;
- (b) partnership, unlimited liability company, joint venture or other similar entity in which such specified entity has more than 50% of the equity interests and the power to direct the policies, management and affairs thereof; and
 - (c) Subsidiary (as defined in clauses (a) and (b) above) of any Subsidiary (as so defined) of such specified entity.

"Surviving Company" shall have the meaning ascribed to it in Section 2.1(b).

"Tax" or "Taxes" means all taxes, dues, duties, rates, imposts, fees, levies, other assessments, tariffs, charges or obligations of the same or similar nature, however denominated, imposed, assessed or collected by any Governmental Authority, including (i) all income taxes, including any tax on or based on net income, gross income, income as specifically defined, earnings gross receipts, capital, capital gains, profits, business royalty or selected items of income, earnings or profits, and specifically including any federal, provincial, state, territorial, county, municipal, local or foreign taxes, state profit share taxes, windfall or excess profit taxes, capital taxes, royalty taxes, production taxes, payroll taxes, health taxes, employment taxes, withholding taxes, sales taxes, use taxes, goods and services taxes, custom duties, value added taxes, ad valorem taxes, excise taxes, alternative or add-on minimum taxes, franchise taxes, gross receipts taxes, license taxes, occupation taxes, real and personal property taxes, land transfer taxes, severance taxes, capital stock taxes, stamp taxes, anti-dumping taxes, countervailing taxes, occupation taxes, environment taxes, transfer taxes, and employment or unemployment insurance premiums, social insurance premiums and worker's compensation premiums and pension payments, surtaxes, harmonized sales tax, abandoned or unclaimed property liabilities (escheat) and other taxes, fees, imposts, assessments or charges of any kind whatsoever together with any interest, penalties, additional taxes, fines and other charges and additions that may become payable in respect thereof; (ii) any tax imposed, assessed, collected or payable pursuant to any tax-sharing agreement or any other contract relating to the sharing or payment of any such tax, levy, assessment, tariff, duty, deficiency or fee; and (iii) any liability for any of the foregoing of a transferee, successor, guarantor, or by contract, or by operation of law, as a result of being a member of a consolidated, combined or unitary tax group.

- "Transaction" means, collectively, all the transactions contemplated by this Agreement, including the Merger.
- "U.S. GAAP" means accounting principles generally accepted in the United States, consistently applied.
- "U.S. Securities Laws" means the Securities Act, the Exchange Act and all other state and federal securities Laws and the rules, regulations and published policies made thereunder.
 - "Voting Agreement" shall have the meaning ascribed to it in the Recitals.
- 1.2 <u>Currency</u>. Except where otherwise specified, all references to currency herein are to lawful money of the United States of America and "\$" refers to U.S. dollars.
- 1.3 <u>Interpretation Not Affected by Headings</u>. The division of this Agreement into Articles and sections and the insertion of a table of contents and headings are for convenience of reference only and do not affect the construction or interpretation of this Agreement. The terms "this Agreement", "hereof", "herein", "hereunder" and similar expressions refer to this Agreement, including the Schedules hereto, and not to any particular Article, section or other portion hereof. Unless something in the subject matter or context is clearly inconsistent therewith, references herein to an Article, section or schedule by number or letter or both are to that Article, section or schedule in this Agreement.
- Knowledge and Disclosure. Any reference in this Agreement to the "knowledge" or the "awareness" of DPRX means to the actual knowledge of the DPRX Senior Management, in their capacities as officers or directors of DPRX and not in their personal capacities or in any other capacity, after making reasonable inquiry regarding the relevant matter, and does not include any knowledge or awareness of any other individual. Any reference in this Agreement to the "knowledge" or the "awareness" of PLx means to the actual knowledge of PLx Senior Management, in their capacities as officers or directors of PLx and not in their personal capacities or in any other capacity, after making reasonable inquiry regarding the relevant matter, and does not include any knowledge or awareness of any other individual.

ARTICLE II THE MERGER

2.1 <u>The Merger</u>.

- (a) DPRX, AcquireCo and PLx agree that the Merger shall be implemented in accordance with and subject to the terms and conditions contained in this Agreement.
- (b) On the terms and subject to the conditions set forth in this Agreement, and in accordance with the DGCL, on the Closing Date, AcquireCo shall be merged with and into PLx. At the Merger Effective Time, the separate corporate existence of AcquireCo shall cease and PLx shall continue as the surviving company in the Merger (the "Surviving Company").
- (c) Subject to the provisions of this Agreement, as soon as practicable on the Closing Date, the parties to the Merger shall file with the Secretary of State of the State of Delaware the Certificate of Merger, executed and acknowledged in accordance with the relevant provisions of the DGCL, and, as soon as practicable on or after the Closing Date, shall make all other filings required under the DGCL or by the Secretary of State of the State of Delaware in connection with the Merger. The Merger shall become effective at the time that the Certificate of Merger has been duly filed with the Secretary of State of the State of Delaware, or at such later time as DPRX and PLx shall agree and specify in the Certificate of Merger. At and immediately after the Merger Effective Time, the Merger will have the effects set forth in the Certificate of Merger and the DGCL.

- (d) The certificate of incorporation of AcquireCo, as in effect immediately prior to the Merger Effective Time, shall be the certificate of incorporation of the Surviving Company until thereafter changed or amended as provided therein or by applicable Law. The certificate of incorporation of DPRX, as in effect immediately prior to the Merger Effective Time, shall be the certificate of incorporation of DPRX until thereafter changed or amended as provided therein or by applicable Law; provided, however, that at the Merger Effective Time, DPRX shall file an amendment to its certificate of incorporation to change the name of DPRX to "PLx Pharma Inc." The bylaws of AcquireCo, as in effect immediately prior to the Merger Effective Time, shall be the bylaws of the Surviving Company until thereafter changed or amended as provided therein or by applicable Law.
- (e) The directors and officers of the Surviving Company upon completion of the Merger shall be as set forth in Section 5.10.
- (f) At the Merger Effective Time, by virtue of the Merger and without any action on the part of the Parties or any of their respective shareholders:
 - (i) Each share of common stock of AcquireCo issued and outstanding immediately prior to the Merger Effective Time, and all rights in respect thereof, shall forthwith be cancelled and cease to exist and be converted into one fully paid and non-assessable share of common stock of the Surviving Company.
 - (ii) Each PLx Share that is owned by PLx as treasury stock and each PLx Share that is owned by PLx or any of its Subsidiaries immediately prior to the Merger Effective Time shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and no consideration shall be delivered in exchange therefor.
 - Effective Time shall be converted into the right to receive such number of DPRX Shares as is equal to the Equity Exchange Ratio and cash in lieu of any fractional shares of DPRX Shares to be issued or paid in consideration therefor (the "Merger Consideration") from DPRX, on behalf of AcquireCo. All such PLx Shares, when so converted, shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and each holder of a certificate (or evidence of shares in book-entry form) that immediately prior to the Merger Effective Time represented any such PLx Share (each, a "Certificate") shall cease to have any rights with respect thereto, except the right to receive the Merger Consideration. Notwithstanding the foregoing, if between the date of this Agreement and the Merger Effective Time, the outstanding DPRX Shares or PLx Shares shall have been changed into a different number of shares or a different class, by reason of any stock dividend, subdivision, reclassification, recapitalization, split, combination or exchange of shares, or any similar event shall have occurred, then any number or amount contained herein which is based upon the number of DPRX Shares or PLx Shares, as the case may be, will be appropriately adjusted to provide to PLx and the holders of PLx Shares the same economic effect as contemplated by this Agreement prior to such event.

- (g) The exchange of Certificates shall be effected as follows:
- (i) Prior to the Merger Effective Time, DPRX shall appoint a bank or trust company reasonably acceptable to PLx to act as exchange agent (the "Exchange Agent") for the payment and delivery of the Merger Consideration. At or prior to the Merger Effective Time, DPRX shall deposit with the Exchange Agent, for the benefit of the holders of Certificates, for exchange in accordance with this ARTICLE II through the Exchange Agent, on behalf of itself, certificates representing the DPRX Shares to be delivered as Merger Consideration (or, if uncertificated DPRX Shares will be delivered, DPRX shall make appropriate alternative arrangements) and cash sufficient to make payments in lieu of fractional shares in accordance with Section 2.1(m).
- (ii) As promptly as reasonably practicable after the Merger Effective Time (and in any event within four Business Days after the Merger Effective Time), DPRX shall cause the Exchange Agent to mail to each holder of record of PLx Shares a form of letter of transmittal (the "Letter of Transmittal") which shall specify that delivery shall be effected, and risk of loss and title to the Certificates shall pass, only upon delivery of the Certificates to the Exchange Agent, shall be in such form and have such other provisions (including customary provisions with respect to delivery of an "agent's message" with respect to shares held in book-entry form) as PLx may specify acting reasonably, and shall be prepared prior to the Closing, together with instructions thereto.
- Upon (i) in the case of PLx Shares represented by a Certificate, the surrender of such Certificate (iii) for cancellation to the Exchange Agent, or (ii) in the case of PLx Shares held in book-entry form, the receipt of an "agent's message" by the Exchange Agent, in each case together with the Letter of Transmittal, duly, completely and validly executed in accordance with the instructions thereto, and such other documents as may reasonably be required by the Exchange Agent, the holder of such shares shall be entitled to receive in exchange therefor the DPRX Shares into which such PLx Shares have been converted pursuant to Section 2.1(f) (and cash in lieu of any fractional DPRX Share pursuant to Section 2.1(m)). In the event of a transfer of ownership of PLx Shares that is not registered in the transfer records of PLx, the DPRX Shares may be delivered to a transferee if the Certificate representing such PLx Share (or, if such PLx Share is held in book-entry form, proper evidence of such transfer) is presented to the Exchange Agent, accompanied by all documents required to evidence and effect such transfer and by evidence that any applicable stock transfer Taxes have been paid. Until surrendered as contemplated by this Section 2.1(g), each PLx Share, and any Certificate with respect thereto, shall be deemed at any time from and after the Merger Effective Time to represent only the right to receive upon such surrender the Merger Consideration that the holders of PLx Shares are entitled to receive in respect of such shares pursuant to Section 2.1(f) (and cash in lieu of any fractional DPRX Share pursuant to Section 2.1(m)).

- (iv) The DPRX Shares delivered and credited as fully paid in accordance with the terms of this ARTICLE II upon conversion of any PLx Shares shall be deemed to have been delivered and paid in full satisfaction of all rights pertaining to such PLx Shares. From and after the Merger Effective Time, there shall be no further registration of transfers on the stock transfer books of the Surviving Company of PLx Shares that were outstanding immediately prior to the Merger Effective Time. If, after the Merger Effective Time, any Certificates formerly representing PLx Shares (or PLx Shares held in book-entry form) are presented to DPRX or the Exchange Agent for any reason, they shall be cancelled and exchanged as provided in this ARTICLE II.
- (v) Any portion of the Merger Consideration that remains undistributed to the holders of PLx Shares for one year after the Merger Effective Time shall be delivered to DPRX or its designee, and any holder of PLx Shares who has not theretofore complied with this ARTICLE II shall thereafter look only to DPRX for its claim for DPRX Shares.
- (vi) None of PLx, DPRX, AcquireCo, the Surviving Company or the Exchange Agent or any of their respective Affiliates shall be liable to any Person in respect of any portion of the Merger Consideration delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.
- (h) Each of DPRX and the Exchange Agent (without duplication) shall be entitled to deduct and withhold from the consideration otherwise payable to any holder of PLx Shares pursuant to this Agreement such amounts as are required to be deducted and withheld with respect to the making of such payment under applicable Tax Law. Amounts so withheld and paid over to the appropriate taxing authority shall be treated as having been paid to the holder of PLx Shares in respect of which such deduction or withholding was made.
- (i) If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by DPRX, the posting by such Person of a bond, in such reasonable and customary amount as DPRX may direct, as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent shall, in exchange for such lost, stolen or destroyed Certificate, issue the Merger Consideration deliverable in respect thereof pursuant to this Agreement.
- (j) As soon as practicable following the date of this Agreement and, in any event, prior to the Closing Date, the PLx Board of Directors or an appropriate committee thereof shall adopt such resolutions or take such other actions (including obtaining any required consents and making any required amendments to the PLx Stock Plan) as may be required to effect and/or procure the following:

(i) Each PLx Option that as of the Merger Effective Time is outstanding, shall cease to represent an option or other right to acquire PLx Shares and shall be converted on substantially the same terms and conditions as were applicable under the PLx Option (including vesting conditions, but taking into account any changes thereto provided for in the PLx Stock Plan, in any applicable award agreement, in such option or deemed necessary to comply with applicable Laws (including appropriate adjustments to performance vesting metrics, as applicable)) as of the Merger Effective Time into a stock option to acquire a number of DPRX Shares (rounded up to the nearest whole share) equal to the product of (i) the total number of PLx Shares subject to such PLx Option immediately prior to the Merger Effective Time multiplied by (ii) the Equity Exchange Ratio, at an exercise price per DPRX Share (rounded up to the nearest whole cent) equal to (x) the exercise price per PLx Share applicable to such PLx Option immediately prior to the Merger Effective Time divided by (y) the Equity Exchange Ratio;

(k) The treatment of PLx Stock Plan shall be as follows:

- (i) It is the intent of the Parties hereto that the treatment of PLx Options contemplated herein be in a manner that is consistent with the requirements of Section 409A of the Code, including all guidance and regulations issued thereunder.
- (ii) DPRX shall, prior to Closing, take all corporate action reasonably necessary to reserve for issuance a sufficient number of DPRX Shares as is equal to the aggregate number of DPRX Shares issuable after the Merger Effective Time upon exercise of the Replacement PLx Options, including seeking the approval, adoption, and filing of the DPRX Charter Amendments.
- As of the Merger Effective Time, DPRX will assume the PLx Stock Plan. As of the Merger Effective Time and assuming the approval, adoption and filing of the DPRX Charter Amendments, DPRX will be able to grant stock awards and options to purchase DPRX Shares, to the extent permissible by applicable Laws and NASDAQ regulations, under the terms of the PLx Stock Plan and issue the reserved but unissued PLx Shares (including shares subject to the unexercised or unissued portions of any PLx Options that expire, terminate or are canceled and shares subject to any PLx Option that are reacquired pursuant to the terms of the agreements under which such shares were issued that return to the PLx Stock Plan pursuant to their terms), except that (i) PLx Shares covered by such awards will be DPRX Shares and (ii) all references to a number of PLx Shares will be (A) changed to reference DPRX Shares and (B) converted to a number of DPRX Shares equal to the applicable number of PLx Shares multiplied by the Equity Exchange Ratio, rounded down to the nearest whole number of DPRX Shares. As soon as reasonably practicable following the date of this Agreement, and in any event prior to the Merger Effective Time (subject to the approval, adoption and filing of the DPRX Charter Amendments), the board of directors of DPRX (or, if appropriate, any committee administering employee, individual consultant and director compensation plans) shall adopt such resolutions and take such other actions as may be reasonably required to assume the PLx Stock Plan or to adopt share plans having terms substantially identical to the PLx Stock Plan and covering the awards of DPRX Shares resulting from Section 2.1(j), subject to any adjustments that may be required by applicable Laws. The DPRX Stock Plan (and all DPRX securities issued thereunder) shall continue in full force and effect in accordance with their respective terms.

- (l) It is intended that, for U.S. federal income tax purposes, (i) the Merger shall qualify as a reorganization within the meaning of Section 368(a) of the Code and (ii) this Agreement is hereby adopted as a "plan of reorganization" for purposes of Sections 354 and 361 of the Code.
- (m) No fraction of a share of DPRX Shares will be issued in connection with the Merger, and no certificates or scrip for any such fractional shares will be issued. PLx Stockholders will not be entitled to any voting rights, rights to receive any dividends or distributions or other rights as a stockholder of DPRX with respect to any such fraction of a share that would have otherwise been issued to such PLx Stockholder. Any PLx Stockholder who would otherwise be entitled to receive a fraction of a share of DPRX Shares (after aggregating all fractional shares of DPRX Shares issuable to such holder) will, in lieu of such fraction of a share and upon surrender of such holders' Certificate(s), be paid in cash the dollar amount (rounded to the nearest whole cent), without interest, determined by multiplying such fraction by the average of the closing sale price of DPRX Shares as quoted on The NASDAQ Capital Market for the ten (10) consecutive trading days ending with the trading day immediately preceding the date of the signing of this Agreement (as adjusted pursuant to Section 2.1(f)(iii) above).
- (n) For purposes of this Agreement, "Dissenting Shares" mean any PLx Shares outstanding immediately prior to the Merger Effective Time and held by a person who has not voted such shares in favor of the adoption of this Agreement and the Merger, has properly demanded appraisal for such shares in accordance with Delaware Law and has not effectively withdrawn or forfeited such demand for appraisal. Notwithstanding anything to the contrary contained herein, Dissenting Shares will not be converted into a right to receive the Merger Consideration unless such holder fails to perfect or withdraws or otherwise loses its rights to appraisal or it is determined that such holder does not have appraisal rights in accordance with Delaware Law. If after the Merger Effective Time, such holder fails to perfect or withdraws or loses its right to appraisal, or if it is determined that such holder does not have appraisal rights, such shares will be treated as if they had been converted as of the Merger Effective Time into the right to receive the merger consideration set forth in Section 2.1(f)(iii) hereof (if any). PLx will give DPRX prompt notice of any demands received by PLx for appraisal of PLx Shares, withdrawals of such demands, and any other instruments that relate to such demands received by PLx. DPRX and PLx shall jointly participate in all negotiations and proceedings with respect to such demands except as limited by applicable Law. Neither DPRX nor PLx will, except with prior written consent of the other, make any payment with respect to, or settle or offer to settle, any such demands, unless and to the extent required to do so under applicable Law.

The Closing. The closing (the "Closing") of the Merger shall take place at the offices of Jackson Walker L.L.P., located at 2323 Ross Ave., Suite 600, Dallas, Texas 75201, at 11:00 a.m. CST (or such other time and place as DPRX and PLx may mutually agree in writing) on the date (the "Closing Date") which shall be (i) the earlier of: (A) the date that is three Business Days after the satisfaction or waiver (subject to applicable Laws) of the conditions set forth in ARTICLE VIII (other than the satisfaction of those conditions); and (B) the date that is the day prior to the Outside Date; provided that the conditions set forth in ARTICLE VIII have been satisfied or waived as of such date; or (ii) such date as mutually agreed in writing by PLx and DPRX. Subject to the satisfaction or waiver (subject to applicable Laws) of the conditions (excluding conditions that, by their terms, cannot be satisfied until the Closing Date, but subject to the satisfaction or, where permitted, waiver of those conditions as of the Closing Date) set forth in ARTICLE VIII, the Merger shall, from and after the Merger Effective Time, have all of the effects provided under applicable Laws.

2.3 <u>Preparation of Joint Proxy Statement and Registration Statements.</u>

- (a) As promptly as reasonably practicable following the date hereof, each of the Parties shall cooperate in preparing and shall cause to be filed with the SEC (and, if applicable, any other Governmental Authority) (i) mutually acceptable proxy materials which shall constitute (A) the proxy statement relating to the matters to be submitted to the DPRX Stockholders at the DPRX Meeting, and (B) the proxy statement relating to the matters to be submitted to PLx Stockholders at the PLx Meeting (such joint proxy statement, and any amendments or supplements thereto, the "Joint Proxy Statement") and (ii) a registration statement on Form S-4 (of which the Joint Proxy Statement will form a part) with respect to the issuance of DPRX Shares in respect of the Merger (the "Form S-4").
- (b) Each Party will provide legal counsel to the other Party with a reasonable opportunity to review and comment on drafts of the Joint Proxy Statement, Form S-4 and other documents related to the DPRX Meeting or PLx Meeting, as applicable, prior to filing such documents with applicable Governmental Authorities and mailing such documents to the DPRX Stockholders or PLx Stockholders, as applicable. Each Party will include in the Joint Proxy Statement, Form S-4 or such other documents all comments reasonably and promptly proposed by the other Party or its legal counsel, *provided*, *however*, that all information relating to PLx and its Subsidiaries included in the Joint Proxy Statement shall be in form and content satisfactory to PLx, acting reasonably, and all information relating to DPRX and its Subsidiaries included in the Joint Proxy Statement shall be in form and content satisfactory to DPRX, acting reasonably.
- (c) Each Party shall use all commercially reasonable efforts to have the Joint Proxy Statement cleared by the SEC (and, if applicable, any other Governmental Authority), the Form S-4 to be declared effective by the SEC (and, if applicable, any other Governmental Authority) and to keep the Form S-4 effective as long as is necessary to consummate the Merger. As promptly as practicable after such clearance, DPRX and PLx shall, unless otherwise agreed to by the Parties, cause the Joint Proxy Statement and other documentation required in connection with the DPRX Meeting and the PLx Meeting to be sent contemporaneously to (x) in the case of DPRX, each DPRX Stockholder and (y) in the case of PLx, each PLx Stockholder, as required by applicable Laws. Each Party shall, as promptly as practicable after receipt thereof, provide the other Party with copies of any written comments and advise the other Party of any oral comments with respect to the Joint Proxy Statement or the Form S-4 received from the SEC.

- (d) Each Party shall use its commercially reasonable efforts to ensure that the Joint Proxy Statement complies in all material respects with applicable Laws. Each Party shall cooperate and provide the other Party with a reasonable opportunity to review and comment on any amendment or supplement to the Joint Proxy Statement or the Form S-4 prior to filing such documents with the SEC.
- (e) DPRX shall advise PLx, promptly after it receives notice thereof, of the time when the Form S-4 has become effective, the issuance of any stop order, the suspension of the qualification of the DPRX Shares issuable in connection with the Merger for offering or sale in any jurisdiction, or any request by the SEC (or, if applicable, any other Governmental Authority) for amendment of the Joint Proxy Statement or the Form S-4.
- (f) If, at any time prior to the Closing, any information relating to any of the Parties, or their respective Affiliates, officers or directors, should be discovered by any Party, and such information should be set forth in an amendment or supplement to the Joint Proxy Statement or the Form S-4 so that such documents would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the Party that discovers such information shall promptly notify the other Parties and, to the extent required by Law an appropriate amendment or supplement describing such information shall be promptly filed by DPRX and PLx with the SEC (or, if applicable, any other Governmental Authority) and, to the extent required by Law, disseminated to the DPRX Stockholders and PLx Stockholders, as applicable.
 - (g) The Joint Proxy Statement shall include:
 - (i) unless DPRX shall have effected a DPRX Change of Recommendation in accordance with the terms of this Agreement, the DPRX Recommendation, a copy of the DPRX Fairness Opinion and the rationale for the DPRX Recommendation; and
 - (ii) the PLx Recommendation and the rationale for the PLx Recommendation.
- (h) As promptly as practicable after the Closing Date, but in any event within 30 days thereafter, DPRX shall file with the SEC a registration statement on Form S-8 (or other applicable form) (the "Form S-8") in order to register under the Securities Act the DPRX Shares to be issued, offered or sold, as applicable, from time to time after the Merger Effective Time upon exercise of the DPRX Options and Replacement PLx Options.

2.4 <u>Shareholder Meetings</u>.

- (a) PLx shall duly take all lawful action to call, give notice of, convene and hold the PLx Meeting in accordance the governing documents of PLx and applicable Law as promptly as practicable following the date upon which the Form S-4 becomes effective for the purpose of obtaining PLx Stockholder Approval as required by the DGCL and this Agreement.
- (b) DPRX shall duly take all lawful action to call, give notice of, convene and hold the DPRX Meeting in accordance the governing documents of DPRX and applicable Law, as promptly as practicable following the date upon which the Form S-4 becomes effective for the purpose of obtaining the DPRX Stockholder Approval in accordance with the applicable Laws and this Agreement.

- (c) Subject to the terms of this Agreement, PLx shall use its commercially reasonable efforts to solicit from PLx Stockholders proxies in favor of the PLx Stockholder Approval and take all other actions that are reasonably necessary or desirable to obtain the approval of the Merger and this Agreement by PLx Stockholders, and take all other actions reasonably requested by DPRX that are reasonably necessary to obtain PLx Stockholder Approval and permit DPRX to assist, and consult with DPRX and keep DPRX apprised, with respect to such solicitation and other actions. Unless this Agreement has been terminated in accordance with ARTICLE VII, subject to Section 2.4(a) this Agreement shall be submitted to PLx Stockholders at the PLx Meeting for the purpose of obtaining PLx Stockholder Approval, and nothing contained herein shall be deemed to relieve PLx of such obligation.
- (d) Subject to the terms of this Agreement (including Section 6.2), DPRX shall use its commercially reasonable efforts to solicit from the DPRX Stockholders proxies in favor of the approval of the DPRX Stockholder Resolution, and cooperate with any Persons engaged by PLx, to solicit proxies in favor of the approval of the DPRX Stockholder Resolution and take all other actions that are reasonably necessary to obtain the DPRX Stockholder Approval and permit PLx to assist, and consult with PLx and keep PLx apprised, with respect to such solicitation and other actions. Unless this Agreement has been terminated in accordance with ARTICLE VII, subject to Section 2.4(b), this Agreement shall be submitted to the DPRX Stockholders at the DPRX Meeting for the purpose of obtaining the DPRX Stockholder Approval, and nothing contained herein shall be deemed to relieve DPRX of such obligation.
- (e) Unless there has been a DPRX Change of Recommendation in accordance with Section 6.2, neither the DPRX Board of Directors nor any committee thereof shall withdraw (or modify in any manner adverse to PLx), or propose publicly to withdraw (or modify in any manner adverse to PLx), the DPRX Recommendation.
- (f) Neither the PLx Board of Directors nor any committee thereof shall withdraw (or modify in any manner adverse to DPRX), or propose publicly to withdraw (or modify in any manner adverse to DPRX), the PLx Recommendation.
- (g) PLx shall, prior to the PLx Meeting, keep DPRX reasonably informed of the number of proxy votes received in respect of matters to be acted upon at PLx Meeting, and in any event shall provide such number promptly upon the request of DPRX or its Representatives.
- (h) DPRX shall, prior to the DPRX Meeting, keep PLx reasonably informed of the number of proxy votes received in respect of matters to be acted upon at the DPRX Meeting, and in any event shall provide such number promptly upon the request of PLx or its Representatives.
- (i) Subject to the terms of this Agreement, PLx shall use commercially reasonable efforts to ensure that the PLx Meeting will occur no more than two Business Days following the DPRX Meeting. Each of DPRX and PLx shall not adjourn, postpone, delay or cancel (or propose for adjournment, postponement, delay or cancellation) the DPRX Meeting or the PLx Meeting, as applicable, without the other Party's prior written consent, in each case; *provided*, that:

- (i) PLx shall be permitted to adjourn, delay or postpone convening the PLx Meeting if in the good faith judgment of the PLx Board of Directors (after consultation with its outside legal advisors) the failure to adjourn, delay or postpone the PLx Meeting could be reasonably likely to (A) be inconsistent with the fiduciary duties of the PLx Board of Directors under applicable Laws, (B) not allow sufficient time under applicable Laws for the distribution of any required or appropriate supplement or amendment to the Joint Proxy Statement or Form S-4 or (C) result in the failure to obtain the PLx Stockholder Approval; and
- (ii) DPRX shall be permitted to adjourn, delay or postpone convening the DPRX Meeting if in the good faith judgment of the DPRX Board of Directors (after consultation with its outside legal advisors) the failure to adjourn, delay or postpone the DPRX Meeting could be reasonably likely to (A) be inconsistent with the fiduciary duties of the DPRX Board of Directors under applicable Laws, (B) not allow sufficient time under applicable Laws for the distribution of any required or appropriate supplement or amendment to the Joint Proxy Statement or Form S-4 or (C) result in the failure to obtain the DPRX Stockholder Approval.
- (j) PLx and DPRX will each provide notice to the other of the PLx Meeting or the DPRX Meeting, respectively, and shall allow Representatives of the other and its counsel to attend the applicable meeting.
- 2.5 <u>Calculation of DPRX Closing Cash</u>. DPRX shall prepare and deliver to PLx three (3) Business Days prior to the Closing, a schedule (the "**DPRX Closing Cash Schedule**") setting forth, in reasonable detail, DPRX's good faith estimate of DPRX Closing Cash to be held by DPRX as of the Closing, together with the work papers and back-up materials used in preparing such DPRX Closing Cash Schedule.

ARTICLE III REPRESENTATIONS AND WARRANTIES

3.1 Representations and Warranties of DPRX. Except as disclosed in the applicable section or subsection of the DPRX Disclosure Letter (it being agreed that disclosure of any item in any section or subsection of the DPRX Disclosure Letter shall be deemed disclosure with respect to any other section or subsection of the DPRX Disclosure Letter only to the extent the relevance of such item to such other section or subsection is reasonably apparent on its face) or the DPRX Public Disclosure Record (other than any disclosure contained under the captions "Risk Factors" or "Forward-Looking Statements" or similar captions and any other disclosure contained therein that is predictive, cautionary or forward-looking in nature), DPRX represents and warrants to and in favor of PLx as follows:

- (a) Organization and Qualification. DPRX has been duly incorporated, validly exists and is in good standing under the DGCL and has the requisite corporate and legal power and capacity to own its assets as now owned and to carry on its business as it is now being carried on. Each of the DPRX Subsidiaries is a corporation or other entity duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, organization or formation and has the requisite corporate, legal or other power and authority to own its assets as now owned and to carry on its business as it is now being carried on. DPRX and each of the DPRX Subsidiaries is duly qualified to carry on business in each jurisdiction in which the nature or character of the respective properties and assets, owned, leased or operated by it, or the nature of its business or activities, makes such qualification necessary, except where the failure to be so qualified would not reasonably be expected to have a Material Adverse Effect on DPRX. DPRX has provided or otherwise made available to PLx true, complete and correct copies of the governing documents of each of DPRX and DPRX's Subsidiaries, in each case as amended.
- (b) Authority Relative to this Agreement. Each DPRX Party has the requisite corporate power, authority and capacity to enter into this Agreement and (subject to obtaining the DPRX Stockholder Approval, as contemplated in this Agreement) to perform its obligations hereunder and to complete the Transaction. The execution and delivery of this Agreement and the completion by each DPRX Party of the Transaction has been duly authorized by its respective board of directors and no other corporate proceedings on the part of any DPRX Party are necessary to authorize the execution and delivery by it of this Agreement or, subject to obtaining the DPRX Stockholder Approval as contemplated in this Agreement, to adoption of this Agreement by DPRX as sole stockholder of AcquireCo immediately following the execution hereof, and the filing and recordation of the Certificate of Merger pursuant to Delaware Law, the completion by any DPRX Party of the Transaction. This Agreement has been duly executed and delivered by each DPRX Party and constitutes a legal, valid and binding obligation of each DPRX Party enforceable against such DPRX Party in accordance with its terms, subject to bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and other Laws relating to limitations of actions or affecting the availability of equitable remedies and the enforcement of creditors' rights generally and general principles of equity.
- (c) Required Approvals. No authorization, license, Permit, certificate, registration, consent or approval of, or filing with, or notification to, any Governmental Authority is necessary for the execution and delivery of this Agreement, the performance by any DPRX Party of its obligations hereunder, the completion by the DPRX Parties of the Transaction, other than:
 - (i) such filings and other actions required under applicable U.S. Securities Laws and the rules and policies of NASDAQ, in each case, as are contemplated by this Agreement; and
 - (ii) any other authorizations, licenses, Permits, certificates, registrations, consents, approvals and filings and notifications with respect to which the failure to obtain or make the same would not reasonably be expected to have a Material Adverse Effect on DPRX, or could not reasonably be expected to prevent or significantly impede or materially delay the completion of the Merger.
- (d) *No Violation*. Subject to obtaining the authorizations, consents and approvals and making the filings referred to in Section 3.1(c) and complying with applicable Laws and Orders, the execution and delivery by each DPRX Party of this Agreement, the performance by such DPRX Party of its obligations hereunder and the completion of the Transaction do not and will not (nor will they with the giving of notice or the lapse of time or both):

- (i) to the knowledge of DPRX, result in a contravention, breach, violation or default under any Law or Order applicable to DPRX or any of the DPRX Subsidiaries or by which any of its or their respective properties or assets are bound or affected:
- (ii) result in a contravention, conflict, violation, breach or default under the governing documents of DPRX or any of the DPRX Subsidiaries;
- (iii) result in a contravention, breach or default under or termination of, or acceleration or permit the acceleration of the performance required by, or loss of any benefit under, any DPRX Material Contract or material Permit to which it or any of the DPRX Subsidiaries is a party or by which it or any of the DPRX Subsidiaries is bound or to which any of its or any of the DPRX Subsidiaries' properties or assets is subject or give to any Person any interest, benefit or right, including any right of purchase or sale, termination, payment, modification, reimbursement, penalty, cancellation or acceleration, under any such Material Contract or material Permit; or
- (iv) result in the suspension or alteration in the terms of any material Permit held by DPRX or any of the DPRX Subsidiaries or in the creation of any Lien upon any of their properties or assets;

except, in the case of each of clauses (i), (iii) and (iv) above, as would not reasonably be expected to have a Material Adverse Effect on DPRX or would not prevent or materially delay the consummation of the Merger.

- Capitalization of DPRX. The authorized capital of DPRX consists of 30,000,000 DPRX Shares. As at the date of this Agreement, there are (i) 10,401,122 DPRX Shares issued and outstanding, all of which have been duly authorized and validly issued and are fully paid and non-assessable and there are no preferred shares outstanding, (ii) 1,445,013 DPRX Options outstanding under the DPRX Stock Plan providing for the issuance of an aggregate of 1,445,013 DPRX Shares upon the exercise thereof in accordance with their respective terms, (iii) 12,250 DPRX Warrants outstanding providing for the issuance of an aggregate of 12,250 DPRX Shares upon the conversion thereof, (iv) 14,000 DPRX unvested common shares, and (iv) 696,156 additional DPRX Shares reserved for issuance under the Stock Option Plan. None of such 10,401,122 DPRX Shares, 1,445,013 DPRX Options or 12,250 DPRX Warrants are owned by DPRX or any Subsidiary of DPRX. There is no outstanding contractual obligation of DPRX or any of its Subsidiaries to repurchase, redeem or otherwise acquire any DPRX Shares. Except for the DPRX Options and DPRX Warrants, DPRX has no outstanding agreement, subscription, warrant, option, conversion or exchange privilege right, arrangement or commitment (nor has it granted any right or privilege (contingent or otherwise) capable of becoming an agreement, subscription, warrant, option, conversion or exchange privilege, right, arrangement or commitment) obligating it to issue or sell any DPRX Shares or other securities of DPRX, including any security or obligation of any kind convertible into or exchangeable or exercisable for any DPRX Shares or other security of DPRX. Except for the DPRX Options and DPRX Warrants, neither DPRX nor any of the DPRX Subsidiaries has outstanding any stock appreciation right, phantom equity, restricted share unit, deferred share unit or similar right, agreement, arrangement or commitment based on the book value, DPRX Share price, income or any other attribute of or related to DPRX or any of its Subsidiaries. The DPRX Shares are listed on NASDAQ and, except for such listings, no securities of DPRX or any of the DPRX Subsidiaries are listed on any other stock or securities exchange or market or registered under any securities Laws. There are no outstanding bonds, debentures or other evidences of indebtedness of DPRX or any of the DPRX Subsidiaries having the right to vote (or that are convertible into or exchangeable or exercisable for securities having the right to vote) with the holders of DPRX Shares on any matter. Section 3.1(e) of the DPRX Disclosure Letter sets out a true, complete and correct list of all DPRX Options and DPRX Warrants, the names of the holders thereof, and the grant date of such securities. A true, correct and complete copy of the DPRX Stock Plan has been provided or otherwise made available to PLx. All DPRX Shares issuable in connection with the Merger in accordance with the terms of this Agreement will, prior to the Closing Date, be duly authorized and, as of Closing, will be validly issued as fully paid and non-assessable and will not be subject to any pre-emptive rights. All DPRX Shares issuable upon exercise of the Replacement PLx Options or upon conversion of PLx Convertible Notes, will, prior to the Closing Date or as soon as practicable thereafter, be duly authorized and reserved for issuance and will, upon exercise of such securities or delivery of underlying DPRX Shares, as applicable, in accordance with their respective terms, be validly issued as fully paid and non-assessable and will not be subject to any pre-emptive rights.
- (f) DPRX Subsidiaries. Section 3.1(f) of the DPRX Disclosure Letter sets forth a true, complete and correct list of each of the DPRX Subsidiaries, its jurisdiction and form of organization. DPRX or a DPRX Subsidiary is the sole registered and beneficial owner of all of the outstanding shares in the capital of or outstanding shares of capital stock or other ownership, equity or voting interests of the DPRX Subsidiaries free and clear of any Liens (other than Permitted Liens), and no other Person has any option, right, entitlement, understanding or commitment (contingent or otherwise) regarding the right to acquire any such share or interest in any of the DPRX Subsidiaries and no outstanding option, warrant, conversion or exchange privilege or other right, agreement, arrangement or commitment obligating any such entity to issue or sell any share or ownership, equity or voting interest of such entity or security or obligation of any kind convertible into or exchangeable or exercisable for any shares or ownership, equity or voting interests of any such entity. Neither DPRX nor any of the DPRX Subsidiaries own any interest or investment (whether equity or debt) in any other Person, other than a Subsidiary of DPRX, which interest or investment is material to DPRX and the DPRX Subsidiaries, taken as a whole.

(g) Securities Laws Matters.

(i) All documents in the DPRX Public Disclosure Record have been timely filed and, as of the time a document in the DPRX Public Disclosure Record was filed with the SEC (or, if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing): (i) each of the documents in the DPRX Public Disclosure Record complied in all material respects with the applicable requirements of the Exchange Act and (ii) none of the documents in the DPRX Public Disclosure Record contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading. As used in this Section 3.1(g), the term "file" and variations thereof will be broadly construed to include any manner in which a document or information is furnished, supplied or otherwise made available to the SEC.

- (ii) DPRX and its Subsidiaries maintain disclosure controls and procedures required by Rule 13a-15 or 15d-15 under the Exchange Act. Such disclosure controls and procedures are designed to ensure that all material information concerning DPRX required to be disclosed by DPRX in the reports that it is required to file, submit or furnish under the Exchange Act is recorded, processed, summarized and reported on a timely basis to the individuals responsible for the preparation of such reports.
- (h) Financial Statements. The DPRX Financial Statements have been prepared in accordance with U.S. GAAP applied on a basis consistent with those of previous periods and in accordance with applicable Laws except as otherwise stated in the notes to such statements or in the auditor's report thereon and subject, in the case of the DPRX Interim Financial Statements, to normal year-end audit adjustments, which are not material to DPRX and the DPRX Subsidiaries, taken as a whole, individually or in the aggregate, and may omit notes which are not material and are not required by applicable Laws or U.S. GAAP. The DPRX Financial Statements present fairly, in all material respects, the consolidated financial position and consolidated results of operations, changes in shareholders' equity and cash flows of DPRX and the DPRX Subsidiaries as of the respective dates thereof and for the respective periods set forth therein. There are no outstanding loans made by DPRX or any of the DPRX Subsidiaries to any director or officer of DPRX. All documents in the DPRX Public Disclosure Record (including any financial statements included or incorporated by reference therein), as of their respective dates (and as of the date of any amendment to the respective document in the DPRX Public Disclosure Record), complied as to form in all material respects with the applicable requirements of the Securities Act and the Exchange Act.
- (i) No Undisclosed Liabilities. DPRX and the DPRX Subsidiaries have no liability or obligation of any nature (whether accrued, absolute, contingent or otherwise) that would be required to be disclosed on a balance sheet (or the footnotes thereto) prepared in accordance with U.S. GAAP, other than (i) liabilities and obligations disclosed in the DPRX Public Disclosure Record, (ii) liabilities and obligations incurred in the ordinary course of business since the date of the most recent DPRX Annual Financial Statements (other than those specifically disclosed in the DPRX Public Disclosure Record) that have not had and would not reasonably be expected to have, individually or in aggregate with all other liabilities and obligations of DPRX and the DPRX Subsidiaries (other than those disclosed in DPRX Public Disclosure Record), a Material Adverse Effect on DPRX, and (iii) liabilities and obligations incurred in connection with this Agreement and the Transaction. Without limiting anything set forth herein, the DPRX Financial Statements reflected and continued to reflect, in each case as of the date filed, appropriate reserves under U.S. GAAP for contingent liabilities relating to pending or anticipated litigation and other contingent obligations of DPRX and the DPRX Subsidiaries.
 - (i) [Reserved].

- (k) Compliance with Laws. Since December 31, 2013, the business of DPRX and of each of the DPRX Subsidiaries has been and is currently being conducted in material compliance with all applicable Laws, Orders and Regulatory Guidelines and neither DPRX nor any DPRX Subsidiary has received any written notice of any alleged material non-compliance or violation of any such Laws, Orders or Regulatory Guidelines. Neither DPRX nor any of the DPRX Subsidiaries has knowingly taken or committed to take any action which would cause DPRX or any of the DPRX Subsidiaries to be in violation of the United States Foreign Corrupt Practices Act, or any applicable Law of similar effect, and, to the knowledge of DPRX, no such action has been taken by any Person acting on behalf of DPRX or any of the DPRX Subsidiaries.
- (l) Litigation. Except as set forth in Section 3.1(l) of the DPRX Disclosure Letter, there is no Proceeding against or involving DPRX or any of the DPRX Subsidiaries (whether in progress, pending or, to the knowledge of DPRX, threatened), and no event or circumstance has occurred which would reasonably be expected to give rise to any such Proceeding. Neither DPRX nor any of the DPRX Subsidiaries nor any of their respective properties or assets is subject to any outstanding Order that would reasonably be expected to (i) prevent or significantly impede or materially delay the completion of the Merger or (ii) have a Material Adverse Effect on DPRX.
- (m) Real Property. Section 3.1(m) of the DPRX Disclosure Letter contains a list of all leases pursuant to which DPRX or any DPRX Subsidiary currently leases real property as tenant. Neither DPRX nor any of the DPRX Subsidiaries owns any real property.
 - (n) [Reserved].
 - (o) Contracts.
 - (i) Except as set forth in the DPRX Public Disclosure Record or in Section 3.1(o) of the DPRX Disclosure Letter, as of the date of this Agreement, none of DPRX or any of the DPRX Subsidiaries is a party to or bound by any of the following types of Contract (each of the following types of Contracts, a "**DPRX Material Contract**"):
 - (A) any collective bargaining agreement, or similar Contract with any labor union or association, with respect to its employees, and any Contract with any officer, employee, consultant or director;
 - (B) any Contract entered into outside of the ordinary course of business that is not terminable by DPRX or any of the DPRX Subsidiaries on three months' notice or less;
 - (C) any credit agreement, loan agreement, indenture, note, mortgage, security agreement, loan commitment or other Contract relating to the indebtedness of DPRX or any DPRX Subsidiary;
 - (D) any Contract granting to any Person a right of first refusal or option to purchase or acquire any assets of DPRX or any DPRX Subsidiary;

- (E) any real property lease, rental or occupancy agreement under which DPRX or any DPRX Subsidiary continues to have obligations or rights;
- (F) any Contract entered into outside of the ordinary course of business pursuant to which DPRX or any DPRX Subsidiary (i) is granted or obtains or agrees to obtain any right to use any material Intellectual Property (excluding commercially available software), (ii) is restricted in its right to use or register any material Intellectual Property owned by DPRX or any of the DPRX Subsidiaries, or (iii) permits or agrees to permit any other Person, to use, obtain, enforce or register any material Intellectual Property owned by DPRX or any of the DPRX Subsidiaries, including any license agreements, option agreements, and covenants not to sue:
- (G) except for any non-solicit obligations, any Contract that obligates DPRX or any DPRX Subsidiary or its Affiliates not to compete with another Person, requires DPRX or any DPRX Subsidiary to acquire any product, assets or service exclusively from any other Person, or otherwise contractually restricts DPRX or any DPRX Subsidiary or its Affiliates from acquiring any material product, asset or service from any other Person, or providing products, assets or services to any other Person, or developing or distributing any product to any Person or in any geographic location;
- (H) any Contract entered into since December 31, 2013: (i) relating to the merger, consolidation, reorganization, liquidation, dissolution or any similar extraordinary transaction with respect to DPRX or any DPRX Subsidiary, (ii) relating to a material acquisition or disposition by a DPRX or any DPRX Subsidiary, (iii) relating to the acquisition, issuance or transfer of any securities of DPRX or any DPRX Subsidiary or (iv) relating to any partnership, strategic alliance or joint venture agreement; and
- (I) except for Contracts entered into in the ordinary course of business with any employee, director or officer of DPRX or any DPRX Subsidiary, any Contract with any shareholder of DPRX or any DPRX Subsidiary entered into since December 31, 2013.
- (ii) True, correct and complete copies of each DPRX Material Contract in effect on the date hereof that has not been part of the DPRX Public Disclosure Record has been provided or otherwise made available to PLx.

- (iii) Except as would not reasonably be expected to have a Material Adverse Effect on DPRX, none of DPRX, the DPRX Subsidiaries or, to the knowledge of DPRX, any of the other parties thereto, is in breach or violation of or in default under, or committed or failed to perform any act which would result in a default under, (in each case, with or without notice or lapse of time or both) any DPRX Material Contract in any material respect, and none of DPRX or any of the DPRX Subsidiaries has received or given any notice of default under any DPRX Material Contract which remains uncured. To the knowledge of DPRX, there exists no state of facts which after notice or lapse of time or both would constitute a default under or breach or violation of any DPRX Material Contract or the inability of a party to any DPRX Material Contract to perform its obligations thereunder where, in any such case, such default, breach, violation or non-performance has had or would reasonably be expected to have a Material Adverse Effect on DPRX. To the knowledge of DPRX, no Person has challenged in writing the validity or enforceability of any DPRX Material Contract.
- (iv) Other than pursuant to Voting Agreements with the DPRX Specified Stockholders and except as set forth in Section 3.1(o) of the DPRX Disclosure Letter, there are no shareholders or stockholders agreements, registration rights agreements, voting trusts, proxies or similar agreements, arrangements or commitments to which DPRX or any of the DPRX Subsidiaries is a party or, to the knowledge of DPRX, with respect to any shares or other equity interests of DPRX or any of the DPRX Subsidiaries or any other Contract relating to disposition, voting or dividends with respect to any shares or other equity securities of DPRX or any of the DPRX Subsidiaries.
- (v) As of the date of this Agreement, neither DPRX nor any of the DPRX Subsidiaries has received written notice of the termination of, or intent to terminate or otherwise fail to materially perform any DPRX Material Contract.
- (p) Taxes.
- (i) DPRX and each of its Subsidiaries has duly and timely made or prepared all income and other material Returns required to be made or prepared by it, has duly and timely filed all income and other material Returns required to be filed by it with the appropriate Governmental Authority and has completely and correctly reported all material income and all other amounts or information required to be reported thereon. All material Returns provided or otherwise made available to PLx are true, complete and correct copies of such Returns.
- (ii) DPRX and each of the DPRX Subsidiaries has: (A) duly and timely paid all material Taxes due and payable by it; (B) duly and timely withheld all material Taxes and other material amounts required by applicable Laws to be withheld by it and has duly and timely remitted to the appropriate Governmental Authority such material Taxes and other material amounts required by applicable Laws to be remitted by it; and (C) duly and timely collected all material amounts on account of sales or transfer taxes, including goods and services, harmonized, sales, value added and federal, provincial, state or territorial sales taxes, required by applicable Laws to be collected by it and has duly and timely remitted to the appropriate Governmental Authority any such material amounts required by applicable Laws to be remitted by it.

- (iii) No audit, action, investigation, deficiency, litigation, proposed adjustment or other Proceeding exists or has been asserted in writing or, to the knowledge of DPRX, threatened in writing with respect to income or other material Taxes or income or other material Returns of DPRX or any of its Subsidiaries, and neither DPRX nor any of its Subsidiaries is a party to any Proceeding for assessment, reassessment, or collection of material Taxes and no such Proceeding has been asserted in writing or, to the knowledge of DPRX, threatened in writing against DPRX or any of its Subsidiaries or any of their respective assets, and there are no matters of dispute or matters under discussion with any Governmental Authority relating to income or other material Taxes assessed by any Governmental Authority against DPRX or any of its Subsidiaries or relating to income or other material Returns. No claim has ever been made in writing by any Governmental Authority in a jurisdiction where DPRX or any of its Subsidiaries does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.
- (iv) There are no currently effective or pending material elections, agreements, or waivers extending the limitation period or providing for an extension of time with respect to the assessment or reassessment of any income or other material Taxes, the filing of any income or other material Return (excluding, for the avoidance of doubt, ordinary course extensions within which to file tax returns), or the payment of any income or other material Taxes by DPRX or any of its Subsidiaries.
- (v) There are no Liens for income or other material Taxes on the property or assets of DPRX or any of its Subsidiaries, except for statutory liens for Taxes not yet due and payable.
- (vi) Neither DPRX nor any Subsidiary thereof has ever been a member of any consolidated, unitary, combined or similar group for any income or other materials Tax Purposes. Neither DPRX nor any Subsidiary there is a party to any Tax allocation, sharing, indemnity, or reimbursement agreement or arrangement (other than any commercial agreement or arrangement a principal purpose of which does not pertain to Tax). Neither DPRX nor any Subsidiary thereof has any liability for the Taxes of any other Person under law, as a transferee or successor, by contract (other than any commercial agreement or arrangement a principal purpose of which does not pertain to Tax) or otherwise.
- (vii) No closing agreement, private letter ruling or similar agreements or rulings concerning Taxes has been entered into, issued or requested by any Governmental Authority with respect to DPRX or any of the DPRX Subsidiaries for any taxable year for which the limitation period has not yet expired.
- (viii) The charges, accruals, and reserves for Taxes reflected on the DPRX Interim Financial Statements (whether or not due and whether or not shown on any Return but excluding any provision for deferred income taxes) are adequate under GAAP to cover Taxes with respect to DPRX and each of its Subsidiaries accruing through the date hereof.
- (ix) DPRX is, and at all times since March 2014 has been, treated as a corporation for U.S. federal income tax purposes.

- (x) Neither DPRX nor any of its Subsidiaries will be required to include any material item of income or exclude any material item of deduction for any taxable period (or portion thereof) beginning after the Closing as a result of (i) any "closing agreement," as described in Section 7121 of the Code (or any corresponding provision of state, local or foreign Law) entered into by DPRX or any of its Subsidiaries on or before the Closing Date, (ii) any installment sale or open transaction disposition by DPRX or any of its Subsidiaries that occurred before the Closing Date, (iii) a change in method of accounting or use of an improper method of accounting with respect to DPRX or any of its Subsidiaries for a taxable period ending on or before the Closing Date, (iv) an election under Section 108(i) of the Code by DPRX or any of its Subsidiaries on or before the Closing Date, or (iv) the receipt of any prepaid revenue by a by DPRX or any of its Subsidiaries before the Closing Date.
- (xi) (x) Neither DPRX nor any of its Subsidiaries is or has been a party to any "listed transaction," as defined in Treasury Regulations Section 1.6011-4(b)(2).
- (xii) Neither DPRX nor any of its Subsidiaries has distributed equity interests of another Person, or has had its equity interests distributed by another Person, during the last two (2) years in a transaction that was purported or intended to be governed in whole or in part by Sections 355 or 361 of the Code.
- (q) Employment Agreements and Collective Agreements. Except as set forth in the DPRX Public Disclosure Record, none of DPRX or any of the DPRX Subsidiaries are parties to or bound or governed by (or currently negotiating in connection with entering into), or subject to, or has any liability with respect to:
 - (i) any employment, retention or change of control agreement with, or any written or oral agreement, commitment, obligation, arrangement, plan or understanding providing for any retention, bonus, severance, change of control, retirement or termination payments to any current or, to the extent remaining outstanding, former director, officer or employee of DPRX or any of DPRX's Subsidiaries (each, a "DPRX Employment Agreement");
 - (ii) any collective bargaining or union agreements or other Contract with a labor union, labor organization or employee association, or any actual or, to the knowledge of DPRX, threatened application for certification, recognition or bargaining rights in respect of DPRX or any of the DPRX Subsidiaries, or any Proceeding seeking to compel DPRX or any of the DPRX Subsidiaries to bargain with any labor organization as to wages or conditions of employment;
 - (iii) any organized labor dispute, work stoppage or slowdown, strike or lock-out relating to or involving any employees of DPRX or any of the DPRX Subsidiaries; or

(iv) any actual or, to the knowledge of DPRX, threatened grievance, claim or other Proceeding arising out of or in connection with any labor or employment matter by DPRX or any of the DPRX Subsidiaries or the termination thereof except as would not be expected to have a Material Adverse Effect on DPRX.

True, complete and correct copies of the agreements, arrangements, plans and understandings referred to in paragraphs (i) and (ii) of this Section 3.1(q) have been provided or otherwise made available to PLx. Except as would not be expected to have a Material Adverse Effect on DPRX, each of DPRX and the DPRX Subsidiaries is in material compliance with all applicable Laws (domestic and foreign), Orders, Contracts and DPRX material policies relating to employment, employment practices, wages, hours and terms and conditions of employment.

(r) Pension and Employee Benefits.

- (i) Section 3.1(r)(i) of the DPRX Disclosure Letter sets forth a true, complete and correct list of each employee benefit and compensation plan, agreement, program or arrangement, whether written or unwritten, including without limitation, any option, restricted share unit, deferred share unit, stock purchase, or other stock or stock-based incentive plan, cash bonus or incentive compensation arrangement, retirement or deferred compensation plan, profit sharing plan, unemployment or severance compensation plan or health and welfare plan, or DPRX Employment Agreement, for any current or former employee or director, to the extent the potential liability remains outstanding, of, or other service provider to, DPRX or any of its Subsidiaries participates in, is a party or contributes to, or with respect to which DPRX or any of its Subsidiaries could reasonably be expected to have any liability (each, a "**DPRX Plan**").
- (ii) With respect to each DPRX Plan, DPRX has provided or otherwise made available to PLx (A) a true and complete copy of each DPRX Plan, including any amendments thereto and all material supporting documents; (B) latest annual report, if any; (C) copies of all material communications received in the last three years with applicable Government Authority; (D) each trust or other funding arrangement, (E) each summary plan description (if applicable) and (F) where applicable, the most recent financial statements and actuarial or other valuation reports prepared with respect thereto.
- (iii) The consummation of the transactions contemplated by this Agreement will not, either alone or in combination with another event, (A) entitle any current or former employee or officer of DPRX to termination or severance pay, unemployment compensation or any other payment, (B) accelerate the time of funding (through a grantor trust or otherwise), payment or vesting, or increase the amount of compensation or benefit due any such employee or officer, or (C) cause amounts payable under the DPRX Plans to fail to be deductible for U.S. federal income tax purposes by virtue of Section 280G of the Code. No employee or individual consultant or independent contractor is entitled to receive any gross- up or additional payment by reason of the tax required by Section 409A or 4999 of the Code being imposed upon such person.

- (iv) Each DPRX Plan has been established, registered, qualified, funded, invested, operated and administered in all material respects in accordance with its terms and applicable Law (including Section 409A of the Code). There are no pending, or to the knowledge of DPRX, threatened actions, suits, disputes or claims by or on behalf of any DPRX Plan, by any employee or beneficiary covered under any such DPRX Plan, as applicable, or otherwise involving any such DPRX Plan (other than routine claims for benefits).
- (v) No DPRX Plan provides welfare or post-retirement benefits, including without limitation, death or medical benefits (whether or not insured), beyond retirement or termination of service to employees or former employees or to the beneficiaries or dependents of such employees, other than coverage mandated solely by applicable Law.
- (vi) No DPRX Plan is governed by, and DPRX has no liability under, Section 401(a) of the Code or US Employee Retirement Income Security Act of 1974, as amended ("ERISA"). Neither DPRX, nor any Person that is a member of a "controlled group of corporations" with, or is under "common control" with, or is a member of the same "affiliated service group", with DPRX, in each case as defined in Sections 414(b), (c), (m) or (o) of the Code sponsors, contributes to or has any liability under, and in the past six years sponsored, contributed to or had liability under, a plan subject to Title IV or Section 302 of ERISA.
- (vii) There has been no amendment to, written interpretation or announcement (whether or not written) by DPRX or any of its Subsidiaries relating to, or change in employee participation or coverage under, a DPRX Plan which would increase materially the expense of maintaining such DPRX Plan above the level of the expense incurred in respect thereof for the fiscal year ended December 31, 2015. There has been no termination of any material DPRX Plan since January 1, 2016.
- (viii) All contributions, premiums or Taxes required to be made or paid by DPRX or any of its Subsidiaries, as the case may be, under or in connection with the DPRX Plans have been made in a timely fashion in accordance with Laws and the terms of the applicable DPRX Plan. There are no unfunded liabilities in respect of any DPRX Plan and have been properly reflected in the DPRX Financial Statements.

(s) *Intellectual Property*.

(i) Section 3.1(s)(i) of the DPRX Disclosure Letter sets forth a correct and complete list of all (A) issued Patents and Patent applications, (B) Trademark registrations and applications and material unregistered Trademarks, (C) Copyright registrations and applications, and (D) material Software, in each case which is owned or exclusively licensed by DPRX and the DPRX Subsidiaries in any jurisdiction in the world (collectively, "DPRX Intellectual Property"). DPRX or one of the DPRX Subsidiaries is the sole and exclusive beneficial and, with respect to applications and registrations (including Patents), record owner or exclusive licensee of the record owner of each item of DPRX Intellectual Property set forth in Section 3.1(s)(i) of the DPRX Disclosure Letter, and, except as could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on DPRX, to the knowledge of DPRX, all such Intellectual Property is subsisting, valid, and enforceable. All required filings and fees related to the (A) issued Patents and Patent applications, (B) Trademark registrations and applications and material unregistered Trademarks, (C) Copyright registrations and applications, and (D) material Software, in each case which is owned or exclusively licensed by DPRX and the DPRX Subsidiaries have been timely filed with and paid to the relevant Governmental Authorities and authorized registrars.

- (ii) DPRX or one of the DPRX Subsidiaries owns, or has a valid right to use, free and clear of all Liens (other than Permitted Liens), all Intellectual Property (A) related to the products or product candidates presently used in the conduct of the business of DPRX or one of the DPRX Subsidiaries and (B) used or held for use in, or necessary to conduct, the business and operations of DPRX and the DPRX Subsidiaries as presently conducted.
- (iii) There are no Orders, writs, injunctions or decrees to which DPRX or any of the DPRX Subsidiaries is subject with respect to any material DPRX Intellectual Property.
- (iv) To the knowledge of DPRX, there is no valid basis for a claim of infringement, misappropriation or other violation of material Intellectual Property rights against DPRX or any of the DPRX Subsidiaries in respect of the conduct of their businesses as currently conducted.
- (v) To the knowledge of DPRX, no Person is infringing, misappropriating or otherwise violating any material DPRX Intellectual Property owned, used or held for use by DPRX and the DPRX Subsidiaries in the conduct of the business of DPRX and the DPRX Subsidiaries as presently conducted, and no such claims have been asserted or threatened against any Person by DPRX or the DPRX Subsidiaries or, to the knowledge of DPRX, any other Person, in the past six years.
- (vi) To the knowledge of DPRX, there has been no claim asserted or threatened, or Proceedings of any kind pending or in progress, challenging the scope, validity or enforceability of any material DPRX Intellectual Property applications or registrations (including Patents) owned by or licensed to DPRX or any of the DPRX Subsidiaries.

(t) Regulatory Matters.

- Since December 31, 2013, the businesses of each of DPRX and the DPRX Subsidiaries have been and are being conducted in material compliance with all Laws governing the quality, identity, strength, purity, safety, efficacy, investigation, development, record keeping, reporting, testing, development, manufacturing, processing, packaging, labeling, storage, transportation, importation, exportation and distribution of pharmaceutical drugs, including, to the extent applicable (A) the Federal Food, Drug, and Cosmetic Act of 1938, 21 U.S.C. § 301 et seq. ("FDCA"); (B) the Public Health Service Act of 1944 (the "PHSA"); (C) United States federal Medicare and Medicaid statutes and related state or local statutes or regulations; (D) United States federal or state criminal or civil Laws (including the federal Anti-Kickback Statute (42 U.S.C. §1320a-7(b))), Stark Law (42 U.S.C. §1395nn), False Claims Act (31 U.S.C. §3729, et seq.), the Physician Payments Sunshine Act, the Prescription Drug Marketing Act of 1987, the Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act (collectively, "HIPAA"), and any comparable state, provincial or local Laws; (E) the Orphan Drug Act of 1983, 96 Stat. 2049 (the "Orphan Act"), (F) state licensing, disclosure and regulations issued under such Laws.
- (ii) Each of DPRX and the DPRX Subsidiaries holds all Regulatory Authorizations necessary for the lawful operating of their businesses and the import, testing, handling, storage, or transportation, as applicable, of each of their products. Except as set forth in Section 3.1(t)(ii) of the DPRX Disclosure Letter, all such Regulatory Authorizations are valid and in full force and effect, or in the process of being obtained in the ordinary course of business. Since December 31, 2013, there has not occurred any violation of, default (with or without notice or lapse of time or both) under, or event giving to others any right of termination, amendment or cancellation of, with or without notice or lapse of time or both, any Regulatory Authorization, except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on DPRX. DPRX and each of the DPRX Subsidiaries are in material compliance with the terms of all Regulatory Authorizations, and no event has occurred that, to the knowledge of DPRX, would reasonably be expected to result in the suspension, revocation, cancellation, nonrenewal or adverse modification of any Regulatory Authorization.
- (iii) All pre-clinical and clinical investigations conducted or sponsored by DPRX or any of DPRX Subsidiaries have been since December 31, 2013 and are being conducted in compliance in all material respects with all applicable Laws and Regulatory Guidelines administered or issued by the applicable Regulatory Authorities, including where applicable (A) FDA standards for conducting non-clinical laboratory studies contained in Title 21 part 58 of the Code of Federal Regulations, (B) FDA standards for the design, conduct, performance, monitoring, auditing, recording, analysis and reporting of clinical trials contained in Title 21 parts 50, 54, 56, 312, 314 and 320 of the Code of Federal Regulations (collectively, the "FDA Regulations"), (C) Division 5 of the Canada Food and Drug Regulations regarding Drugs for Clinical Trials Involving Human Subjects, and (D) federal, state and provincial Laws and Regulatory Guidelines restricting the collection, use and disclosure of individually identifiable health information and personal information. Neither DPRX nor the DPRX Subsidiaries have received any written notice, correspondence or other communication from any Regulatory Authority, since December 31, 2013 initiating or requiring, and are not aware of any facts which are reasonably likely to cause, the termination, suspension or materially adverse modification of any clinical trial conducted or sponsored by DPRX or the DPRX Subsidiaries.

- (iv) All material reports, documents, claims, permits, applications, accreditations and notices required to be filed, maintained or furnished to the FDA or any other Regulatory Authority by DPRX and its Subsidiaries have been so filed, maintained or furnished. All such reports, documents, claims, permits, applications, and notices were complete and accurate in all material respects on the date filed (or were corrected in or supplemented by a subsequent filing) such that no liability exists with respect to such filing. Neither DPRX nor any of its Subsidiaries, nor, to the knowledge of DPRX, any officer, employee, agent or distributor of DPRX or any of its Subsidiaries, has made an untrue statement of a material fact or a fraudulent statement to the FDA or any other Regulatory Authority, failed to disclose a material fact required to be disclosed to the FDA or any other Regulatory Authority, or, to the knowledge of DPRX, committed an act, made a statement, or failed to make a statement that, at the time such disclosure was made, would reasonably be expected to provide a basis for the FDA to invoke its policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities", set forth in the FDA Compliance Policy Guides § 120.100 (the "Fraud Policy") or for any other Regulatory Authority to invoke any similar policy.
- (v) DPRX and any of DPRX Subsidiaries (A) is not a party to and does not have any obligations under any settlement agreement entered into with any Regulatory Authority and (B) since December 31, 2013, has not been the subject of any Regulatory Authority or medical reimbursement investigation other than routine audits and reviews, in either case that would not be expected to have a Material Adverse Effect on DPRX.
- (vi) Neither DPRX nor any of DPRX Subsidiaries, nor, to the knowledge of DPRX, any officer, employee, agent or distributor of DPRX or any of DPRX Subsidiaries, has been convicted of any crime or engaged in any conduct for which debarment is mandated by 21 U.S.C. § 335a(a) or any similar Law or authorized by 21 U.S.C. § 335a(b) or any similar Law. Neither DPRX nor any of its Subsidiaries, nor, to the knowledge of DPRX, any officer, employee, agent or distributor of DPRX or any of its Subsidiaries, has been convicted of any crime or engaged in any conduct for which such Person could be excluded from participating in the United States federal health care programs under Section 1128 of the Social Security Act of 1935, as amended (the "Social Security Act"), or any similar Law or program.

- (vii) Each product candidate currently under development by DPRX and which is subject to the FDCA or any similar Law or Regulatory Guidelines in any foreign jurisdiction that is or has been developed, manufactured, and/or tested by or on behalf of DPRX or any of the DPRX Subsidiaries (each a "DPRX Product") is being or has been developed, imported, tested, handled, stored, transported, or exported in material compliance with all applicable requirements under the FDCA and applicable state, provincial and similar Laws and Regulatory Guidelines, including those relating to investigational use, special access, premarket clearance or marketing approval, good manufacturing practices, good clinical practices, good laboratory practices, labeling, advertising, record keeping, filing of reports and security except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on DPRX. To DPRX's knowledge, no employee of DPRX or a DPRX Subsidiary responsible for management of the import, testing, manufacturing, handling, storage, transportation, or export of the DPRX Products has been sanctioned by a Governmental Authority for non-compliance with applicable Laws or Regulatory Guidelines.
- (viii) (A) Neither DPRX nor any of the DPRX Subsidiaries nor, to DPRX's knowledge to the extent it relates to any DPRX Products, any subcontractors, contract manufacturers or other vendors has, since December 31, 2013, received any FDA Form 483, notice of adverse finding, notice of violation, untitled letter, warning letter, or other similar correspondence or notice from the FDA, state, or any other Regulatory Authority, and (B) there is no action or proceeding pending or, to the knowledge of DPRX, threatened, in the case of either (A) or (B): (I) contesting the premarket clearance or approval of, the uses of, the reimbursement of, or the labeling or promotion of any DPRX Product (II) contesting the compliance with Law or Regulatory Guidelines of any facility where a DPRX Product is developed, tested, manufactured, handled, stored, distributed or transported or (III) otherwise alleging any violation applicable to any DPRX Product or manufacturing process of any Law or Regulatory Guidelines by DPRX or DPRX's Subsidiaries.
- (ix) Since December 31, 2013, DPRX and DPRX's Subsidiaries have not either voluntarily or involuntarily initiated, conducted or issued, or caused to be initiated, conducted or issued, any recall, field notification, field correction, market withdrawal or replacement, warning, "dear doctor" letter, investigator notice, safety alert or other notice or action relating to an alleged lack of safety, lack of efficacy, adulteration, misbranding or lack of regulatory compliance of any DPRX Product. DPRX and DPRX's Subsidiaries are not aware of any facts which are reasonably likely to cause, and neither DPRX nor any of the DPRX Subsidiaries has received any written notice that the FDA or any other Regulatory Authority or Governmental Authority has commenced, or threatened to initiate, any action to cause (A) the seizure, recall, market withdrawal or replacement of any DPRX Product, (B) a change in the marketing classification or a material change in the labeling or advertising of any DPRX Product, or (C) a termination, suspension, or injunction of the manufacture, marketing, storage or distribution of any DPRX Products. DPRX and DPRX's Subsidiaries have complied in all material respects with all recalls, market withdrawals or other corrective action and have no obligation or liability with respect to any recall, market withdrawal or corrective action.

- (u) Books and Records. The corporate records and minute books of DPRX and the DPRX Subsidiaries have been maintained in accordance with all applicable Laws in all material respects, and such corporate records and minute books are complete and accurate in all material respects, including, but not limited to the fact that, the minute books contain the minutes of all meetings of the boards of directors, committees of the board and shareholders and all resolutions passed by the boards of directors, committees of the boards and the shareholders except that minutes of certain recent meetings of the DPRX Board of Directors or committees thereof have not been finalized as of the date hereof. The financial books, records and accounts of DPRX and the DPRX Subsidiaries (i) have in all material respects been maintained in accordance with good business practices and in accordance with U.S. GAAP and with the accounting principles generally accepted in the country of domicile of each such entity on a basis consistent with prior years, and (ii) accurately and fairly reflect, in all material respects, the basis for the consolidated financial statements of DPRX. All such corporate records and minute books of DPRX and the DPRX Subsidiaries have been provided or otherwise made available to PLx.
- (v) Opinion of DPRX Financial Advisor. The DPRX Board of Directors has received the opinion of DPRX's financial advisor to the effect that, as of the date of this Agreement and based on and subject to the assumptions, qualifications and limitations set forth therein, the Equity Exchange Ratio is fair, from a financial point of view, to DPRX. A true, correct and complete copy of such written opinion will be provided by DPRX to PLx, solely for informational purposes, not later than two Business Days after the date hereof.
- (w) Board of Directors Approval. The DPRX Board of Directors has determined that the Transaction is fair, from a financial point of view, to DPRX and is in the best interests of DPRX, has approved the execution and delivery of this Agreement and the entering into of the Transaction, and has resolved to recommend that DPRX Stockholders vote in favor of the DPRX Stockholder Resolution.
- (x) Full Disclosure. No representation or warranty of DPRX contained in this Agreement, no statement of DPRX contained in the DPRX Disclosure Letter or in any certificate furnished to PLx pursuant to any provision of this Agreement and no information included in the DPRX Public Disclosure Record contains or will contain any untrue statement of a material fact or omits or will omit to state a material fact necessary to make the statements herein or therein true in any material respect.
 - (y) [Reserved].
- (z) Insurance. Section 3.1(z) of the DPRX Disclosure Letter contains an accurate and complete list as of the date of this Agreement of all insurance policies owned by DPRX or any DPRX Subsidiary. All current insurance policies and contracts of DPRX and the DPRX Subsidiaries are in full force and effect and are valid and enforceable, and all premiums due thereunder have been paid. None of DPRX nor any of the DPRX Subsidiaries has received notice of cancellation or termination with respect to any material insurance policies or contracts (other than in connection with normal renewals of any such insurance policies or contracts) nor, to the knowledge of DPRX, have any claims been denied under any current insurance policies, and, to the knowledge of DPRX, no threat has been made to cancel any insurance policy or contract of DPRX or any DPRX Subsidiary as of the date of this Agreement, or to deny any claim under current insurance policies or contract.
- (aa) *Stockholder Approval*. The only vote of the DPRX Stockholders required to approve the DPRX Stockholder Resolution in accordance with applicable Law is the DPRX Stockholder Approval. No other vote of the securityholders of DPRX is required by Laws, the governing documents of DPRX or otherwise to adopt this Agreement and approve the Transaction.

(bb) *Brokers and Finders*. Except as set forth in Section 3.1(bb) of the DPRX Disclosure Letter, neither DPRX nor any of its Subsidiaries has used any broker or finder in connection with the transactions contemplated hereby, and no other broker, finder or investment banker is entitled to any fee or commission from DPRX or any of its Subsidiaries in connection with the transactions contemplated hereby, and no Person is or may become entitled to receive any fee or other amount from DPRX or any of its Subsidiaries in connection with the transactions contemplated hereby. A true and correct copy of the engagement letter with DPRX's financial advisor in connection with the transactions contemplated hereby has been provided to PLx or otherwise been made available and has not been subsequently amended, waived or supplemented.

(cc) No Other Representations and Warranties. Except for the representations and warranties made by DPRX in this Section 3.1, neither DPRX nor any other Person makes any express or implied representation or warranty with respect to DPRX or any of its Subsidiaries or their respective businesses, assets, operations, liabilities, condition (financial or otherwise) or prospects, and DPRX hereby disclaims any such other representations or warranties. In particular, without limiting the foregoing disclaimer, except for the representations and warranties made by DPRX in this Section 3.1, neither DPRX nor any other Person makes or has made any representation or warranty to PLx or any of its Representatives, with respect to (i) any financial projection, forecast, estimate, budget or prospective information relating to DPRX, any of the DPRX Subsidiaries or their respective businesses or operations or (ii) any oral or written information furnished or made available to PLx or any of its Representatives in the course of their due diligence investigation of DPRX, the negotiation of this Agreement or the consummation of the Transaction, including the accuracy, completeness or currency thereof, and neither DPRX nor any other Person will have any liability to PLx or any other Person in respect of such information, including any subsequent use of such information, except in the case of fraud. Notwithstanding anything contained in this Agreement to the contrary, DPRX acknowledges and agrees that none of PLx or any other Person has made or is making any representations or warranties whatsoever, express or implied, beyond those expressly made by PLx in Section 3.2, including any implied representation or warranty as to the accuracy or completeness of any information regarding PLx furnished or made available to DPRX, or any of its Representatives.

(dd) *Disclosure*; *DPRX Information*. The information relating to DPRX or its Subsidiaries to be supplied by or on behalf of DPRX for inclusion or incorporation by reference in the Joint Proxy Statement will not, on the date the Joint Proxy Statement is first mailed to DPRX Stockholders or at the time of the DPRX Meeting, contain any untrue statement of any material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not false or misleading at the time and in light of the circumstances under which such statement is made. Notwithstanding the foregoing, no representation is made by DPRX or AcquireCo with respect to the information that has been or will be supplied by PLx or any of it Representatives for inclusion in the Joint Proxy Statement.

- (ee) *DPRX Voting Agreements*. To DPRX's knowledge, the DPRX Specified Stockholders representing, as of the execution of this Agreement, not less than 35% of the outstanding DPRX Shares have delivered to DPRX true and complete and duly executed copies of the DPRX Voting Agreements. As of the date hereof, to DPRX's knowledge, each DPRX Voting Agreement is in full force and effect and, to the knowledge of DPRX, represents a valid, binding and enforceable obligation of the DPRX Specified Stockholders except as enforceability may be limited by bankruptcy and other similar laws and general principles of equity.
- 3.2 <u>Representations and Warranties of PLx.</u> Except as disclosed in the applicable section or subsection of the PLx Disclosure Letter (it being agreed that disclosure of any item in any section or subsection of the PLx Disclosure Letter shall be deemed disclosure with respect to any other section or subsection of the PLx Disclosure Letter only to the extent the relevance of such item to such other section or subsection is reasonably apparent on its face), PLx represents and warrants to and in favor of DPRX as follows:
- (a) Organization and Qualification. PLx has been duly incorporated, validly exists and is in good standing under the DGCL and has the requisite corporate and legal power and capacity to own its assets as now owned and to carry on its business as it is now being carried on. Each of the PLx Subsidiaries is a corporation or other entity duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, organization or formation and has the requisite corporate, legal or other power and authority to own its assets as now owned and to carry on its business as it is now being carried on. PLx and each of the PLx Subsidiaries is duly qualified to carry on business in each jurisdiction in which the nature or character of the respective properties and assets, owned, leased or operated by it, or the nature of its business or activities, makes such qualification necessary, except where the failure to be so qualified would not reasonably be expected to be have a Material Adverse Effect on PLx. PLx has provided or otherwise made available to DPRX true, complete and correct copies of the governing documents of each of PLx and PLx's Subsidiaries, in each case as amended.
- (b) Authority Relative to this Agreement. PLx has the requisite corporate power, authority and capacity to enter into this Agreement and (subject to obtaining the PLx Stockholder Approval, as contemplated in this Agreement) to perform its obligations hereunder and to complete the Transaction. The execution and delivery of this Agreement and the completion by PLx of the Transaction have been duly authorized by the PLx Board of Directors and no other corporate proceedings on the part of PLx are necessary to authorize the execution and delivery by it of this Agreement or, subject to obtaining the PLx Stockholder Approval as contemplated in this Agreement, the completion by PLx of the Transaction. This Agreement has been duly executed and delivered by PLx and constitutes a legal, valid and binding obligation of PLx enforceable against PLx in accordance with its terms, subject to bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and other Laws relating to limitations of actions or affecting the availability of equitable remedies and the enforcement of creditors' rights generally and general principles of equity.
- (c) Required Approvals. No authorization, license, Permit, certificate, registration, consent or approval of, or filing with, or notification to, any Governmental Authority is necessary for the execution and delivery by PLx of this Agreement, the performance by PLx of its obligations hereunder and the completion by PLx of the Transaction, other than:

- (i) such filings and other actions required under applicable U.S. Securities Laws and the rules and policies of NASDAQ, in each case, as are contemplated by this Agreement; and
- (ii) any other authorizations, licenses, Permits, certificates, registrations, consents, approvals and filings and notifications with respect to which the failure to obtain or make the same would not reasonably be expected to have a Material Adverse Effect on PLx, or could not reasonably be expected to prevent or significantly impede or materially delay the completion of the Merger.
- (d) No Violation. Subject to obtaining the authorizations, consents and approvals and making the filings referred to in Section 3.2(c) and complying with applicable Laws and Orders, the execution and delivery by PLx of this Agreement, the performance by PLx of its obligations hereunder and the completion of the Transaction do not and will not (nor will they with the giving of notice or the lapse of time or both):
 - (i) to the knowledge of PLx, result in a contravention, breach, violation or default under any Law or Order applicable to PLx or any of the PLx Subsidiaries or by which any of its or their respective properties or assets are bound or affected:
 - (ii) result in a contravention, conflict, violation, breach or default under the governing documents of PLx or any of the PLx Subsidiaries;
 - (iii) result in a contravention, breach or default under or termination of, or acceleration or permit the acceleration of the performance required by, or loss of any benefit under, any PLx Material Contract or material Permit to which it or any of the PLx Subsidiaries is a party or by which it or any of the PLx Subsidiaries is bound or to which any of its or any of the PLx Subsidiaries' properties or assets is subject or give to any Person any interest, benefit or right, including any right of purchase or sale, termination, payment, modification, reimbursement, penalty, cancellation or acceleration, under any such Material Contract or material Permit; or
 - (iv) result in the suspension or alteration in the terms of any material Permit held by PLx or any of the PLx Subsidiaries or in the creation of any Lien upon any of their properties or assets;

except, in the case of each of clauses (i), (iii) and (iv) above, as would not reasonably be expected to have a Material Adverse Effect on PLx or would not prevent or materially delay the consummation of the Merger.

- Capitalization of PLx. As of the date of this Agreement, the authorized capital of PLx consists of 100,000,000 shares of common stock, of which 5,565,823 shares are issued and outstanding (all of which have been duly authorized and validly issued and are fully paid and non-assessable), and 10,000,000 shares of preferred stock, of which no shares are issued or outstanding. As of the date of this Agreement, 1,000,000 shares of common stock were reserved for issuance pursuant to the PLx Stock Plan and 405,000 shares of common stock were reserved for issuance pursuant to the PLx Convertible Notes. As of the date of this Agreement, there are 877,865 PLx Options outstanding under the PLx Stock Plan providing for the issuance of an aggregate of 877,865 PLx Shares upon the exercise thereof in accordance with their respective terms. None of such 877,865 PLx Shares or 877,865 PLx Options are owned by PLx or any Subsidiary of PLx. There is no outstanding contractual obligation of PLx or any of its Subsidiaries to repurchase, redeem or otherwise acquire any PLx Shares. Except for the PLx Stock Plan and PLx Convertible Notes, as of the date of this Agreement, there are no outstanding agreements, subscriptions, warrants, options, rights or commitments (nor has PLx granted any other right or privilege (contingent or otherwise) capable of becoming an agreement, subscription, warrant, option, conversion or exchange privilege, right, arrangement or commitment) obligating PLx to issue or sell any shares of common stock or other securities of PLx, including any security or obligation of any kind convertible into or exchangeable or exercisable for any shares of common stock or other security of PLx. Except for the PLx Options, PLx has no outstanding agreement, subscription, warrant, option, conversion or exchange privilege right, arrangement or commitment (nor has it granted any right or privilege (contingent or otherwise) capable of becoming an agreement, subscription, warrant, option, conversion or exchange privilege, right, arrangement or commitment) obligating it to issue or sell any PLx Shares or other securities of PLx, including any security or obligation of any kind convertible into or exchangeable or exercisable for any PLx Shares or other security of PLx. Except for the PLx Options, neither PLx nor any of the PLx Subsidiaries has outstanding any stock appreciation right, phantom equity, restricted share unit, deferred share unit or similar right, agreement, arrangement or commitment based on the book value, income or any other attribute of or related to PLx or any of its Subsidiaries. No securities of PLx or any of the PLx Subsidiaries are listed on any stock or securities exchange or market or registered under any securities Laws. There are no outstanding bonds, debentures or other evidences of indebtedness of PLx or any of the PLx Subsidiaries having the right to vote (or that are convertible into or exchangeable or exercisable for securities having the right to vote) with the holders of PLx Shares on any matter. Section 3.2(e) of the PLx Disclosure Letter sets out a true, complete and correct list of all PLx Options, the names of the holders thereof, and the grant date of such securities. A true, correct and complete copy of the PLx Stock Plan has been provided or otherwise made available to DPRX.
- (f) PLx Subsidiaries. Section 3.2(f) of the PLx Disclosure Letter sets forth a true, complete and correct list of each of PLx Subsidiaries, its jurisdiction and form of organization. PLx or a PLx Subsidiary is the sole registered and beneficial owner of all of the outstanding shares in the capital of or outstanding shares of capital stock or other ownership, equity or voting interests of PLx Subsidiaries free and clear of any Liens (other than Permitted Liens), and no other Person has any option, right, entitlement, understanding or commitment (contingent or otherwise) regarding the right to acquire any such share or interest in any of the PLx Subsidiaries and no outstanding option, warrant, conversion or exchange privilege or other right, agreement, arrangement or commitment obligating any such entity to issue or sell any share or ownership, equity or voting interest of such entity or security or obligation of any kind convertible into or exchangeable or exercisable for any shares or ownership, equity or voting interests of any such entity. Neither PLx nor any of the PLx Subsidiaries own any interest or investment (whether equity or debt) in any other Person, other than a PLx Subsidiary, which interest or investment is material to PLx and the PLx Subsidiaries, taken as a whole.
 - (g) [Reserved].

- (h) Financial Statements. The PLx Financial Statements have been prepared in accordance with U.S. GAAP applied on a basis consistent with those of previous periods and in accordance with applicable Laws except as otherwise stated in the notes to such statements or in the auditor's report thereon and subject, in the case of the PLx Interim Financial Statements, to normal year-end audit adjustments, which are not material to PLx and the PLx Subsidiaries, taken as a whole, individually or in the aggregate, and may omit notes which are not material and are not required by applicable Laws or U.S. GAAP. The PLx Financial Statements present fairly, in all material respects, the consolidated financial position and consolidated results of operations, changes in shareholders' equity and cash flows of PLx and the PLx Subsidiaries as of the respective dates thereof and for the respective periods set forth therein. There are no outstanding loans made by PLx or any of the PLx Subsidiaries to any director or officer of PLx.
- (i) No Undisclosed Liabilities. PLx and the PLx Subsidiaries have no liability or obligation of any nature (whether accrued, absolute, contingent or otherwise) that would be required to be disclosed on a balance sheet (or the footnotes thereto) prepared in accordance with U.S. GAAP, other than (i) liabilities and obligations incurred in the ordinary course of business since the date of the most recent PLx Annual Financial Statements that have not had and would not reasonably be expected to have, individually or in aggregate with all other liabilities and obligations of PLx and the PLx Subsidiaries, a Material Adverse Effect on PLx, and (ii) liabilities and obligations incurred in connection with this Agreement and the Transaction. Without limiting anything set forth herein, the PLx Financial Statements reflected and continued to reflect, in each case as of the date filed, appropriate reserves under U.S. GAAP for contingent liabilities relating to pending or anticipated litigation and other contingent obligations of PLx and the PLx Subsidiaries.
 - (j) [Reserved].
- (k) Compliance with Laws. Since December 31, 2013, the business of PLx and of each of the PLx Subsidiaries has been and is currently being conducted in material compliance with all applicable Laws, Orders and Regulatory Guidelines and neither PLx nor any PLx Subsidiary has received any written notice of any alleged material non-compliance or violation of any such Laws, Orders or Regulatory Guidelines. Neither PLx nor any of the PLx Subsidiaries has knowingly taken or committed to take any action which would cause PLx or any of the PLx Subsidiaries to be in violation of the United States Foreign Corrupt Practices Act, or any applicable Laws of similar effect, and, to the knowledge of PLx, no such action has been taken by any Person acting on behalf of PLx or any of the PLx Subsidiaries.
- (l) Litigation. There is no Proceeding against or involving PLx or any of the PLx Subsidiaries (whether in progress, pending or, to the knowledge of PLx, threatened), and no event or circumstance has occurred which would reasonably be expected to give rise to any such Proceeding. Neither PLx nor any of the PLx Subsidiaries nor any of their respective properties or assets is subject to any outstanding Order that would reasonably be expected to (i) prevent or significantly impede or materially delay the completion of the Merger or (ii) have a Material Adverse Effect on PLx.
- (m) Real Property. Section 3.2(m) of the PLx Disclosure Letter contains a list of all leases pursuant to which PLx or any PLx Subsidiary currently leases real property as tenant. Neither PLx nor any of the PLx Subsidiaries owns any real property.

(n) Assets. PLx or its Subsidiaries own or otherwise hold good and valid legal title to, and, where their interests are registrable, are the sole record owners, or hold a valid leasehold interest or license in, all material tangible assets and tangible properties that are required to conduct the business and operations of PLx and the PLx Subsidiaries as presently conducted and there are no Liens (other than Permitted Liens) on any such assets or properties.

(o) Contracts.

- (i) Except as set forth in Section 3.2(o)(i) of the PLx Disclosure Letter, as of the date of this Agreement, none of PLx or any of the PLx Subsidiaries is a party to or bound by any of the following types of Contract (each of the following types of Contracts, a "PLx Material Contract")
 - (A) any collective bargaining agreement, or similar Contract with any labor union or association, with respect to its employees, and any Contract with any officer, employee, consultant or director;
 - (B) any Contract entered into outside of the ordinary course of business that is not terminable by PLx or any of the PLx Subsidiaries on three months' notice or less;
 - (C) any credit agreement, loan agreement, indenture, note, mortgage, security agreement, loan commitment or other Contract relating to the indebtedness of PLx or any PLx Subsidiary;
 - (D) any Contract granting to any Person a right of first refusal or option to purchase or acquire any assets of PLx or any PLx Subsidiary;
 - (E) any real property lease, rental or occupancy agreement under which PLx or any PLx Subsidiary continues to have obligations or rights;
 - (F) any Contract entered into outside of the ordinary course of business pursuant to which PLx or any PLx Subsidiary (i) is granted or obtains or agrees to obtain any right to use any material Intellectual Property (excluding commercially available software), (ii) is restricted in its right to use or register any material Intellectual Property owned by PLx or any of the PLx Subsidiaries, or (iii) permits or agrees to permit any other Person, to use, obtain, enforce or register any material Intellectual Property owned by PLx or any of the PLx Subsidiaries, including any license agreements, option agreements, and covenants not to sue;
 - (G) except for any non-solicit obligations, any Contract that obligates PLx or any PLx Subsidiary or its Affiliates not to compete with another Person, requires PLx or any PLx Subsidiary to acquire any product, assets or service exclusively from any other Person, or otherwise contractually restricts PLx or any PLx Subsidiary or its Affiliates from acquiring any material product, asset or service from any other Person, or providing products, assets or services to any other Person, or developing or distributing any product to any Person or in any geographic location;

- (H) any Contract entered into since December 31, 2013: (i) relating to the merger, consolidation, reorganization, liquidation, dissolution or any similar extraordinary transaction with respect to PLx or any PLx Subsidiary, (ii) relating to a material acquisition or disposition by PLx or any PLx Subsidiary, (iii) relating to the acquisition, issuance or transfer of any securities of PLx or any PLx Subsidiary or (iv) relating to any partnership, strategic alliance or joint venture agreement; and
- (I) except for Contracts entered into in the ordinary course of business with any employee, director or officer of PLx or any PLx Subsidiary, any Contract with any shareholder of PLx or any PLx Subsidiary entered into since December 31, 2013.
- (ii) True, correct and complete copies of each PLx Material Contract in effect on the date hereof has been provided or otherwise made available to DPRX.
- (iii) Except as would not reasonably be expected to have a Material Adverse Effect on PLx, none of PLx, the PLx Subsidiaries or, to the knowledge of PLx, any of the other parties thereto, is in breach or violation of or in default under, or committed or failed to perform any act which would result in a default under, (in each case, with or without notice or lapse of time or both) any PLx Material Contract in any material respect, and none of PLx or any of the PLx Subsidiaries has received or given any notice of default under any PLx Material Contract which remains uncured. To the knowledge of PLx, there exists no state of facts which after notice or lapse of time or both would constitute a default under or breach or violation of any PLx Material Contract or the inability of a party to any PLx Material Contract to perform its obligations thereunder where, in any such case, such default, breach, violation or non-performance has had or would reasonably be expected to have a Material Adverse Effect on PLx. To the knowledge of PLx, no Person has challenged in writing the validity or enforceability of any PLx Material Contract.
- (iv) Other than pursuant to Voting Agreements with PLx Specified Stockholders, there are no shareholders or stockholders agreements, registration rights agreements, voting trusts, proxies or similar agreements, arrangements or commitments to which PLx or any of the PLx Subsidiaries is a party or, to the knowledge of PLx, with respect to any shares or other equity interests of PLx or any of the PLx Subsidiaries or any other Contract relating to disposition, voting or dividends with respect to any shares or other equity securities of PLx or any of the PLx Subsidiaries.
- (v) As of the date of this Agreement, neither PLx nor any of the PLx Subsidiaries has received written notice of the termination of, or intent to terminate or otherwise fail to materially perform any PLx Material Contract.

(p) Taxes.

- (i) PLx and each of its Subsidiaries has duly and timely made or prepared all income and other material Returns required to be made or prepared by it, has duly and timely filed all income and other material Returns required to be filed by it with the appropriate Governmental Authority and has completely and correctly reported all material income and all other amounts or information required to be reported thereon.
- (ii) PLx and each of its Subsidiaries has (A) duly and timely paid all material Taxes due and payable by it and (B) duly and timely withheld all material Taxes and other material amounts required by applicable Laws to be withheld by it and has duly and timely remitted to the appropriate Governmental Authority such material Taxes and other material amounts required by applicable Laws to be remitted by it.
- (iii) No audit, action, investigation, deficiency, litigation, proposed adjustment or other Proceeding exists or has been asserted in writing or, to the knowledge of PLx, threatened in writing with respect to income or other material Taxes or income or other material Returns of PLx or any of its Subsidiaries, and neither PLx nor any of its Subsidiaries is a party to any Proceeding for assessment, reassessment, or collection of material Taxes and no such Proceeding has been asserted in writing or, to the knowledge of PLx, threatened in writing against PLx or any of its Subsidiaries or any of their respective assets, and there are no matters of dispute or matters under discussion with any Governmental Authority relating to income or other material Taxes assessed by any Governmental Authority against PLx or any of its Subsidiaries or relating to income or other material Returns No claim has ever been made in writing by any Governmental Authority in a jurisdiction where PLx or any of its Subsidiaries does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.
- (iv) There are no currently effective or pending material elections, agreements, or waivers extending the limitation period or providing for an extension of time with respect to the assessment or reassessment of any income or other material Taxes, the filing of any income or other material Return (excluding, for the avoidance of doubt, ordinary course extensions within which to file tax returns), or the payment of any income or other material Taxes by PLx or any of its Subsidiaries.
- (v) There are no Liens for income or other material Taxes on the property or assets of PLx or any of its Subsidiaries except for statutory liens for Taxes not yet due and payable.
- (vi) Neither PLx nor any Subsidiary thereof has ever been a member of any consolidated, unitary, combined or similar group for any income or other materials Tax Purposes. Neither PLx nor any Subsidiary there is a party to any Tax allocation, sharing, indemnity, or reimbursement agreement or arrangement (other than any commercial agreement or arrangement a principal purpose of which does not pertain to Tax). Neither PLx nor any Subsidiary thereof has any liability for the Taxes of any other Person under law, as a transferee or successor, by contract (other than any commercial agreement or arrangement a principal purpose of which does not pertain to Tax) or otherwise.

- (vii) No closing agreement, private letter ruling or similar agreements or rulings concerning Taxes has been entered into, issued or requested by any Governmental Authority with respect to PLx or any of the PLx Subsidiaries for any taxable year for which the limitation period has not yet expired.
- (viii) The charges, accruals, and reserves for Taxes reflected on the PLx Interim Financial Statements (whether or not due and whether or not shown on any Return but excluding any provision for deferred income taxes) are adequate under GAAP to cover Taxes with respect to PLx and each of its Subsidiaries accruing through the date hereof.
- (ix) PLx is, and at all times since July 28, 2015, has been, treated as a corporation for U.S. federal income tax purposes.
- (x) Neither DPRX nor any of its Subsidiaries will be required to include any material item of income or exclude any material item of deduction for any taxable period (or portion thereof) beginning after the Closing as a result of (i) any "closing agreement," as described in Section 7121 of the Code (or any corresponding provision of state, local or foreign Law) entered into by PLx or any of its Subsidiaries on or before the Closing Date, (ii) any installment sale or open transaction disposition by PLx or any of its Subsidiaries that occurred before the Closing Date, (iii) a change in method of accounting or use of an improper method of accounting by PLx or any of its Subsidiaries for a taxable period ending on or before the Closing Date, (iv) an election under Section 108(i) of the Code by PLx or any of its Subsidiaries on or before the Closing Date, or (iv) the receipt of any prepaid revenue by a by PLx or any of its Subsidiaries before the Closing Date.
- (xi) Neither PLx nor any of its Subsidiaries is or has been a party to any "listed transaction," as defined in Treasury Regulations Section 1.6011- 4(b)(2).
- (xii) Neither PLx nor any of its Subsidiaries has distributed equity interests of another Person, or has had its equity interests distributed by another Person, during the last two (2) years in a transaction that was purported or intended to be governed in whole or in part by Sections 355 or 361 of the Code.
- (q) Employment Agreements and Collective Agreements. Except as set forth in Section 3.2(q) of the PLx Disclosure Letter, none of PLx or any of the PLx Subsidiaries are parties to or bound or governed by (or currently negotiating in connection with entering into), or subject to, or has any liability with respect to:

- (i) any employment, retention or change of control agreement with, or any written or oral agreement, commitment, obligation, arrangement, plan or understanding providing for any retention, bonus, severance, change of control, retirement or termination payments to any current (or, to the extent remaining outstanding, former) director, officer or employee of PLx or any of PLx's Subsidiaries (each, a "PLx Employment Agreement");
- (ii) any collective bargaining or union agreements or other Contract with a labor union, labor organization or employee association, or any actual or, to the knowledge of PLx, threatened application for certification, recognition or bargaining rights in respect of PLx or any of the PLx Subsidiaries, or any Proceeding seeking to compel PLx or any of the PLx Subsidiaries to bargain with any labor organization as to wages or conditions of employment;
- (iii) any organized labor dispute, work stoppage or slowdown, strike or lock-out relating to or involving any employees of PLx or any of the PLx Subsidiaries; or
- (iv) any actual or, to the knowledge of PLx, threatened grievance, claim or other Proceeding arising out of or in connection with any labor or employment matter by PLx or any of the PLx Subsidiaries or the termination thereof except as would not be expected to have a Material Adverse Effect on PLx.

True, complete and correct copies of the agreements, arrangements, plans and understandings referred to in paragraphs (i) and (ii) of this Section 3.2(q) have been provided or otherwise made available to DPRX. Except as would not be expected to have a Material Adverse Effect on PLx, each of PLx and the PLx Subsidiaries is in material compliance with all applicable Laws (domestic and foreign), Orders, Contracts and PLx material policies relating to employment, employment practices, wages, hours and terms and conditions of employment.

(r) Pension and Employee Benefits.

- (i) Section 3.2(r)(i) of the PLx Disclosure Letter sets forth a true, complete and correct list of each employee benefit and compensation plan, agreement, program or arrangement, whether written or unwritten, including without limitation, any option, restricted share unit, deferred share unit, stock purchase, or other stock or stock-based incentive plan, cash bonus or incentive compensation arrangement, retirement or deferred compensation plan, profit sharing plan, unemployment or severance compensation plan or health and welfare plan, or PLx Employment Agreement, for any current or former employee or director, to the extent the potential liability remains outstanding, of, or other service provider to, PLx or any of its Subsidiaries participates in, is a party or contributes to, or with respect to which PLx or any of its Subsidiaries could reasonably be expected to have any liability (each, a "PLx Plan").
- (ii) With respect to each PLx Plan, PLx has provided or otherwise made available to DPRX (A) a true and complete copy of each PLx Plan, including any amendments thereto and all material supporting documents; (B) latest annual report, if any; (C) copies of all material communications received in the last three years with applicable Government Authority; (D) each trust or other funding arrangement, (E) each summary plan description (if applicable) and (F) where applicable, the most recent financial statements and actuarial or other valuation reports prepared with respect thereto.

- (iii) The consummation of the transactions contemplated by this Agreement will not, either alone or in combination with another event, (A) entitle any current or former employee or officer of PLx to termination or severance pay, unemployment compensation or any other payment, (B) accelerate the time of funding (through a grantor trust or otherwise), payment or vesting, or increase the amount of compensation or benefit due any such employee or officer, or (C) cause amounts payable under the PLx Plans to fail to be deductible for U.S. federal income tax purposes by virtue of Section 280G of the Code. No employee or individual consultant or independent contractor is entitled to receive any gross-up or additional payment by reason of the tax required by Section 409A or 4999 of the Code being imposed upon such person.
- (iv) Each PLx Plan has been established, registered, qualified, funded, invested, operated and administered in all material respects in accordance with its terms and applicable Law (including Section 409A of the Code). There are no pending, or to the knowledge of PLx, threatened actions, suits, disputes or claims by or on behalf of any PLx Plan, by any employee or beneficiary covered under any such PLx Plan, as applicable, or otherwise involving any such PLx Plan (other than routine claims for benefits).
- (v) No PLx Plan provides welfare or post-retirement benefits, including without limitation, death or medical benefits (whether or not insured), beyond retirement or termination of service to employees or former employees or to the beneficiaries or dependents of such employees, other than coverage mandated solely by applicable Law.
- (vi) No PLx Plan is governed by, and PLx has no liability under, Section 401(a) of the Code or ERISA. Neither PLx, nor any Person that is a member of a "controlled group of corporations" with, or is under "common control" with, or is a member of the same "affiliated service group", with PLx, in each case as defined in Sections 414(b), (c), (m) or (o) of the Code sponsors, contributes to or has any liability under, and in the past six years sponsored, contributed to or had liability under, a plan subject to Title IV or Section 302 of ERISA.
- (vii) There has been no amendment to, written interpretation or announcement (whether or not written) by PLx or any of its Subsidiaries relating to, or change in employee participation or coverage under, a PLx Plan which would increase materially the expense of maintaining such PLx Plan above the level of the expense incurred in respect thereof for the fiscal year ended December 31, 2015. There has been no termination of any material PLx Plan since January 1, 2016.

(viii) All contributions, premiums or Taxes required to be made or paid by PLx or any of its Subsidiaries, as the case may be, under or in connection with the PLx Plans have been made in a timely fashion in accordance with Laws and the terms of the applicable PLx Plan. There are no unfunded liabilities in respect of any PLx Plan and have been properly reflected in the PLx Financial Statements.

(s) Intellectual Property.

- (i) Section 3.2(q)(i) of the PLx Disclosure Letter sets forth a correct and complete list of all (A) issued Patents and Patent applications, (B) Trademark registrations and applications and material unregistered Trademarks, (C) Copyright registrations and applications, and (D) material Software, in each case which is owned or exclusively licensed by PLx and the PLx Subsidiaries in any jurisdiction in the world. PLx or one of the PLx Subsidiaries is the sole and exclusive beneficial and, with respect to applications and registrations (including Patents), record owner or exclusive licensee of the Intellectual Property set forth in Section 3.2(q)(i) of the PLx Disclosure Letter, and, except as could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on PLx, to the knowledge of PLx and the PLx Subsidiaries, all such Intellectual Property is subsisting, valid, and enforceable. All required filings and fees related to the (A) issued Patents and Patent applications, (B) Trademark registrations and applications and material unregistered Trademarks, (C) Copyright registrations and applications, and (D) material Software, in each case which is owned or exclusively licensed by PLx have been timely filed with and paid to the relevant Governmental Authorities and authorized registrars.
- (ii) PLx or one of the PLx Subsidiaries owns, or has a valid right to use, free and clear of all Liens (other than Permitted Liens), all Intellectual Property (A) related to the products or product candidates presently used in the conduct of the business of PLx or one of the PLx Subsidiaries and (B) used or held for use in, or necessary to conduct, the business and operations of PLx and the PLx Subsidiaries as presently conducted.
- (iii) There are no Orders, writs, injunctions or decrees to which PLx or any of the PLx Subsidiaries is subject with respect to any material Intellectual Property.
- (iv) To the knowledge of PLx and the PLx Subsidiaries, there is no valid basis for a claim of infringement, misappropriation or other violation of material Intellectual Property rights against PLx or any of the PLx Subsidiaries in respect of the conduct of their businesses as currently conducted.
- (v) To the knowledge of PLx and the PLx Subsidiaries, no Person is infringing, misappropriating or otherwise violating any material Intellectual Property owned, used or held for use by PLx and any of the PLx Subsidiaries in the conduct of the business of PLx and any of the PLx Subsidiaries as presently conducted, and no such claims have been asserted or, to the knowledge of PLx and the PLx Subsidiaries, threatened against any Person by PLx or any of the PLx Subsidiaries or, to the knowledge of PLx and the PLx Subsidiaries, any other Person, in the past six years.

(vi) To the knowledge of PLx, there has been no claim asserted or threatened, or Proceedings of any kind pending or in progress, challenging the scope, validity or enforceability of any material Intellectual Property applications or registrations (including Patents) owned by or licensed to PLx or any of the PLx Subsidiaries.

(t) Regulatory Matters.

- (i) Since December 31, 2013, the businesses of each of PLx and the PLx Subsidiaries have been and are being conducted in material compliance with all Laws governing the quality, identity, strength, purity, safety, efficacy, investigation, development, record keeping, reporting, testing, development, manufacturing, processing, packaging, labeling, storage, transportation, importation, exportation and distribution of pharmaceutical drugs, including, to the extent applicable (A) FDCA; (B) the PHSA; (C) United States federal Medicare and Medicaid statutes and related state or local statutes or regulations;
- (D) United States federal or state criminal or civil Laws (including the federal Anti-Kickback Statute (42 U.S.C. §1320a-7(b))), Stark Law (42 U.S.C. §1395nn), False Claims Act (31 U.S.C. §3729, et seq.), the Physician Payments Sunshine Act, the Prescription Drug Marketing Act of 1987, HIPAA, and any comparable state, provincial or local Laws; (E) the Orphan Act; (F) state licensing, disclosure and reporting requirements; (G) all Laws similar to the foregoing in all other jurisdictions; and (H) all binding rules and regulations issued under such Laws.
- (ii) Each of PLx and the PLx Subsidiaries holds all material Regulatory Authorizations necessary for the lawful operating of their businesses and the import, testing, handling, storage, or transportation, as applicable, of each of their products. All such material Regulatory Authorizations are valid and in full force and effect or in the process of being obtained in the ordinary course of business. Since December 31, 2013, there has not occurred any violation of, default (with or without notice or lapse of time or both) under, or event giving to others any right of termination, amendment or cancellation of, with or without notice or lapse of time or both, any Regulatory Authorization, except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on PLx. PLx and each of the PLx Subsidiaries are in material compliance with the terms of all Regulatory Authorizations, and no event has occurred that, to the knowledge of PLx, would reasonably be expected to result in the suspension, revocation, cancellation, non-renewal or adverse modification of any Regulatory Authorization.

- (iii) All pre-clinical and clinical investigations conducted or sponsored by PLx or any of its Subsidiaries have been since December 31, 2013, and are being conducted in compliance in all material respects with all applicable Laws and Regulatory Guidelines administered or issued by the applicable Regulatory Authorities, including where applicable the FDA Regulations and the federal, state and provincial Laws and Regulatory Guidelines restricting the collection, use and disclosure of individually identifiable health information and personal information, except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on PLx. Neither PLx nor any PLx Subsidiary has received any written notice, correspondence or other communication from the FDA or any other Regulatory Authority since December 31, 2013 initiating or requiring, and are not aware of any facts which are reasonably likely to cause, the termination, suspension or materially adverse modification of any clinical trial conducted or sponsored by PLx or PLx Subsidiaries.
- (iv) All material reports, documents, claims, permits, applications, accreditations and notices required to be filed, maintained or furnished to the FDA or any other Regulatory Authority by PLx and any PLx Subsidiary have been so filed, maintained or furnished. To the knowledge of PLx, all such reports, documents, claims, permits, applications, and notices were complete and accurate in all material respects on the date filed (or were corrected in or supplemented by a subsequent filing) such that no liability exists with respect to such filing. Neither PLx nor any PLx Subsidiaries, nor, to the knowledge of PLx, any officer, employee, agent or distributor of PLx or any of its Subsidiaries, has made an untrue statement of a material fact or a fraudulent statement to the FDA or any other Regulatory Authority, failed to disclose a material fact required to be disclosed to the FDA or any other Regulatory Authority, or, to the knowledge of PLx, committed an act, made a statement, or failed to make a statement that, at the time such disclosure was made, would reasonably be expected to provide a basis for the FDA to invoke the Fraud Policy or for any other Regulatory Authority to invoke any similar policy.
- (v) Neither PLx nor any of the PLx Subsidiaries has received any written information from the FDA or any other Regulatory Authority that would reasonably be expected to lead to the denial of any application for marketing approval currently pending before the FDA or such other Regulatory Authority.
- (vi) Neither PLx nor any of the PLx Subsidiaries (A) is party to or has any obligations under any settlement agreement entered into with any Regulatory Authority or (B) since December 31, 2013, has been the subject of any Regulatory Authority or medical reimbursement investigation other than routine audits and reviews, in each case that would be expected to have a Material Adverse Effect on PLx.
- (vii) Neither PLx nor any of the PLx Subsidiaries, nor, to the knowledge of PLx, any officer, employee, agent or distributor of PLx or any of its Subsidiaries, has been convicted of any crime or engaged in any conduct for which debarment is mandated by 21 U.S.C. § 335a(a) or any similar Law or authorized by 21 U.S.C. § 335a(b) or any similar Law. Neither PLx nor any of its Subsidiaries, nor, to the knowledge of PLx, any officer, employee, agent or distributor of PLx or any of its Subsidiaries, has been convicted of any crime or engaged in any conduct for which such Person could be excluded from participating in the United States federal health care programs under Section 1128 of the Social Security Act or any similar Law or program.

- (viii) To the knowledge of PLx, each product or product candidate currently under development or being sold by PLx and which is subject to the FDCA or any similar Law or Regulatory Guidelines in any foreign jurisdiction that is or has been developed, manufactured, tested, distributed and/or marketed by or on behalf of PLx or any of the PLx Subsidiaries (each a "PLx Product") is being or has been developed, imported, tested, manufactured, handled, stored, transported, sold, distributed, marketed, promoted, or exported in material compliance with all applicable requirements under the FDCA, and applicable state, provincial and similar Laws and Regulatory Guidelines, including those relating to investigational use, special access, premarket clearance or marketing approval, good manufacturing practices, good clinical practices, good laboratory practices, labeling, advertising, record keeping, filing of reports and security except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on PLx. To PLx's knowledge, since December 31, 2013 no employee of PLx or a PLx Subsidiary responsible for management of the import, testing, manufacturing, handling, storage, transportation, sale, distribution, marketing, promotion, or export of PLx Products has been sanctioned by a Governmental Authority for non-compliance with applicable Laws or Regulatory Guidelines.
- (ix) Neither PLx nor any PLx Subsidiary has, since December 31, 2013 received any FDA Form 483, notice of adverse finding, notice of violation, untitled letter, warning letter, or other similar correspondence or notice from the FDA, any state, or any other Regulatory Authority and there is no action or proceeding pending or, to the knowledge of PLx, threatened (A) contesting the premarket clearance or approval of, the uses of, the reimbursement of, or the labeling or promotion of any PLx Product (B) contesting the compliance with Law or Regulatory Guidelines of any facility where a PLx Product is developed, tested, manufactured, handled, stored, distributed or transported or (C) otherwise alleging any violation applicable to any PLx Product or manufacturing process of any Law or Regulatory Guidelines by PLx or PLx's Subsidiaries.
- (x) Since December 31, 2013, PLx and PLx's Subsidiaries have not either voluntarily or involuntarily initiated, conducted or issued, or caused to be initiated, conducted or issued, any recall, field notification, field correction, market withdrawal or replacement, warning, "dear doctor" letter, investigator notice, safety alert or other notice or action relating to an alleged lack of safety, lack of efficacy, adulteration, misbranding or lack of regulatory compliance of any PLx Product. PLx and PLx Subsidiaries are not aware of any facts which are reasonably likely to cause, and neither PLx nor any of the PLx Subsidiaries has received any written notice that the FDA or any other Regulatory Authority or Governmental Authority has commenced, or threatened to initiate, any action to cause (A) the seizure, recall, market withdrawal or replacement of any PLx Product, (B) a change in the marketing classification or a material change in the labeling or advertising of any PLx Products, or (C) a termination, suspension, or injunction of the manufacture, marketing, storage or distribution of any PLx Products. PLx and the PLx Subsidiaries have complied in all material respects with all recalls, market withdrawals or other corrective action and have no obligation or liability with respect to any recall, market withdrawal or corrective action.

- (u) Insurance. Section 3.2(s) of the PLx Disclosure Letter contains an accurate and complete list as of the date of this Agreement of all insurance policies owned by PLx or any PLx Subsidiary. All current insurance policies and contracts of PLx and the PLx Subsidiaries are in full force and effect and are valid and enforceable, and all premiums due thereunder have been paid. None of PLx nor any of the PLx Subsidiaries has received notice of cancellation or termination with respect to any material insurance policies or contracts (other than in connection with normal renewals of any such insurance policies or contracts) nor, to the knowledge of PLx, have any claims been denied under any current insurance policies, and, to the knowledge of PLx, no threat has been made to cancel any insurance policy or contract of PLx or any PLx Subsidiary as of the date of this Agreement, or to deny any claim under current insurance policies or contract.
- (v) Stockholder Approval. The only vote of the stockholders of PLx required to adopt this Agreement and approve the Merger in accordance with applicable Law is the PLx Stockholder Approval. No other vote of the PLx Stockholders is required by Law, the governing documents of PLx or otherwise to adopt this Agreement and approve the Transaction.
- (w) Fairness Opinion. The PLx Board of Directors has received the PLx Fairness Opinion to the effect that, subject to the assumptions, limitations and qualifications set forth therein, as of the date of such opinion, the Equity Exchange Ratio was fair, from a financial point of view, to the PLx Stockholders. A true, correct and complete copy of such written opinion will be provided by PLx to DPRX, solely for informational purposes, not later than two Business Dates after the date hereof.
- (x) Board of Directors Approval. The PLx Board of Directors has determined that this Agreement, and the Merger are fair to PLx Stockholders and are in the best interests of PLx, has approved the execution and delivery of this Agreement and the transactions contemplated by this Agreement and, subject to Section 6.4, has resolved to recommend that PLx Stockholders vote in favor of the adoption of this Agreement.
- (y) Full Disclosure. No representation or warranty of PLx contained in this Agreement, no statement of PLx contained in the PLx Disclosure Letter or in any certificate furnished to DPRX pursuant to any provision of this Agreement contains or will contain any untrue statement of a material fact or omits or will omit to state a material fact necessary to make the statements herein or therein true in any material respect.
- (z) Brokers and Finders. Except as set forth in Section 3.2(z) of the PLx Disclosure Letter, neither PLx nor any of its Subsidiaries has used any broker or finder in connection with the transactions contemplated hereby, and no other broker, finder or investment banker is entitled to any fee or commission from PLx or any of its Subsidiaries in connection with the transactions contemplated hereby, and no Person is or may become entitled to receive any fee or other amount from PLx or any of its Subsidiaries in connection with the transactions contemplated hereby. A true and correct copy of the engagement letter with PLx's financial advisor in connection with the transactions contemplated hereby has been provided to DPRX or otherwise been made available and has not been subsequently amended, waived or supplemented.

(aa) No Other Representations and Warranties. Except for the representations and warranties made by PLx in this Section 3.2, neither PLx nor any other Person makes any express or implied representation or warranty with respect to PLx or any of its Subsidiaries or their respective businesses, assets, operations, liabilities, condition (financial or otherwise) or prospects, and PLx hereby disclaims any such other representations or warranties. In particular, without limiting the foregoing disclaimer, except for the representations and warranties made by PLx in this Section 3.2, neither PLx nor any other Person makes or has made any representation or warranty to DPRX or any of its Representatives, with respect to (i) any financial projection, forecast, estimate, budget or prospective information relating to PLx, any PLx Subsidiary or their respective businesses or operations or (ii) any oral or written information furnished or made available to DPRX or any of its Representatives in the course of their due diligence investigation of PLx, the negotiation of this Agreement or the consummation of the Transaction, including the accuracy, completeness or currency thereof, and neither PLx nor any other Person will have any liability to DPRX or any other Person in respect of such information, including any subsequent use of such information, except in the case of fraud. Notwithstanding anything contained in this Agreement to the contrary, PLx acknowledges and agrees that none of DPRX or any other Person has made or is making any representations or warranties whatsoever, express or implied, beyond those expressly made by DPRX in Section 3.1, including any implied representation or warranty as to the accuracy or completeness of any information regarding DPRX furnished or made available to PLx, or any of its Representatives.

(bb) *Disclosure; PLx Information*. The information relating to PLx or its Subsidiaries to be supplied by or on behalf of PLx for inclusion or incorporation by reference in the Joint Proxy Statement will not, on the date the Joint Proxy Statement is first mailed to PLx stockholders or at the time of the PLx Meeting, contain any untrue statement of any material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not false or misleading at the time and in light of the circumstances under which such statement is made. Notwithstanding the foregoing, no representation is made by PLx with respect to the information that has been or will be supplied by DPRX or any of it Representatives for inclusion in the Joint Proxy Statement.

(cc) PLx Voting Agreements. To PLx's knowledge, the PLx Specified Stockholders representing, as of the execution of this Agreement, not less than 35% of the outstanding PLx Shares have delivered to PLx true and complete and duly executed copies of the PLx Voting Agreements. As of the date hereof, to PLx's knowledge, each PLx Voting Agreement is in full force and effect and, to the knowledge of PLx, represents a valid, binding and enforceable obligation of the PLx Specified Stockholders except as enforceability may be limited by bankruptcy and other similar laws and general principles of equity.

3.3 <u>Survival of Representations and Warranties</u>. The representations and warranties of the Parties contained in this Agreement will not survive the completion of the Merger and will expire and be terminated on the earlier of the Merger Effective Time and, subject to the obligation to make any payment hereunder pursuant to Section 7.2, the date on which this Agreement is terminated in accordance with its terms. This Section 3.3 will not limit any covenant or agreement of any of the Parties, which, by its terms, contemplates performance after the Closing or the date on which this Agreement is terminated, as the case may be.

ARTICLE IV

COVENANTS REGARDING THE CONDUCT OF BUSINESS

- 4.1 <u>Covenants of DPRX</u>. Except as set forth in Section 4.1 of the DPRX Disclosure Letter, DPRX covenants and agrees that, until the earlier of the Closing and the time that this Agreement is terminated in accordance with its terms, unless PLx otherwise consents in writing (to the extent that such consent is permitted by applicable Law), which consent shall not be unreasonably withheld, conditioned or delayed (except in the case of clauses (c)(i) and (xxi) below, for which PLx's consent may be withheld, conditioned or delayed in its sole discretion), or expressly permitted or specifically contemplated by this Agreement or as is required by applicable Law or Order:
- (a) the respective businesses of DPRX and its Subsidiaries will be conducted, their respective facilities will be maintained, and DPRX and its Subsidiaries will continue to operate their respective businesses only in the ordinary course of business;
- (b) DPRX and its Subsidiaries will comply in all material respects with the terms of all DPRX Material Contracts and DPRX will use its commercially reasonable efforts to maintain and preserve intact its and its Subsidiaries' respective business organizations, assets, Permits, properties, rights, goodwill and business relationships and keep available the services of its and its Subsidiaries' respective officers and employees as a group;
 - (c) DPRX will not, and will cause its Subsidiaries not to, directly or indirectly:
 - (i) alter or amend its certificate of incorporation, bylaws or other governing documents except as may be required to effect the Transaction;
 - (ii) declare, set aside or pay any dividend on or make any distribution or payment or return of capital in respect of the DPRX Shares (whether in cash or property);
 - (iii) split, divide, consolidate, combine or reclassify the DPRX Shares or any other securities of DPRX;
 - (iv) issue, grant, sell or pledge or authorize or agree to issue, grant, sell or pledge any DPRX Shares or other securities of DPRX or its Subsidiaries (including options or any equity-based or equity-linked awards such as restricted or deferred share units or phantom share plans), which are convertible into or exchangeable or exercisable for, or otherwise evidencing a right to acquire, the DPRX Shares, other than the issuance of the DPRX Shares issuable pursuant to (A) the Merger; (B) the issuance of Replacement PLx Options as provided in Section 2.1(j) and (k); (C) the exercise of the DPRX Options or DPRX Warrants outstanding on the date hereof or (D) Section 4.1(c) of the DPRX Disclosure Letter;

- (v) (A) grant any increases in the compensation or benefits of any of its directors, individual independent contractors, executive officers, employees or consultants; or (B) except as contemplated by this Agreement or as required by applicable Law or the terms of any DPRX Plan in effect as of the date hereof (i) grant or increase any severance, change in control, termination or similar compensation or benefits payable to any director, individual independent contractor, officer or employee, (ii) accelerate the time of payment or vesting of, or the lapsing of restrictions with respect to, or fund or otherwise secure the payment of, any compensation (including bonuses) or benefits under any DPRX Plan; (iii) enter into, terminate or materially amend any DPRX Plan (or, except as provided in Section 2.1(j) and (k), any plan, program, agreement, or arrangement that would constitute a DPRX Plan if in effect on the date hereof) or make any loans to employees; (iv) terminate any person who is, or hire any person to be, employed by or a consultant of DPRX or any of its Subsidiaries other than the hiring or termination of employees or consultants with total annual compensation not in excess of \$100,000 in the ordinary course of business consistent with past practice, and in the case of hiring of employees, is solely to replace employees or consultants who are reasonably essential to DPRX; and (v) loan or advance any money to any employee or individual independent contractor of DPRX or any of its Subsidiaries;
- (vi) redeem, purchase or otherwise acquire any outstanding DPRX Shares or other securities convertible into or exchangeable or exercisable for DPRX Shares;
 - (vii) amend the terms of any securities of DPRX or any of its Subsidiaries;
- (viii) adopt a plan of liquidation or resolution providing for the liquidation or dissolution of DPRX or any of its Subsidiaries;
 - (ix) reorganize, amalgamate or merge with any other Person other than pursuant to the Merger;
- (x) make any material changes to any of its accounting policies, principles, methods, practices or procedures (including by adopting any material new accounting policies, principles, methods, practices or procedures) or as contemplated hereby or in connection with any transactions contemplated hereby, except as required by applicable Laws or U.S. GAAP;
- (xi) except for sales in the ordinary course of business, or as contemplated hereby or in connection with any transactions contemplated hereby, sell, pledge, lease, license, abandon or dispose of any assets or properties of DPRX (including the shares or other equity securities of any Subsidiary of DPRX) or of any of its Subsidiaries;

- (xii) (A) acquire (by merger, amalgamation, consolidation, arrangement or acquisition of shares or other equity securities or interests or assets or otherwise) any corporation, partnership, association or other business organization or division thereof or any property or asset, or make any investment by the purchase of securities, contribution of capital, property transfer, or purchase of any property or assets of any other Person other than pursuant to the Merger; or (B) enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement with respect to such a transaction;
- (xiii) (A) incur any indebtedness, other than trade payables in the ordinary course of business, (B) enter into any hedging, derivative or swap transaction or Contract, (C) issue any debt securities, or assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other Person, or (D) make any loans or advances;
- (xiv) (A) pay, discharge or satisfy any material claim, liability or obligation prior to the same being due, other than the payment, discharge or satisfaction of liabilities reflected or reserved against in the DPRX Financial Statements or in the ordinary course of business and consistent with past practice, or (B) voluntarily waive, release, assign, settle or compromise any Proceeding where such waivers, releases, assignments, settlements or compromises exceed \$50,000 in the aggregate or in any case would entail any non-monetary damages;
- (xv) settle or compromise any action, claim or other Proceeding brought by any present, former or purported holder of its securities in connection with the Transaction;
 - (xvi) enter into any material new line of business, enterprise or other activity;
 - (xvii) expend or commit to expend any amounts with respect to capital expenses in excess of \$50,000;
- (xviii) other than in the ordinary course of business, enter into any contract that would, if entered into prior to the date hereof, be a DPRX Material Contract, or (y) materially modify, materially amend or terminate any DPRX Material Contract or waive, release or assign any material rights or claims thereunder;
- (xix) make, change, revoke or rescind in any manner that is material and adverse to DPRX any election relating to Taxes, settle or compromise any Tax controversy, or make any material amendment with respect to any Return, change any method of Tax accounting or change in annual Tax accounting period, settle or compromise any audit or proceeding relating to a material amount of Taxes, agree to an extension or waiver of the statute of limitations with respect to a material amount of Taxes, or surrender any right to claim a material Tax refund;
- (xx) make, or permit any of DPRX's Subsidiaries to, make, any loan to any officer or director of DPRX or any of its Subsidiaries;

- (xxi) negotiate or enter into any collective bargaining agreement, collective agreement or other contract with any labor organization or union or other employee association; or
- (xxii) enter into, modify or terminate any Contract with respect to any of the foregoing or otherwise agree or announce an intention to do any of the foregoing.
- (d) DPRX will promptly notify PLx in writing of the occurrence of any event which would have a Material Adverse Effect with respect to DPRX.

Nothing in this Section 4.1 shall give PLx or any PLx Subsidiary the right to control, directly or indirectly, the operations or the business of DPRX or any of its Subsidiaries at any time prior to the Closing.

- 4.2 <u>Covenants of PLx.</u> PLx covenants and agrees that, until the earlier of the Closing and the time that this Agreement is terminated in accordance with its terms, unless DPRX otherwise consents in writing (to the extent that such consent is permitted by applicable Law), which consent shall not be unreasonably withheld, conditioned or delayed, or as is otherwise disclosed in Section 4.2 of the PLx Disclosure Letter or expressly permitted or specifically contemplated by this Agreement or as is otherwise required by applicable Law or Order:
- (a) the respective businesses of PLx and its Subsidiaries will be conducted, their respective facilities will be maintained, and PLx and its Subsidiaries will continue to operate their respective businesses in the ordinary course of business;
- (b) PLx and its Subsidiaries will comply in all material respects with the terms of all PLx Material Contracts and PLx will use its commercially reasonable efforts to maintain and preserve intact its and its Subsidiaries' respective business organizations, assets, Permits, properties, rights, goodwill and business relationships and keep available the services of its and its Subsidiaries' respective officers and employees as a group;
 - (c) PLx will not and will not permit any of the PLx Subsidiaries to, directly or indirectly:
 - (i) alter or amend its certificate of incorporation, bylaws or other governing documents except as may be required to effect the Transaction;
 - (ii) declare, set aside or pay any dividend on or make any distribution or payment or return of capital in respect of any of its equity securities (whether in cash or property) except (A) the payment of interest or other amounts as and when due pursuant to the terms of PLx Convertible Notes and (B) in the case of any of PLx's wholly-owned Subsidiaries, for dividends payable to PLx or among wholly owned Subsidiaries of PLx;
 - (iii) split, divide, consolidate, combine or reclassify PLx Shares or any other securities of PLx;

- (iv) issue, grant, sell or pledge or authorize or agree to issue, grant, sell or pledge any PLx Shares or other securities of PLx or its Subsidiaries (including options or any equity-based or equity-linked awards such as restricted or deferred share units or phantom share plans), which are convertible into or exchangeable or exercisable for, or otherwise evidencing a right to acquire, the PLx Shares, other than up to 405,000 PLx Shares issuable upon conversion of PLx Convertible Notes in accordance with their terms.
- (v) (A) grant any increases in the compensation or benefits of any of its directors, individual independent contractors, executive officers, employees or consultants; or (B) except as contemplated by this Agreement or as required by applicable Law or the terms of any PLx Plan in effect as of the date hereof (i) grant or increase any severance, change in control, termination or similar compensation or benefits payable to any director, individual independent contractor, officer or employee, (ii) accelerate the time of payment or vesting of, or the lapsing of restrictions with respect to, or fund or otherwise secure the payment of, any compensation (including bonuses) or benefits under any PLx Plan; (iii) enter into, terminate or materially amend any PLx Plan or make any loans to employees; (iv) grant any equity or equity-based awards; (v) terminate any person who is, or hire any person to be, employed by or a consultant of PLx or any of its Subsidiaries other than the hiring or termination of employees or consultants with total annual compensation not in excess of \$100,000 in the ordinary course of business consistent with past practice, and in the case of hiring of employees, is solely to replace employees or consultants who are reasonably essential to PLx; and (vi) loan or advance any money to any employee or individual independent contractor of PLx or any of its Subsidiaries;
- (vi) redeem, purchase or otherwise acquire any outstanding PLx Shares or other securities convertible into or exchangeable for PLx Shares, other than upon conversion of PLx Convertible Notes in accordance with their terms;
- (vii) amend the terms of any securities of PLx or its Subsidiaries, other than amendments to the PLx Convertible Notes solely necessary to provide for their conversion in connection with the Merger;
- (viii) adopt a plan of liquidation or resolution providing for the liquidation or dissolution of PLx or any of its Subsidiaries;
 - (ix) reorganize, amalgamate or merge with any other Person other than pursuant to the Merger;
- (x) make any material changes to any of its accounting policies, principles, methods, practices or procedures (including by adopting any material new accounting policies, principles, methods, practices or procedures) or as contemplated hereby or in connection with any transactions contemplated hereby, except as required by applicable Laws or U.S. GAAP;
- (xi) except for sales in the ordinary course of business, or as contemplated hereby or in connection with any transactions contemplated hereby, sell, pledge, lease, license, abandon or dispose of any assets or properties of PLx (including the shares or other equity securities of any Subsidiary of PLx) or of any of its Subsidiaries;

- (xii) (A) acquire (by merger, amalgamation, consolidation, arrangement or acquisition of shares or other equity securities or interests or assets or otherwise) any corporation, partnership, association or other business organization or division thereof or any property or asset, or make any investment by the purchase of securities, contribution of capital, property transfer, or purchase of any property or assets of any other Person other than pursuant to the Merger; or (B) enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement with respect to such a transaction;
- (xiii) (A) incur any indebtedness, other than trade payables in the ordinary course of business, (B) enter into any hedging, derivative or swap transaction or Contract, (C) issue any debt securities, or assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other Person, or (D) make any loans or advances;
- (xiv) (A) pay, discharge or satisfy any material claim, liability or obligation prior to the same being due, other than the payment, discharge or satisfaction of liabilities reflected or reserved against in the PLx Financial Statements or in the ordinary course of business and consistent with past practice, or (B) voluntarily waive, release, assign, settle or compromise any Proceeding where such waivers, releases, assignments, settlements or compromises exceed \$50,000 in the aggregate or in any case would entail any non-monetary damages;
- (xv) settle or compromise any action, claim or other Proceeding brought by any present, former or purported holder of its securities in connection with the Transaction;
 - (xvi) enter into any material new line of business, enterprise or other activity;
 - (xvii) expend or commit to expend any amounts with respect to capital expenses in excess of \$50,000;
- (xviii) other than in the ordinary course of business, enter into any contract that would, if entered into prior to the date hereof, be a PLx Material Contract, or (y) materially modify, materially amend or terminate any PLx Material Contract or waive, release or assign any material rights or claims thereunder;
- (xix) make, change, revoke or rescind in any manner that is material and adverse to PLx any election relating to Taxes, settle or compromise any Tax controversy, or make any material amendment with respect to any Return, change any method of Tax accounting or change in annual Tax accounting period, settle or compromise any audit or proceeding relating to a material amount of Taxes, agree to an extension or waiver of the statute of limitations with respect to a material amount of Taxes, or surrender any right to claim a material Tax refund;

- (xx) make, or permit any of PLx's Subsidiaries to, make, any loan to any officer or director of PLx or any of its Subsidiaries;
- (xxi) expend or commit to expend any amounts for the preparation and filing of reports with the FDA or any other Regulatory Authority;
- (xxii) negotiate or enter into any collective bargaining agreement, collective agreement or other contract with any labor organization or union or other employee association; or
- (xxiii) enter into, modify or terminate any Contract with respect to any of the foregoing or otherwise agreed or announce an intention to do any of the foregoing.
- (d) PLx will promptly notify DPRX in writing of the occurrence of any event which would have a Material Adverse Effect with respect to PLx.

ARTICLE V ADDITIONAL COVENANTS

Access to Information. Subject to compliance with applicable Laws and Orders and the terms of any existing Contracts, each Party shall, and shall cause its respective wholly- owned Subsidiaries to, afford to the other Parties and their respective Representatives, until the earlier of the Closing or the termination of this Agreement in accordance with its terms, continuing access to the other parties' virtual data rooms, and reasonable access, during normal business hours and upon reasonable notice, to its businesses, properties, books and records and such other data and information as a Party may reasonably request, as well as to the other Party's and its Subsidiaries' personnel, subject, however, to such access not interfering with the ordinary conduct of its businesses. Each party will keep such information confidential in accordance with the terms of the Non-Disclosure Agreement.

5.2 <u>Consents and Approvals</u>.

(a) Subject to the terms and conditions of this Agreement (including Section 5.2(e)), each Party shall, and shall cause its wholly-owned Subsidiaries to, use commercially reasonable efforts to take, or cause to be taken, all actions, and do, or cause to be done, and to assist and cooperate with the other Party in doing, all things required or reasonably necessary to consummate and make effective the Transaction as promptly as practicable, including, as promptly as reasonably practicable, taking reasonable actions to provide notice to any third party, or obtain from any third party any waivers, consents and approvals required or reasonably necessary to consummate the Transaction.

- (b) Subject to the terms and conditions hereof, including Section 5.2(e), each of the Parties agrees, and shall cause each of their respective Subsidiaries, to cooperate and to use commercially reasonable efforts to (i) provide such notices and obtain such waivers, consents, clearances and approvals as are required or reasonably necessary to consummate the Transaction under the HSR Act, if applicable, and any other federal, provincial, state or foreign Law designed to prohibit, restrict or regulate actions relating to monopolization or restraint of trade or foreign investment (collectively, "Relevant Laws"), and (ii) respond to any requests of any Governmental Authority for information or documentary material under any Relevant Law, and to contest and resist any action, including any legislative, administrative or judicial action, and to have vacated, lifted, reversed or overturned any Order (whether temporary, preliminary or permanent) that restricts, prevents or prohibits the consummation of the Transaction under any Relevant Law. The Parties shall consult and cooperate with one another, and consider in good faith the views of one another, regarding the form and content of any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any Party in connection with proceedings under or relating to any Relevant Law prior to their submission.
- (c) Each of DPRX and PLx shall, other than in respect of routine correspondence and dealings with NASDAQ regarding the Transaction: (i) promptly advise each other of any written or oral communication (including communications received by their respective Subsidiaries) from any Governmental Authority or third party from whom a waiver, consent or approval is required or reasonably necessary to consummate the Transaction; (ii) not participate in any meeting or discussion with any Governmental Authority in respect of any filing, investigation, or enquiry concerning this Agreement or the Transaction unless it reasonably consults with the other Party in advance, and, unless prohibited by such Governmental Authority, gives the other Party the opportunity to attend; and (iii) as promptly as practicable furnish the other Party with copies of all correspondence, filings, and written communications between them and their Subsidiaries and Representatives, on the one hand, and any Governmental Authority or its staff, on the other hand, with respect to this Agreement and the Transaction, except that materials may be redacted as necessary to address reasonable privilege, competitively sensitive information, or confidentiality concerns.
- (d) Each Party will provide as promptly as practicable such information and documentary material as may be requested by a Governmental Authority following any such filing or notification.
- (e) In furtherance and not in limitation of the other covenants contained in this Section 5.2, but subject to the last sentence of this Section 5.2(e), each of DPRX and PLx agrees to take, or cause to be taken (including by its Subsidiaries), any and all steps and to make, or cause to be made (including by its Subsidiaries), any and all reasonable undertakings necessary to resolve such objections, if any, that a Governmental Authority may assert under any Relevant Law with respect to the Merger, and to use commercially reasonable efforts to avoid or eliminate any impediments under any Relevant Law that may be asserted by any Governmental Authority with respect to the Merger, so as to enable the Merger Effective Time to occur as promptly as practicable and in any event no later than the Outside Date. Notwithstanding anything to the contrary contained in this Agreement, no Party shall have any obligation under this Agreement: (i) to dispose of or transfer or cause any of its Subsidiaries to discontinue offering any product or service; (iii) to license or otherwise make available, or cause any of its Subsidiaries to license or otherwise make available to any Person any Intellectual Property; (iv) to hold separate or cause any of its Subsidiaries to hold separate any assets or operations (either before or after the Closing Date); (v) to make or cause any of its Subsidiaries to make any commitment (to any Governmental Authority or otherwise) regarding its future operations; or (vi) to contest any Legal Proceeding or any order, writ, injunction or decree relating to the Merger or the Transaction if such Party determines in good faith that contesting such Legal Proceeding or order, writ, injunction or decree might not be advisable (each of clauses (i) through (vi), a "Restraint").

- 5.3 Covenants of DPRX Regarding the Transaction. Subject to the terms and conditions of this Agreement (including Section 5.2), DPRX will use commercially reasonable efforts to perform all obligations required to be performed by DPRX under this Agreement, reasonably cooperate with PLx in connection therewith, and use commercially reasonable efforts to do such other acts and things as may be necessary or desirable in order to complete the Transaction including:
- (a) subject to Section 9.5, publicly announcing the entering into of this Agreement, the support of the DPRX Board of Directors of the Transaction and the DPRX Recommendation;
- (b) using commercially reasonable efforts to defend all lawsuits or other legal, regulatory or other Proceedings against DPRX challenging or affecting this Agreement or the completion of the Transaction;
- (c) taking all necessary actions and causing AcquireCo to take all necessary actions to give effect to the Merger, including to provide the Exchange Agent with sufficient Merger Consideration to complete the Merger as provided herein.
- 5.4 <u>Covenants of PLx Regarding the Transaction</u>. Subject to the terms and conditions of this Agreement (including Section 5.2), PLx shall and shall cause each of its Subsidiaries to, use commercially reasonable efforts to perform all obligations required to be performed by it under this Agreement, reasonably cooperate with DPRX in connection therewith, and use commercially reasonable efforts to do such other acts and things as may be necessary or desirable in order to complete the Transaction including:
- (a) subject to Section 9.5, publicly announcing the entering into of this Agreement, the support of the PLx Board of Directors of the Transaction and the PLx Recommendation;
- (b) using commercially reasonably efforts to defend all lawsuits or other legal, regulatory or other Proceedings against or relating to PLx challenging or affecting this Agreement or the completion of the Transaction; and
 - (c) taking all necessary actions to give effect to the Merger.
- 5.5 Loan. As an inducement for PLx to enter into this Agreement and effect the Merger, DPRX shall make available to PLx, on or before January 15, 2017, a loan in the amount of \$2,000,000 (the "Loan Amount"), which DPRX shall fund upon three (3) Business Days' prior written notice to DPRX (such notice to include the wire instructions to the PLx account into which the Loan Amount is to be funded), pursuant to a secured promissory note in substantially the form attached hereto as Exhibit E. Prior to the making of any loan from DPRX to PLx, PLx shall deliver to DPRX a signed copy of the secured promissory note. If PLx requests that DPRX loan to PLx the Loan Amount and PLx accepts such loan, PLx shall not, prior to the Closing, permit more than \$500,000 of the Loan Amount to be used for purposes other than those mutually agreed between PLx and DPRX under such note.

5.6 Indemnification and Insurance.

- Effective Time occurs, each of DPRX and PLx shall, jointly and severally, indemnify and hold harmless each person who is now, or has been at any time prior to the date hereof, or who becomes prior to the Merger Effective Time, a director or officer of DPRX or PLx (each such present or former director or officer (i) of PLx being referred to as an "PLx Indemnified Party", and (ii) of DPRX being herein referred to as a "DPRX Indemnified Party" and each PLx Indemnified Party and DPRX Indemnified Party being an "Indemnified Party" and such Persons collectively being referred to as the "Indemnified Parties"), against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys' fees and disbursements, incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the Indemnified Party is or was a director or officer of DPRX or PLx, whether asserted or claimed prior to, at or after the Merger Effective Time, to the fullest extent permitted under Delaware Law for directors or officers of Delaware corporations. Each Indemnified Party will be entitled to advancement of expenses incurred in the defense of any such claim, action, suit, proceeding or investigation from each of DPRX and PLx, jointly and severally, upon receipt by DPRX or PLx from the Indemnified Party of a request therefor; provided that any person to whom expenses are advanced provides an undertaking, to the extent then required by Delaware Law, to repay such advances if it is ultimately determined that such person is not entitled to indemnification.
- (b) Each of DPRX, PLx and PLx's Subsidiaries agree that all rights to indemnification or exculpation now existing in favor of the present and former directors and officers of DPRX, PLx or of any of their respective Subsidiaries as provided in the governing documents of DPRX, PLx or any of their respective Subsidiaries or any Contract by which DPRX, PLx or any of their respective Subsidiaries is bound and which is in effect as of the date hereof, will survive the completion of the Transaction and continue in full force and effect and without modification, with respect to actions or omissions of the Indemnified Parties occurring prior to the Closing, for the period currently contemplated therein.
- coverage for seven years from the Closing Date customary policies of directors' and officers' liability insurance providing protection no less favorable to the protection provided by the policies maintained by DPRX, PLx and their respective Subsidiaries, which are in effect immediately prior to the Closing Date and providing protection in respect of claims arising from facts or events which occurred on or prior to the Closing Date; *provided*, *however*, that each of DPRX and PLx may, prior to the Closing Date, purchase prepaid non-cancellable run-off directors' and officers' liability insurance on terms substantially similar to the directors' and officers' liability policies currently maintained by DPRX or PLx, as applicable, but providing coverage for a period of seven years from the Closing Date with respect to claims arising from or related to facts or events which occurred on or prior to the Closing Date; *provided*, *further*, *however*, that in no event shall either DPRX, PLx or their respective Subsidiaries spend premiums for any of the insurance referenced in this Section 5.6(b) to the extent it would exceed 300% of the relevant party's current annual premium for directors' and officers' liability insurance, as applicable.

- (d) The covenants contained in this Section 5.6 are intended to be for the irrevocable benefit of, and shall be enforceable by, the Indemnified Parties and their respective heirs, executors, administrators and other legal representatives and shall not be deemed exclusive of any other rights which an Indemnified Party has under Law, Contract or otherwise, and shall be binding on DPRX and its successors and assigns. DPRX will act as agent and trustee for the DPRX Indemnified Parties not a party to this Agreement for the covenants of PLx and DPRX under this Section 5.6, and DPRX agrees to accept such appointment and to hold and enforce the obligations and covenants on behalf of each such person. PLx will act as agent and trustee for PLx Indemnified Parties not a party to this Agreement for the covenants of DPRX under this Section 5.6, and PLx agrees to accept such trust and to hold and enforce the obligations and covenants on behalf of each such person.
- (e) In the event DPRX or PLx or any of their respective successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfers all or substantially all of its properties and assets to any Person, then, and in each such case, proper provision shall be made so that the successors and assigns of DPRX or PLx, as the case may be, shall succeed to the obligations set forth in this Section 5.6.
- 8.7 Rule 16b-3 Actions. Prior to the Closing, DPRX and PLx shall take all such steps as may be required (to the extent permitted under applicable Law and no-action letters issued by the SEC) to cause any acquisitions of DPRX Shares (including derivative securities with respect to DPRX Shares) resulting from the Merger and the other transactions contemplated by this Agreement, by each individual who may become or is reasonably expected to become subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to DPRX to be exempt under Rule 16b-3 promulgated under the Exchange Act.
- 5.8 Stock Exchange Listing. DPRX shall use commercially reasonable efforts to cause the DPRX Shares (i) issued as Merger Consideration, and (ii) issuable on exercise of Replacement PLx Options to be approved for listing on NASDAQ, subject only to official notice of issuance, prior to the Closing. PLx shall provide the information required for an initial listing application pursuant to NASDAQ Rule 5110 and shall fully cooperate and participate in preparing such application and obtaining such listing, including, without limitation, the payment of the costs thereof. If reasonably determined by PLx to be in the best interests of the Surviving Corporation and its stockholders, DPRX agrees to cooperate and participate in seeking to effect the Reverse Stock Split, including the preparation of an additional DPRX Charter Amendment seeking to effect such Reverse Stock Split and inclusion of same for approval in the Joint Proxy Statement.
- 5.9 <u>Takeover Statutes</u>. If any anti-takeover statute or similar statute or regulation is or may become applicable to the Transaction, each of the Parties and its respective Affiliates shall (a) grant such approvals and take all such actions as are legally permissible so that the Transaction may be consummated as promptly as practicable on the terms contemplated hereby and (b) otherwise act to eliminate or minimize the effects of any such statute or regulation on the Transaction.

- 5.10 <u>Board of Directors and Officers</u>. DPRX and PLx shall take all actions necessary so that, as of the Merger Effective Time, the board of directors of DPRX shall consist of six individuals designated by PLx prior to Closing and one individual designated by DPRX and acceptable to PLx prior to Closing, each until their respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal; *provided*, *however*, the Parties acknowledge that so long as DPRX remains a public reporting company, the board of directors of DPRX will continue to satisfy applicable securities laws, including, without limitation, maintaining an independent audit committee, and the designees by PLx and DPRX hereunder will allow DPRX to comply with such applicable Law. At and immediately after the Merger Effective Time, the officers of DPRX as the remaining public company shall be specified in <u>Schedule</u> 5.10.
- 5.11 <u>Lock-Up Agreements</u>. On or prior to the initial filing date of the Joint Proxy Statement, each of DPRX and PLx shall take all actions necessary to deliver the DPRX Lock-Up Agreements and the PLx Lock-Up Agreements, as applicable.

ARTICLE VI ACQUISITION PROPOSALS

6.1 DPRX Non-Solicitation.

- (a) Subject to Section 6.2, until the earlier of the Closing or the date, if any, on which this Agreement is terminated pursuant to Section 7.1, DPRX shall not, and DPRX shall cause its Subsidiaries and each of its and their respective Representatives not to, directly or indirectly through any other Person:
 - (i) initiate, solicit, facilitate or knowingly encourage (including by way of furnishing or affording access to information), or take any other action that promotes, directly or indirectly, or may reasonably cause, any inquiries or the making of any proposal or offer with respect to a DPRX Acquisition Proposal or potential DPRX Acquisition Proposal;
 - (ii) participate or engage in any discussions or negotiations regarding, or otherwise cooperate in any way with, or assist or participate in, knowingly encourage or otherwise facilitate, any effort or attempt by any other Person (other than PLx and its Affiliates) to make or complete a DPRX Acquisition Proposal;
 - (iii) effect any DPRX Change of Recommendation; or
 - (iv) accept or enter into, or publicly propose to accept or enter into, any letter of intent, memorandum of understanding, agreement in principle, merger agreement, acquisition agreement, transaction agreement, implementation agreement, option agreement, joint venture agreement, alliance agreement, partnership agreement or other agreement, arrangement or undertaking constituting or related to, or that would reasonably be expected to lead to, any DPRX Acquisition Proposal (a "DPRX Acquisition Agreement").

- (b) Notwithstanding Section 6.1(a) or any other provision of this Agreement to the contrary, if after the date hereof and prior to the approval of the DPRX Stockholder Resolution at the DPRX Meeting, DPRX or any of its Subsidiaries, or any of its or their respective Representatives, receives a written DPRX Acquisition Proposal (including, an amendment, change or modification to a DPRX Acquisition Proposal made prior to the date hereof) that was not solicited after the date hereof in contravention of this Section 6.1, DPRX and its Representatives may:
 - (i) contact the Person making such DPRX Acquisition Proposal and its Representatives solely for the purpose of clarifying the terms and conditions of such DPRX Acquisition Proposal and the likelihood of its consummation so as to determine whether such DPRX Acquisition Proposal is, or could reasonably be expected to lead to, a DPRX Superior Proposal; and
 - (ii) if the DPRX Board of Directors determines in good faith, after consultation with its outside legal counsel and financial advisors, that such DPRX Acquisition Proposal is, or could reasonably be expected to lead to, a DPRX Superior Proposal:
 - (A) furnish information with respect to DPRX and its Subsidiaries to the Person making such DPRX Acquisition Proposal and its Representatives, provided that (i) DPRX first enters into a confidentiality agreement with such Person that is no less favorable (including with respect to any "standstill" and similar provisions) to DPRX than the Non-Disclosure Agreement, and sends a copy of such agreement to PLx promptly following its execution for informational purposes only and (ii) DPRX contemporaneously provides to PLx any non-public information concerning DPRX and its Subsidiaries that is provided to such Person which was not previously provided to PLx or its Representatives; and
 - (B) engage in discussions and negotiations with respect to a DPRX Acquisition Proposal with the Person making such DPRX Acquisition Proposal and its Representatives.

- 6.2 <u>DPRX Superior Proposal or Intervening Event.</u> Notwithstanding Section 6.1(a) or any other provision of this Agreement to the contrary, DPRX may, at any time after the date of this Agreement and prior to the approval of the DPRX Stockholder Resolution at the DPRX Meeting, (x) accept, approve or enter into any agreement, understanding or arrangement in respect of a DPRX Acquisition Proposal (with the exception of a confidentiality agreement described in Section 6.1(b)(ii)(A), the execution of which shall not be subject to the conditions of this Section 6.2(a)) or (y) effect a DPRX Change of Recommendation with respect to any DPRX Acquisition Proposal, if:
- (a) such DPRX Acquisition Proposal did not result from a breach of Section 6.1 and DPRX has complied with the other terms of this Section 6.2;
- (b) the DPRX Board of Directors has determined in good faith, after consultation with its outside legal counsel and financial advisors, that a DPRX Change of Recommendation is required in order to comply with its fiduciary duties under applicable Laws either based upon (1) an Intervening Event or (2) receipt of a DPRX Acquisition Proposal that the DPRX Board of Directors determines in good faith constitutes a DPRX Superior Proposal.
- 6.3 <u>Securities Law Compliance</u>. Nothing in this Article 6 shall prohibit DPRX from complying with Rule 14e-2 or Rule 14d-9 promulgated under the Exchange Act with regard to a DPRX Acquisition Proposal, respectively, or from the Board of Directors of DPRX making any disclosure to the DPRX Stockholders if, in the good faith judgment of the Board of Directors of DPRX, after consultation with its outside legal counsel, that taking such action or making such disclosure would be required to comply with its fiduciary duties under applicable Laws.

6.4 PLx Non-Solicitation.

- (a) Until the earlier of the Closing or the date, if any, on which this Agreement is terminated pursuant to Section 7.1, PLx shall not, and PLx shall cause its Subsidiaries and each of its and their respective Representatives not to, directly or indirectly through any other Person:
 - (i) initiate, solicit, facilitate or knowingly encourage (including by way of furnishing or affording access to information), or take any other action that promotes, directly or indirectly, or may reasonably cause, any inquiries or the making of any proposal or offer with respect to a PLx Acquisition Proposal or potential PLx Acquisition Proposal;
 - (ii) participate or engage in any discussions or negotiations regarding, or otherwise cooperate in any way with, or assist or participate in, knowingly encourage or otherwise facilitate, any effort or attempt by any other Person (other than DPRX and its Affiliates) to make or complete a PLx Acquisition Proposal; or
 - (iii) accept or enter into, or publicly propose to accept or enter into, any letter of intent, memorandum of understanding, agreement in principle, merger agreement, acquisition agreement, transaction agreement, implementation agreement, option agreement, joint venture agreement, alliance agreement, partnership agreement or other agreement, arrangement or undertaking constituting or related to, or that would reasonably be expected to lead to, any PLx Acquisition Proposal.

ARTICLE VII TERMINATION

7.1 Termination.

- (a) This Agreement may be terminated at any time prior to the Closing by mutual written consent of DPRX and PLx duly authorized by the Boards of Directors of DPRX and PLx.
 - (b) This Agreement may be terminated by either DPRX or PLx at any time prior to the Closing:
 - (i) if the Closing does not occur on or before the Outside Date, provided, however, that the right to terminate this Agreement under this Section 7.1(b)(i) shall not be available to a Party if the failure of that Party or its Affiliate to fulfill any of its obligations or breach of any of its representations and warranties under this Agreement has been a principal cause of, or resulted in, the failure of the Closing to occur by the Outside Date; provided, further, however, that, in the event that the Joint Proxy Statement is still being reviewed or commented on by the SEC after March 15, 2017, either Party shall be entitled to extend the Outside Date by an additional sixty (60) days;
 - (ii) if the DPRX Stockholder Resolution (other than the Reverse Stock Split) is not adopted by the DPRX Stockholders in accordance with applicable Laws at the DPRX Meeting or any adjournment or postponement thereof;
 - (iii) if the PLx Stockholder Approval is not obtained at the PLx Meeting or any adjournment or postponement thereof; or
 - (iv) there shall be passed any Law that makes consummation of the Transaction illegal or otherwise prohibited or if any Governmental Authority of competent jurisdiction shall have issued an Order or taken any other action restraining, enjoining or otherwise prohibiting the Merger, and such Order or other action is or shall have become final and non-appealable.
 - (c) This Agreement may be terminated by PLx at any time prior to the Closing if:
 - (i) DPRX shall have effected a DPRX Change of Recommendation;
 - (ii) DPRX materially breaches any of the provisions of Section 6.1 or Section 6.2; or

- (iii) DPRX breaches any of its representations, warranties, covenants or agreements contained in this Agreement (other than as provided in Section 7.1(c)(i) above), which breach would cause any of the conditions set forth in Section 8.3 not to be satisfied and which breach is not cured within 30 days following written notice of such breach or by its nature or timing cannot be cured within that time.
- (d) This Agreement may be terminated by DPRX at any time prior to the Closing if:
- (i) DPRX concurrently enters into a DPRX Acquisition Agreement that constitutes a DPRX Superior Proposal;
 - (ii) PLx materially breaches any of the provisions of Section 6.4; or
- (iii) PLx breaches any of its representations, warranties, covenants or agreements contained in this Agreement, which breach would cause any of the conditions set forth in Section 8.2 not to be satisfied and which breach is not cured within 30 days following written notice of such breach or by its nature or timing cannot be cured within that time

7.2 Termination Fee Events.

(a) If a DPRX Termination Fee Event occurs, DPRX shall pay to PLx a termination fee of \$700,000 (the "**DPRX Termination Fee**") by wire transfer in immediately available funds to an account specified by PLx. The DPRX Termination Fee shall be payable at the time specified in Section 7.2(b).

(b) "DPRX Termination Fee Event" means:

- (i) the termination of this Agreement by DPRX pursuant to Section 7.1(d)(i);
- (ii) the termination of this Agreement by PLx pursuant to Section 7.1(c)(i); or
- (iii) the termination of this Agreement by either DPRX or PLx pursuant to Section 7.1(b)(ii) or by PLx pursuant to Section 7.1(c)(ii), if, in any of the foregoing cases: (x) prior to such termination, a DPRX Acquisition Proposal shall have been publicly announced or made to DPRX or the DPRX Stockholders and has not been publicly withdrawn prior to the DPRX Meeting and (y) within twelve months following such termination, DPRX or one or more of DPRX's Subsidiaries shall have consummated any transaction in respect of such DPRX Acquisition Proposal, in which case the DPRX Termination Fee shall be paid by DPRX on the date of consummation of such transaction.
- (c) If a PLx Termination Fee Event occurs, PLx shall pay to DPRX a termination fee of \$500,000 (the "PLx Termination Fee") by wire transfer in immediately available funds to an account specified by DPRX. The PLx Termination Fee shall be payable at the time specified in Section 7.2(d).

- (d) **"PLx Termination Fee Event"** means the termination of this Agreement by either DPRX or PLx pursuant to Section 7.1(b)(iii).
- (e) Each Party acknowledges that the PLx Termination Fee and DPRX Termination Fee set out in this Section 7.2 are payments of liquidated damages which are a genuine pre-estimate of the damages which PLx or DPRX, as the case may be, will suffer or incur as a result of the event giving rise to such payment and are not penalties. Each of DPRX and PLx irrevocably waives any right that it may have to raise as a defense that any such liquidated damages are excessive or punitive. The Parties agree that the payment of an amount pursuant to this Section 7.2 in the manner provided herein is the sole and exclusive remedy of the Parties in respect of the event giving rise to such payment; *provided*, *however*, that nothing contained in this Section 7.2, and no payment of any such amount, shall relieve or have the effect of relieving a Party in any way from liability for damages incurred or suffered by the other Party as a result of an intentional or willful breach of this Agreement.

ARTICLE VIII CONDITIONS PRECEDENT

- 8.1 <u>Mutual Conditions Precedent.</u> The respective obligations of the Parties to complete the Merger are subject to the satisfaction, or mutual waiver by PLx and DPRX, on or before the Closing Date, of each of the following conditions, each of which are for the mutual benefit of the Parties and which may be waived, in whole or in part, by PLx and DPRX at any time:
- (a) the DPRX Stockholder Approval (other than the stockholder approval of the Reverse Stock Split) shall have been obtained at the DPRX Meeting in accordance with applicable Laws;
- (b) the PLx Stockholder Approval shall have been obtained at the PLx Meeting in accordance with applicable Laws;
- (c) the Form S-4 shall have been declared effective and no stop order suspending the effectiveness of the Form S-4 shall be in effect:
- (d) the DPRX Shares (i) to be issued as Merger Consideration, and (ii) issuable on exercise of Replacement PLx Options shall have been approved for listing on NASDAQ, subject only to official notice of issuance;
- (e) no applicable Law or Order shall be and remain in effect which imposes, and no suit, action, claim, proceeding or investigation shall be pending or threatened by any Governmental Authority which seeks to impose, any material limitations on DPRX's ownership of PLx or any Subsidiary of PLx or any requirement that PLx, AcquireCo or DPRX or any of their respective Subsidiaries agree to or implement any Restraint;
- (f) No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the Merger shall have been issued by any court of competent jurisdiction or other Governmental Authority and remain in effect, and there shall not be any Law which has the effect of making the consummation of the Merger illegal; and

- (g) There shall not be any Legal Proceeding pending, or overtly threatened in writing, by an official of a Governmental Authority in which such Governmental Authority indicates that it intends to conduct any Legal Proceeding or taking any other action: (a) challenging or seeking to restrain or prohibit the consummation of the Merger; (b) relating to the Merger and seeking to obtain from DPRX, AcquireCo or PLx any damages or other relief that may be material to DPRX or PLx; or (c) seeking to prohibit or limit in any material and adverse respect a Party's ability to vote, transfer, receive dividends with respect to or otherwise exercise ownership rights with respect to the stock of DPRX.
- 8.2 Additional Conditions Precedent to the Obligations of DPRX. The obligation of DPRX to complete the Merger shall be subject to the satisfaction, or waiver by DPRX, on or before the Closing Date, of each of the following conditions, each of which is for the exclusive benefit of DPRX and which may be waived by DPRX at any time, in whole or in part, in its sole discretion and without prejudice to any other rights that DPRX may have:
- (a) PLx shall have complied in all material respects with its obligations, covenants and agreements in this Agreement to be performed and complied with on or before the Closing Date;
- (b) (i) the representations and warranties of PLx in Sections 3.2(a), 3.2(b), 3.2(t), 3.2(v) and 3.2(x) shall be true and correct in all material respects, as of the date of this Agreement and as of the Closing Date, as if made on such date (except for such representations and warranties which refer to or are made as of another specified date, in which case such representations and warranties shall have been true and correct as of that date); (ii) the representations and warranties of PLx in Section 3.2(e) shall be true and correct as of the date of this Agreement and as of the Closing Date, as if made on such date; and (iii) the representations and warranties of PLx set forth in Section 3.2 (other than those referenced in clauses (i) and (ii) above) shall be true and correct (disregarding for this purpose all materiality or Material Adverse Effect qualifications contained therein) as of the date of this Agreement and as of the Closing Date, as if made on and as of such date (except for such representations and warranties which refer to or are made as of another specified date, in which case such representations and warranties shall have been true and correct as of that date), except for breaches of representations and warranties which have not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect with respect to PLx;
 - (c) since the date of this Agreement, no Material Adverse Effect with respect to PLx shall have occurred;
- (d) DPRX shall have received a certificate of PLx signed by a senior officer of PLx for and on behalf of PLx and dated the Closing Date certifying that the conditions set out in Section 8.2(a), Section 8.2(b) and Section 8.2(c) have been satisfied;
- (e) The holders of no more than five percent (5%) of the PLx Shares on an as- converted to PLx Shares basis will have demanded, and not lost or withdrawn, or will be eligible to demand, appraisal rights;

- (f) [Reserved];
- (g) [Reserved]; and
- (h) DPRX shall have received a certificate of PLx signed by a senior officer of PLx for and on behalf of PLx and dated the Closing Date (i) identifying all closing or transactional costs of PLx in connection with the Transaction, including amounts payable to financial advisors (including investment banks), attorneys, accountants or proxy solicitors, and (ii) certifying as to which of such amounts remain payable as of the Closing.
- 8.3 Additional Conditions Precedent to the Obligations of PLx. The obligation of PLx to complete the Merger shall be subject to the satisfaction, or waiver by PLx, on or before the Closing Date, of each of the following conditions, each of which is for the exclusive benefit of PLx and which may be waived by PLx at any time, in whole or in part, in its sole discretion and without prejudice to any other rights that PLx may have:
 - (a) DPRX Closing Cash at Closing shall be at least \$12,000,000;
- (b) DPRX shall have complied in all material respects with its obligations, covenants and agreements in this Agreement to be performed and complied with on or before the Closing Date;
- (c) (i) the representations and warranties of DPRX set forth in Sections 3.1(a), 3.1(b), 3.1(e) 3.1(w), 3.1(aa) and 3.1(bb) shall be true and correct in all material respects, as of the date of this Agreement and as of the Closing Date, as if made on such date (except for such representations and warranties which refer to or are made as of another specified date, in which case such representations and warranties shall have been true and correct as of that date); and (ii) the representations and warranties of DPRX set forth in Section 3.1 (other than those referenced in clause (i) above) shall be true and correct (disregarding for this purpose all materiality or Material Adverse Effect qualifications contained therein) as of the date of this Agreement and as of the Closing Date, as if made on and as of such date (except for such representations and warranties which refer to or are made as of another specified date, in which case such representations and warranties shall have been true and correct as of that date), except for breaches of representations and warranties which have not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect with respect to DPRX;
 - (d) since the date of this Agreement, no Material Adverse Effect with respect to DPRX shall have occurred; and
 - (e) DPRX shall have made the Loan Amount available to PLx in accordance with Section 5.5.
- (f) PLx shall have received a certificate of DPRX signed by the Chairman of the Board or a senior officer of DPRX for and on behalf of DPRX and dated the Closing Date certifying that the conditions set out in Section 8.3(a), Section 8.3(b) and Section 8.3(c) have been satisfied.

- 8.4 <u>Notice and Cure Provisions</u>. Each Party will give prompt notice to the other of the occurrence, or failure to occur, at any time from the date hereof until the Merger Effective Time, of any event or state of facts which occurrence or failure would, or would be reasonably likely to:
- (a) cause any of the representations or warranties of such Party contained herein to be untrue or inaccurate between the date hereof and the Merger Effective Time such that the conditions set forth in Section 8.2(b) or Section 8.3(c) would fail to be satisfied; or
- (b) result in the failure to comply with or satisfy any covenant or agreement to be complied with or satisfied by such Party hereunder prior to the Merger Effective Time such that the condition set forth in Section 8.2(a) or Section 8.3(b) would fail to be satisfied.

ARTICLE IX GENERAL

- 9.1 <u>Notices</u>. Any demand, notice or other communication to be given in connection with this Agreement must be given in writing and will be given by personal delivery, registered, certified or first class mail or by electronic transmission, addressed to the recipient as follows:
 - (a) if to PLx:

PLx Pharma Inc. 8285 El Rio Street, Ste. 130 Houston, TX 77054

Attention: President and Chief Executive Officer

E-mail: ngiordano@plxpharma.com

with a copy (which will not constitute notice) to:

Jackson Walker L.L.P. 2323 Ross Ave, Ste. 600 Dallas, TX 75202 Attention: Michael Laussade E-mail: mlaussade@jw.com

(b) if to DPRX:

Dipexium Pharmaceuticals, Inc.
14 Wall Street, Suite 3D
New York, New York 10005
Attention: President and Chief Executive Officer

E-mail: davidluci@dipexium.com

with a copy (which will not constitute notice) to:
Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C.
666 Third Avenue
New York, New York 10017
Attention: Lea V. Plumonthal, For

Attention: Ivan K. Blumenthal, Esq. E-mail: ikblumenthal@mintz.com

or to such other street address, individual or electronic communication number or address as may be designated by written notice given by either Party to the other in any manner stated in this Section 9.1. Any demand, notice or other communication given by personal delivery will be conclusively deemed to have been given on the day of actual delivery thereof, if given by registered, certified or first class mail, the third (3rd) Business Day after being sent and, if given by electronic communication, on the day of transmittal thereof if given during the normal business hours of the recipient and on the Business Day during which such normal business hours next occur if not given during such hours on any day.

- 9.2 Expenses. Except as otherwise specified herein and except in respect of any filing fees associated with any filings made pursuant to Relevant Laws, which fees shall be split evenly between PLx and DPRX, each Party will pay its respective legal and accounting costs and expenses incurred in connection with the preparation, execution and delivery of this Agreement and all documents and instruments executed pursuant to this Agreement and any other costs and expenses whatsoever and howsoever incurred.
- 9.3 <u>No Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by any Party without the prior written consent of the other Parties.
- 9.4 <u>Benefit of Agreement</u>. Subject to Section 9.8, this Agreement will inure solely to the benefit of and be binding upon each Party hereto.

9.5 <u>Public Announcements</u>.

- (a) PLx and DPRX shall jointly publicly announce the Transaction promptly following the execution of this Agreement, the text and timing of such announcement to be mutually agreed by the Parties in advance, acting reasonably.
- (b) No Party shall issue any press release or otherwise make any written public statement with respect to the Merger or this Agreement without the consent of the other Parties (which consent shall not be unreasonably withheld, conditioned or delayed).
- (c) DPRX and AcquireCo shall not make any filing with any Governmental Authority with respect to the Transaction without prior consultation with PLx, and PLx shall not make any filing with any Governmental Authority with respect to the Transaction without prior consultation with DPRX.

The provisions of Section 9.5(b) and 9.5(c) shall be subject to each Party's overriding obligation to make any disclosure or filing required under applicable Laws, and the Party making the disclosure shall use commercially reasonable efforts to give prior oral or written notice to the other Party and reasonable opportunity for the other Party to review or comment on the disclosure or filing (other than with respect to confidential information contained in such disclosure or filing), and if such prior notice is not possible, to give notice immediately following the making of any such disclosure or filing, and *provided further*, *however*, that, except as otherwise required pursuant to this Agreement (other than this Section 9.5), neither DPRX nor PLx shall have any obligation to obtain the consent of or consult with the other Party prior to any press release, public statement, disclosure or filing with regard to any DPRX Acquisition Proposal or DPRX Change of Recommendation.

- 9.6 <u>Governing Law; Attornment; Service of Process; Waiver of Jury.</u>
- (a) This Agreement, and any dispute arising out of, relating to, or in connection with this Agreement shall be governed by and construed in accordance with the Laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware of any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware. Each of the Parties (a) consents to submit itself to the personal jurisdiction of the Court of Chancery of the State of Delaware (the "Chancery Court") or, if, but only if, the Chancery Court lacks subject matter jurisdiction, any federal court located in the State of Delaware with respect to any dispute arising out of, relating to or in connection with this Agreement or any transaction contemplated hereby, (b) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court and (c) agrees that it will not bring any action arising out of, relating to or in connection with this Agreement or any transaction contemplated by this Agreement, in any court other than any such court. The parties irrevocably and unconditionally waive any objection to the laying of venue of any action, suit or proceeding arising out of this Agreement or the transactions contemplated hereby in the Chancery Court or, if, but only if, the Chancery Court lacks subject matter jurisdiction, in any federal court located in the State of Delaware, and hereby further irrevocably and unconditionally waive and agree not to plead or claim in any such court that any such action, suit or proceeding brought in any such court has been brought in an inconvenient forum.
- (b) Each Party hereby agrees that any service of process, summons, notice or document by registered mail addressed to such Person at its address set forth in Section 9.1 shall be effective service of process for any suit, action or proceeding relating to any dispute arising out of this Agreement or the transactions contemplated by this Agreement.
- (c) EACH PARTY HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY SUIT, ACTION OR OTHER PROCEEDING ARISING OUT OF THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.
- 9.7 <u>Entire Agreement</u>. This Agreement, together with the Non-Disclosure Agreement and the Voting Agreements, and any documents delivered hereunder, constitutes the entire agreement between the Parties and supersedes all prior agreements and understandings, both written and oral, among the Parties, with respect to the subject matter thereof.
- 9.8 <u>Third Party Beneficiaries</u>. Except as provided in Sections 5.1, 5.6, this Agreement shall not confer any rights or remedies upon any Person other than the Parties and their respective successors and permitted assigns.

- 9.9 <u>Amendment</u>. This Agreement may, at any time and from time to time but not later than the Closing, be amended by written agreement of the Parties hereto without, subject to applicable Laws, further notice to or authorization on the part of the DPRX Stockholders or PLx Stockholders.
- Waiver and Modifications. Any Party may (a) waive, in whole or in part, any inaccuracy of, or consent to the modification of, any representation or warranty made to it hereunder or in any document to be delivered pursuant hereto, (b) extend the time for the performance of any of the obligations or acts of the other Parties (c) waive or consent to the modification of any of the covenants herein contained for its benefit or waive or consent to the modification of any of the obligations of the other Parties hereto or (d) waive the fulfillment of any condition to its own obligations contained herein. No waiver or consent to the modifications of any of the provisions of this Agreement will be effective or binding unless made in writing and signed by the Party or Parties purporting to give the same and, unless otherwise provided, will be limited to the specific breach or condition waived. The rights and remedies of the Parties hereunder are cumulative and are in addition to, and not in substitution for, any other rights and remedies available at Law or in equity or otherwise. No single or partial exercise by a Party of any right or remedy precludes or otherwise affects any further exercise of such right or remedy or the exercise of any other right or remedy to which that Party may be entitled. No waiver or partial waiver of any nature, in any one or more instances, will be deemed or construed a continued waiver of any condition or breach of any other term, representation or warranty in this Agreement.
- 9.11 <u>Severability</u>. Upon such determination that any provision is illegal, invalid or unenforceable, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in a mutually acceptable manner in order that the Merger be consummated as originally contemplated to the fullest extent possible.
- 9.12 <u>Further Assurances</u>. Subject to the provisions of this Agreement, the Parties will, from time to time, do all acts and things and execute and deliver all such further documents and instruments, as the other Parties may, either before or after the Closing, reasonably require to effectively carry out or better evidence or perfect the full intent and meaning of this Agreement.
- 9.13 <u>Injunctive Relief.</u> The Parties agree that irreparable harm may occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached for which money damages may not be an adequate remedy at Law. It is accordingly agreed that the Parties will be entitled to seek an injunction or injunctions and other equitable relief to prevent breaches of this Agreement, any requirement for the securing or posting of any bond in connection with the obtaining of any such injunctive or other equitable relief hereby being waived.
- 9.14 <u>Counterparts</u>. This Agreement may be executed and delivered in any number of counterparts (including by facsimile or electronic transmission), each of which will be deemed to be an original and all of which taken together will be deemed to constitute one and the same instrument, and each Party may enter into this Agreement by executing a counterpart and delivering it to the other Party (by personal delivery, facsimile, electronic transmission or otherwise).

IN WITNESS WHEREOF the Parties have executed this Agreement as of the date first written above.

PLX PHARMA INC.

By: /s/ Natasha Giordano

Name: Natasha Giordano

Title: President and Chief Executive Officer

DIPEXIUM PHARMACEUTICALS, INC.

By: /s/ David P. Luci

Name: David P. Luci

Title: President and Chief Executive Officer

DIPEXIUM ACQUISITION CORP.

By: /s/ David P. Luci

Name: David P. Luci

Title: President and Chief Executive Officer

Signature Page to Agreement and Plan of Merger and Reorganization

SCHEDULE I

PLx and DPRX Executive Officers, Directors and Holders of PLx Shares and DPRX Shares Signing Lock-Up Agreements

PLx Lock-Up Agreements:

- Aurus Bios Fondo de Inversion Privado
- Integra Ventures III, L.P.
- Charles E. Sheedy
- S. Reed Morian
- Michael J. Valentino
- Ronald R. Zimmerman
- Natasha Giordano
- David E. Jorden
- Gary Mossman
- Gary Balkema
- Robert Casale
- Kirk Calhoun
- John W. Hadden II
- And such additional PLx Stockholders as necessary for Lock-Up Agreements to be delivered with respect to at least 50% of the PLx Shares.

DPRX Lock-Up Agreements:

- David P. Luci
- Robert J. DeLuccia
- William J. McSherry, Jr.
- Dr. Jack H. Dean
- Thomas Harrison
- Michael Duffy
- Barry Kagan

SCHEDULE 5.10

Officers of DPRX After the Merger Effective Time

Michael J. Valentino, Executive Chairman Natasha Giordano, President and Chief Executive Officer David E. Jorden, Chief Financial Officer Gary Mossman, Chief Operating Officer

EXHIBIT A

Form of DPRX Voting Agreement

VOTING AGREEMENT

among:

DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation;

PLX PHARMA INC., a Delaware corporation; and

the undersigned Stockholder

Dated as of December 22, 2016

TABLE OF CONTENTS

1.	Agreement to Vote Shares	
2.	Expiration Date	2
3.	Additional Purchases	2
4.	Agreement to Retain Shares	2
5.	Representations and Warranties of Stockholder	3
6.	Irrevocable Proxy	3
7.	No Solicitation	4
8.	Release; No Legal Actions	4
9.	Other Remedies; Specific Performance	5
10.	Directors and Officers	5
11.	No Ownership Interest	6
12.	Termination	6
13.	Further Assurances	6
14.	Disclosure	6
15.	Notice	6
16.	Severability	6
17.	Assignability	7
18.	No Waivers	7
19.	Applicable Law; Jurisdiction	7
20.	Waiver of Jury Trial	7
21.	No Agreement Until Executed	7
22.	Entire Agreement; Counterparts; Exchanges by Facsimile	8
23.	Amendment	8
24.	Definition of Merger Agreement	8
25.	Construction	8
	1	

VOTING AGREEMENT

THIS VOTING AGREEMENT ("Agreement"), dated as of December 22, 2016, is made by and among Dipexium Pharmaceuticals, Inc., a Delaware corporation ("DPRX"), PLx Pharma Inc., a Delaware corporation ("PLx"), and the undersigned holder ("Stockholder") of shares of capital stock (the "Shares") of DPRX.

WHEREAS, DPRX, Dipexium Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of DPRX ("<u>AcquireCo</u>"), and PLx, have entered into an Agreement and Plan of Merger and Reorganization, dated of even date herewith (in the form in effect on the date hereof and attached hereto as <u>Exhibit A</u> or as amended pursuant to <u>Section 24</u> hereof, the "<u>Merger Agreement</u>"), providing for the merger of AcquireCo with and into PLx (the "<u>Merger</u>");

WHEREAS, Stockholder beneficially owns and has sole or shared voting power with respect to the number of Shares, and holds stock options or other rights to acquire the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of DPRX, AcquireCo and PLx to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, DPRX's, AcquireCo's and PLx's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the expenses incurred and to be incurred by them in connection therewith, Stockholder, DPRX and PLx agree as follows:

- 1. <u>Agreement to Vote Shares</u>. Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of DPRX or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of DPRX, with respect to approval of the Merger as contemplated by the Merger Agreement and adoption of the Merger Agreement or any DPRX Acquisition Proposal, Stockholder shall:
- (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat for purposes of calculating a quorum; and
- (b) vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that such Stockholder shall be entitled to so vote: (i) in favor of (A) the adoption of the Merger Agreement and the approval of the Merger, including without limitation the issuance of the DPRX Shares by virtue of the Merger as contemplated by the Merger Agreement, and (B) the DPRX Charter Amendment; (ii) against any action, proposal, transaction or agreement that, to the knowledge of Stockholder, would reasonably be expected to result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of DPRX under the Merger Agreement or that would reasonably be expected to result in any of the conditions to DPRX's, AcquireCo's or PLx's obligations under the Merger Agreement not being fulfilled; and (iii) against any DPRX Acquisition Proposal, or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement. The Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.

- 2. Expiration Date. As used in this Agreement, the term "Expiration Date" shall mean the earlier to occur of (a) the Merger Effective Time, (b) such date and time as the Merger Agreement shall be terminated pursuant to Article VII thereof or otherwise, or (c) upon mutual written agreement of the parties to terminate this Agreement. Upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; provided, however, such termination or expiration shall not relieve any party from liability for any willful breach of this Agreement or acts of bad faith prior to termination hereof.
- 3. <u>Additional Purchases</u>. Stockholder agrees that any shares of capital stock or other equity securities of DPRX that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any stock options or otherwise ("<u>New Shares</u>"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Agreement to Retain Shares. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below) on) any Shares, (b) deposit any Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens on) any Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing. Stockholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee and transferee shall sign a voting agreement in substantially the form hereof, (b) with respect to such Stockholder's DPRX Options which expire on or prior to the Expiration Date, transfers, sales, or other dispositions of Shares to DPRX as payment for the (i) exercise price of such Stockholder's DPRX Options and (ii) taxes applicable to the exercise of such Stockholder's DPRX Options, (c) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an affiliated corporation, trust or other business entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (d) any transfer to another holder of the capital stock of PLx that has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (e) any transfer to a person if, as a condition precedent to the transfer, such person executes and delivers to PLx an agreement containing voting and transfer provisions with respect to the Shares so transferred that are substantially identical in all material respects to those set forth in this Agreement, and (f) as PLx may otherwise agree in writing in its sole discretion.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to DPRX and PLx as follows:
- (a) Stockholder has the full power and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder;
- (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of PLx and DPRX, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of law or a court of equity and by bankruptcy, insolvency and similar laws affecting creditors' rights and remedies generally;
- (c) except as set forth on <u>Schedule 1</u>, Stockholder beneficially owns the number of Shares indicated opposite such Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("<u>Liens</u>"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares and none of the Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares, except as contemplated by this Agreement;
- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his or her obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares pursuant to, any agreement, instrument, note, bond, mortgage, contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other entity, any bylaw or other organizational document of Stockholder; and
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any governmental or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his or her obligations under this Agreement in any material respect.
- 6. <u>Irrevocable Proxy.</u> Subject to the penultimate sentence of this <u>Section 6</u>, by execution of this Agreement, Stockholder does hereby appoint PLx with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of the undersigned's rights with respect to the Shares, to vote, or give consent with respect to, each of such Shares solely with respect to the matters set forth in <u>Section 1</u> hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date and hereby revokes any proxy previously granted by Stockholder with respect to the Shares. Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date. The Stockholder hereby revokes any proxies previously granted and represents that none of such previously-granted proxies are irrevocable.

7. No Solicitation. From and after the date hereof until the Expiration Date, Stockholder shall not (a) initiate, solicit, seek or knowingly encourage or support any inquiries, proposals or offers that constitute or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (b) engage or participate in, or knowingly facilitate, any discussions or negotiations regarding any inquiries, proposals or offers that constitute, or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (c) furnish to any Person other than PLx any non-public information that could reasonably be expected to be used for the purposes of formulating any DPRX Acquisition Proposal, (d) enter into any letter of intent, agreement in principle or other similar type of agreement relating to a DPRX Acquisition Proposal, or enter into any agreement or agreement in principle requiring DPRX to abandon, terminate or fail to consummate the transactions contemplated hereby, (e) initiate a stockholders' vote or action by consent of DPRX's stockholders with respect to a DPRX Acquisition Proposal, (f) except by reason of this Agreement, become a member of a "group" (as such term is defined in Section 13(d) of the Exchange Act) with respect to any voting securities of DPRX that takes any action in support of a DPRX Acquisition Proposal or (g) propose or agree to do any of the foregoing. In the event that Stockholder is a corporation, partnership, trust or other entity, it shall not permit any of its Subsidiaries or Affiliates to, undertake any of the actions contemplated by this Section 7.

8. Release; No Legal Actions.

The undersigned Stockholder acknowledges that the release of certain claims by stockholders of DPRX against PLx, DPRX, AcquireCo and their respective affiliates constitutes a material inducement for the completion of the transactions contemplated by the Merger Agreement and that PLx, DPRX and AcquireCo would not enter into the Merger Agreement without being released from such claims by the undersigned Stockholder. The undersigned Stockholder, and, to the extent within the undersigned's control, each of the undersigned's equity holders and each of their respective subsidiaries, affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns (each, a "Releasor"), hereby completely releases, acquits and forever discharges, to the fullest extent permitted by law, PLx, DPRX, AcquireCo, the Surviving Company and their respective affiliates and each of their respective current, former and future officers, directors, employees, agents, advisors, successors and assigns (each, a "Releasee"), from any and all losses, liabilities, suits, actions, debts or rights, whether fixed or contingent, known or unknown, matured or unmatured, arising out of, relating to, or in any manner connected with any facts, events or circumstances, or any actions taken, at or prior to the Merger Effective Time that any Releasor ever had or now has against the Releasees ("Released Matters"), excluding any rights of the Releasor under the Merger Agreement. Notwithstanding anything to the contrary in this Agreement, nothing herein shall release PLx or any of its Affiliates of obligations to the undersigned Stockholder with respect to (A) any employment or consulting agreement, (B) any other employment-related obligations of PLx or any of its Affiliates, (C) vested retirement benefits, (D) any rights that cannot be waived as a matter of law, (E) any indemnification obligations to the undersigned Stockholder under PLx's or any of its Affiliates' bylaws, certificate of incorporation, or other organizational documents, or under Delaware law or otherwise, or (F) any rights relating to the undersigned's relationship with PLx or any of its Affiliates (other than as a stockholder). The undersigned hereby waives the provisions of section 1542 of the California Civil Code, or any successor thereto, which currently states: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." Effective as of the Merger Effective Time, the undersigned Stockholder shall not, and, to the extent within the undersigned's control, shall not cause or permit its equity holders or any of their respective Subsidiaries, Affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns, to assert any claims against the Releasees in respect of any Released Matters. The undersigned Stockholder acknowledges that it would be difficult to fully compensate DPRX or any of its Affiliates (including the Surviving Company) for damages resulting from any breach by him/her/it of the provisions of this release. Accordingly, in the event of any actual or threatened breach of such provisions, DPRX and its Affiliates (including the Surviving Company) shall (in addition to any other remedies which it may have) be entitled to seek temporary and/or permanent injunctive relief to enforce such provisions and recover attorneys' fees and costs for same. The undersigned Stockholder further acknowledges that this release constitutes a material inducement to DPRX to complete the transactions contemplated by the Merger Agreement and DPRX will be relying on the enforceability of this release in completing such transactions contemplated by the Merger Agreement.

- (b) The Stockholder will not in his, her or its capacity as a stockholder of DPRX bring, commence, institute, maintain, prosecute or voluntarily aid any legal proceeding which (i) challenges the validity or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this agreement by the Stockholder, either alone or together with the other voting agreements and proxies to be delivered in connection with the execution of the Merger Agreement, or the approval of the Merger Agreement by the DPRX Board of Directors, constitutes a breach of any fiduciary duty of the DPRX Board of Directors or any member thereof.
- 9. Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at Law or in equity.
- 10. <u>Directors and Officers</u>. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of DPRX and/or holder of options to purchase DPRX Shares and not in such Stockholder's capacity as a director, officer or employee of DPRX or any of its Subsidiaries or in such Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require Stockholder to attempt to) prohibit, limit, prevent, preclude or restrict a director and/or officer of DPRX in the exercise of his or her fiduciary duties consistent with the terms of the Merger Agreement as a director and/or officer of DPRX or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director and/or officer, trustee or fiduciary of any employee benefit plan or trust from taking any action in his or her capacity as such director, officer, trustee and/or fiduciary.

- 11. <u>No Ownership Interest</u>. Nothing contained in this Agreement shall be deemed to vest in PLx any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Stockholder, and PLx does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of DPRX or exercise any power or authority to direct Stockholder in the voting of any of the Shares, except as otherwise provided herein.
- 12. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, nothing set forth in this <u>Section 12</u> or elsewhere in this Agreement shall relieve either party hereto from any liability, or otherwise limit the liability of either party from any liability for any intentional breach of any obligation or other provision contained in this Agreement.
- 13. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as PLx or DPRX may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Merger Agreement.
- 14. <u>Disclosure</u>. Stockholder hereby agrees that DPRX and PLx may publish and disclose in the Joint Proxy Statement, any prospectus filed with any regulatory authority in connection with the Merger and any related documents filed with such regulatory authority and as otherwise required by Law, such Stockholder's identity and ownership of Shares and the nature of such Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to the Joint Proxy Statement or prospectus or in any other filing made by DPRX or PLx as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Merger.
- 15. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) or by facsimile transmission (providing confirmation of transmission) to PLx or DPRX, as the case may be, in accordance with <u>Section 9.1</u> of the Merger Agreement and to each Stockholder at its address set forth on <u>Schedule 1</u> attached hereto (or at such other address for a party as shall be specified by like notice).
- 16. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- 17. <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided, however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 18. No Waivers. Except as set forth in Section 23, no waivers of any breach of this Agreement extended by PLx or DPRX to Stockholder shall be construed as a waiver of any rights or remedies of PLx or DPRX, as applicable, with respect to any other stockholder of DPRX who has executed an agreement substantially in the form of this Agreement with respect to Shares held or subsequently held by such stockholder or with respect to any subsequent breach of the Stockholder or any other such stockholder of DPRX. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- Applicable Law; Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (i) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (ii) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (i) of this Section 19, (iii) waives any objection to laying venue in any such action or proceeding in such courts, (iv) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (v) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 15 of this Agreement.
- 20. <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to trial by jury with respect to any action or proceeding related to or arising out of this Agreement, any document executed in connection herewith and the matters contemplated hereby and thereby.
- 21. <u>No Agreement Until Executed.</u> Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until (a) the DPRX Board of Directors has approved, for purposes of any applicable anti-takeover laws and regulations and any applicable provision of the DPRX certificate of incorporation, the transactions contemplated by the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.

- 22. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by facsimile or electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.
- 23. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 24. <u>Definition of Merger Agreement</u>. For purposes of this Agreement, the term "<u>Merger Agreement</u>" may include such agreement as amended or modified as long as such amendments or modifications (a) do not (i) change the form of consideration or (ii) change the Equity Exchange Ratio in a manner adverse to Stockholder, or (b) have been agreed to in writing by Stockholder.

25. Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page has Intentionally Been Left Blank]

EXECUTED as of the date first above written.

STOCKHOLDER

By:

Name:

Title:

[Signature Page to Voting Agreement]

DIPEXIUM PHARMACEUTICALS, INC.

By:

Name:

PLX PHARMA INC.

By:

Name:

Title:

Signature Page to Voting Agreement]

EXHIBIT B

Form of PLx Voting Agreement

VOTING AGREEMENT

among:

PLX PHARMA INC., a Delaware corporation;

DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation; and

the undersigned Stockholder

Dated as of December [●], 2016

TABLE OF CONTENTS

1.	Agreement to Vote Shares	1
2.	Expiration Date	2
3.	Additional Purchases	2
4.	Agreement to Retain Shares	2
5.	Representations and Warranties of Stockholder	3
6.	Irrevocable Proxy	3
7.	No Solicitation	4
8.	Waiver of Appraisal Rights; Release; No Legal Actions	4
9.	Other Remedies; Specific Performance	5
10.	Directors and Officers	5
11.	No Ownership Interest	6
12.	Termination	6
13.	Further Assurances	6
14.	Disclosure	6
15.	Notice	6
16.	Severability	6
17.	Assignability	7
18.	No Waivers	7
19.	Applicable Law; Jurisdiction	7
20.	Waiver of Jury Trial	7
21.	No Agreement Until Executed	7
22.	Entire Agreement; Counterparts; Exchanges by Facsimile	7
23.	Amendment	8
24.	Definition of Merger Agreement	8
25.	Construction	8
	i	

VOTING AGREEMENT

THIS VOTING AGREEMENT ("Agreement"), dated as of December [__], 2016, is made by and among Dipexium Pharmaceuticals, Inc., a Delaware corporation ("DPRX"), PLx Pharma Inc., a Delaware corporation ("PLx"), and the undersigned holder ("Stockholder") of shares of capital stock (the "Shares") of PLx.

WHEREAS, DPRX, Dipexium Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of DPRX ("<u>AcquireCo</u>"), and PLx, have entered into an Agreement and Plan of Merger and Reorganization, dated of even date herewith (in the form in effect on the date hereof and attached hereto as <u>Exhibit A</u> or as amended pursuant to <u>Section 24</u> hereof, the "<u>Merger Agreement</u>"), providing for the merger of AcquireCo with and into PLx (the "<u>Merger</u>");

WHEREAS, Stockholder beneficially owns and has sole or shared voting power with respect to the number of Shares, and holds stock options or other rights to acquire the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of DPRX, AcquireCo and PLx to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, DPRX's, AcquireCo's and PLx's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the expenses incurred and to be incurred by them in connection therewith, Stockholder, DPRX and PLx agree as follows:

- 1. <u>Agreement to Vote Shares.</u> Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of PLx or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of PLx, with respect to approval of the Merger as contemplated by the Merger Agreement and adoption of the Merger Agreement or any PLx Acquisition Proposal, Stockholder shall:
- (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat for purposes of calculating a quorum; and
- (b) vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that such Stockholder shall be entitled to so vote: (i) in favor of adoption of the Merger Agreement and the approval of the Merger; (ii) against any action, proposal, transaction or agreement that, to the knowledge of Stockholder, would reasonably be expected to result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of PLx under the Merger Agreement or that would reasonably be expected to result in any of the conditions to DPRX's, AcquireCo's or PLx's obligations under the Merger Agreement not being fulfilled; and (iii) against any PLx Acquisition Proposal, or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement. The Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.

- 2. <u>Expiration Date</u>. As used in this Agreement, the term "<u>Expiration Date</u>" shall mean the earlier to occur of (a) the Merger Effective Time, (b) such date and time as the Merger Agreement shall be terminated pursuant to <u>Article VII</u> thereof or otherwise, or (c) upon mutual written agreement of the parties to terminate this Agreement. Upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; provided, however, such termination or expiration shall not relieve any party from liability for any willful breach of this Agreement or acts of bad faith prior to termination hereof.
- 3. <u>Additional Purchases.</u> Stockholder agrees that any shares of capital stock or other equity securities of PLx that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any stock options or otherwise ("New Shares"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Agreement to Retain Shares. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below) on) any Shares, (b) deposit any Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens on) any Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing, Stockholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee and transferee shall sign a voting agreement in substantially the form hereof, (b) with respect to such Stockholder's PLx Options which expire on or prior to the Expiration Date, transfers, sales, or other dispositions of Shares to PLx as payment for the (i) exercise price of such Stockholder's PLx Options and (ii) taxes applicable to the exercise of such Stockholder's PLx Options. (c) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an affiliated corporation, trust or other business entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (d) any transfer to another holder of the capital stock of PLx that has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (e) any transfer to a person if, as a condition precedent to the transfer, such person executes and delivers to DPRX an agreement containing voting and transfer provisions with respect to the Shares so transferred that are substantially identical in all material respects to those set forth in this Agreement, and (f) as DPRX may otherwise agree in writing in its sole discretion.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to DPRX and PLx as follows:
- (a) Stockholder has the full power and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder;
- (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of PLx and DPRX, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of law or a court of equity and by bankruptcy, insolvency and similar laws affecting creditors' rights and remedies generally;
- (c) except as set forth on <u>Schedule 1</u>, Stockholder beneficially owns the number of Shares indicated opposite such Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("<u>Liens</u>"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares and none of the Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares, except as contemplated by this Agreement;
- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his or her obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares pursuant to, any agreement, instrument, note, bond, mortgage, contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other entity, any bylaw or other organizational document of Stockholder; and
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any governmental or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his or her obligations under this Agreement in any material respect.
- 6. <u>Irrevocable Proxy.</u> Subject to the penultimate sentence of this Section 6, by execution of this Agreement, Stockholder does hereby appoint DPRX with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of the undersigned's rights with respect to the Shares, to vote, or give consent with respect to, each of such Shares solely with respect to the matters set forth in <u>Section 1</u> hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date and hereby revokes any proxy previously granted by Stockholder with respect to the Shares. Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date. The Stockholder hereby revokes any proxies previously granted and represents that none of such previously-granted proxies are irrevocable.

No Solicitation. From and after the date hereof until the Expiration Date, Stockholder shall not (a) initiate, solicit, seek or knowingly encourage or support any inquiries, proposals or offers that constitute or may reasonably be expected to lead to, a PLx Acquisition Proposal, (b) engage or participate in, or knowingly facilitate, any discussions or negotiations regarding any inquiries, proposals or offers that constitute, or may reasonably be expected to lead to, a PLx Acquisition Proposal, (c) furnish to any Person other than PLx any non-public information that could reasonably be expected to be used for the purposes of formulating any PLx Acquisition Proposal, (d) enter into any letter of intent, agreement in principle or other similar type of agreement relating to a PLx Acquisition Proposal, or enter into any agreement or agreement in principle requiring PLx to abandon, terminate or fail to consummate the transactions contemplated hereby, (e) initiate a stockholders' vote or action by consent of PLx's stockholders with respect to a PLx Acquisition Proposal, (f) except by reason of this Agreement, become a member of a "group" (as such term is defined in Section 13(d) of the Exchange Act) with respect to any voting securities of PLx that takes any action in support of a PLx Acquisition Proposal or (g) propose or agree to do any of the foregoing. In the event that Stockholder is a corporation, partnership, trust or other entity, it shall not permit any of its Subsidiaries or Affiliates to, nor shall it authorize any officer, director or representative of Stockholder, or any of its Subsidiaries or Affiliates to, undertake any of the actions contemplated by this Section 7.

8. <u>Waiver of Appraisal Rights; Release; No Legal Actions.</u>

- (a) The Stockholder hereby waives, and agrees not to exercise or assert, any appraisal rights under applicable law, including Section 262 of the DGCL in connection with the Merger.
- The undersigned Stockholder acknowledges that the release of certain claims by stockholders of PLx against PLx, DPRX. AcquireCo and their respective affiliates constitutes a material inducement for the completion of the transactions contemplated by the Merger Agreement and that PLx, DPRX and AcquireCo would not enter into the Merger Agreement without being released from such claims by the undersigned Stockholder. The undersigned Stockholder, and, to the extent within the undersigned's control, each of the undersigned's equity holders and each of their respective subsidiaries, affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns (each, a "Releasor"), hereby completely releases, acquits and forever discharges, to the fullest extent permitted by law, PLx, DPRX, AcquireCo, the Surviving Company and their respective affiliates and each of their respective current, former and future officers, directors, employees, agents, advisors, successors and assigns (each, a "Releasee"), from any and all losses, liabilities, suits, actions, debts or rights, whether fixed or contingent, known or unknown, matured or unmatured, arising out of, relating to, or in any manner connected with any facts, events or circumstances, or any actions taken, at or prior to the Merger Effective Time that any Releasor ever had or now has against the Releasees ("Released Matters"), excluding any rights of the Releasor under the Merger Agreement. Notwithstanding anything to the contrary in this Agreement, nothing herein shall release PLx or any of its Affiliates of obligations to the undersigned Stockholder with respect to (A) any employment or consulting agreement, (B) any other employment-related obligations of PLx or any of its Affiliates, (C) vested retirement benefits, (D) any rights that cannot be waived as a matter of law, (E) any indemnification obligations to the undersigned Stockholder under PLx's or any of its Affiliates' bylaws, certificate of incorporation, or other organizational documents, or under Delaware law or otherwise, or (F) any rights relating to the undersigned's relationship with PLx or any of its Affiliates (other than as a stockholder). The undersigned hereby waives the provisions of section 1542 of the California Civil Code, or any successor thereto, which currently states: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." Effective as of the Merger Effective Time, the undersigned Stockholder shall not, and, to the extent within the undersigned's control, shall not cause or permit its equity holders or any of their respective Subsidiaries, Affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns, to assert any claims against the Releasees in respect of any Released Matters. The undersigned Stockholder acknowledges that it would be difficult to fully compensate DPRX or any of its Affiliates (including the Surviving Company) for damages resulting from any breach by him/her/it of the provisions of this release. Accordingly, in the event of any actual or threatened breach of such provisions, DPRX and its Affiliates (including the Surviving Company) shall (in addition to any other remedies which it may have) be entitled to seek temporary and/or permanent injunctive relief to enforce such provisions and recover attorneys' fees and costs for same. The undersigned Stockholder further acknowledges that this release constitutes a material inducement to DPRX to complete the transactions contemplated by the Merger Agreement and DPRX will be relying on the enforceability of this release in completing such transactions contemplated by the Merger Agreement.

- (c) The Stockholder will not in his, her or its capacity as a stockholder of PLx bring, commence, institute, maintain, prosecute or voluntarily aid any legal proceeding which (i) challenges the validity or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this agreement by the Stockholder, either alone or together with the other voting agreements and proxies to be delivered in connection with the execution of the Merger Agreement, or the approval of the Merger Agreement by PLx Board of Directors, constitutes a breach of any fiduciary duty of the PLx Board of Directors or any member thereof.
- 9. Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at Law or in equity.
- 10. <u>Directors and Officers</u>. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of PLx and/or holder of options to purchase PLx Shares and/or holder of securities convertible into PLx Shares and not in such Stockholder's capacity as a director, officer or employee of PLx or any of its Subsidiaries or in such Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust.

- 11. <u>No Ownership Interest</u>. Nothing contained in this Agreement shall be deemed to vest in DPRX any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Stockholder, and DPRX does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of PLx or exercise any power or authority to direct Stockholder in the voting of any of the Shares, except as otherwise provided herein.
- 12. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, nothing set forth in this <u>Section 12</u> or elsewhere in this Agreement shall relieve either party hereto from any liability, or otherwise limit the liability of either party from any liability for any intentional breach of any obligation or other provision contained in this Agreement.
- 13. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as PLx or DPRX may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Merger Agreement.
- 14. <u>Disclosure</u>. Stockholder hereby agrees that DPRX and PLx may publish and disclose in the Joint Proxy Statement, any prospectus filed with any regulatory authority in connection with the Merger and any related documents filed with such regulatory authority and as otherwise required by Law, such Stockholder's identity and ownership of Shares and the nature of such Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to the Joint Proxy Statement or prospectus or in any other filing made by DPRX or PLx as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Merger.
- 15. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) or by facsimile transmission (providing confirmation of transmission) to PLx or DPRX, as the case may be, in accordance with <u>Section 9.1</u> of the Merger Agreement and to each Stockholder at its address set forth on Schedule 1 attached hereto (or at such other address for a party as shall be specified by like notice).
- 16. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- 17. <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; provided, however, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 18. No Waivers. Except as set forth in Section 23, no waivers of any breach of this Agreement extended by PLx or DPRX to Stockholder shall be construed as a waiver of any rights or remedies of PLx or DPRX, as applicable, with respect to any other stockholder of PLx who has executed an agreement substantially in the form of this Agreement with respect to Shares held or subsequently held by such stockholder or with respect to any subsequent breach of the Stockholder or any other such stockholder of PLx. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- 19. Applicable Law; Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (i) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (ii) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (i) of this Section 19, (iii) waives any objection to laying venue in any such action or proceeding in such courts, (iv) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (v) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 15 of this Agreement.
- 20. <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to trial by jury with respect to any action or proceeding related to or arising out of this Agreement, any document executed in connection herewith and the matters contemplated hereby and thereby.
- 21. <u>No Agreement Until Executed.</u> Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until (a) the PLx Board of Directors has approved, for purposes of any applicable anti-takeover laws and regulations and any applicable provision of PLx certificate of incorporation, the transactions contemplated by the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.
- 22. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by facsimile or electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.

- 23. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 24. <u>Definition of Merger Agreement</u>. For purposes of this Agreement, the term "<u>Merger Agreement</u>" may include such agreement as amended or modified as long as such amendments or modifications (a) do not (i) change the form of consideration or (ii) change the Equity Exchange Ratio in a manner adverse to Stockholder, or (b) have been agreed to in writing by Stockholder.

25. Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page has Intentionally Been Left Blank]

EXECUTED as of the date first above written.

STOCKHOLDER

By:

Name:

Title:

[Signature Page to Company Voting Agreement]

EXECUTED as of the date first above written.

DIPEXIUM PHARMACEUTICALS, INC.

By:

Name:

PLX PHARMA INC.

By:

Name:

Title:

Title:

[Signature Page to Company Voting Agreement]

EXHIBIT C

Form of PLx Lock-Up Agreement

LOCK-UP AGREEMENT

Dipexium Pharmaceuticals, Inc. 14 Wall Street, Suite 3D New York, New York 10005

PLx Pharma Inc. 8285 El Rio, Suite 130 Houston, Texas 77054

Ladies and Gentlemen:

In connection with the proposed acquisition of PLx Pharma Inc. (the "PLx") by Dipexium Pharmaceuticals, Inc. ("DPRX") whereby Dipexium Acquisition Corp. ("AcquireCo"), a wholly- owned subsidiary of DPRX, will merge with and into the PLx (the "Merger"), and in consideration of DPRX, AcquireCo and PLx entering into the Agreement and Plan of Merger and Reorganization dated on or about December 22, 2016 (the "Merger Agreement;" all capitalized terms used in this Lock-Up Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement), the receipt and sufficiency of such consideration being hereby acknowledged and accepted, and in order to induce DPRX and PLx each to close the Merger, the undersigned ("Securityholder"), a holder of PLx Shares and/or PLx Convertible Debt (the "PLx Securities") who will receive DPRX Shares in exchange for his, her or its PLx Securities hereby agrees with DPRX and PLx as follows:

During the Lock-Up Period (as defined below), Securityholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, or otherwise dispose of (whether by actual disposition or effective economic disposition due to cash settlement or otherwise, including, without limitation, by the creation of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("Liens") on) any (i) PLx Securities and (ii) DPRX Shares and any securities convertible into, exchangeable for or that represent the right to receive DPRX Shares, in each case whether now owned or hereinafter acquired, owned directly by the Securityholder (including holding as a custodian) or with respect to which the Securityholder has beneficial ownership within the rules and regulations of the Securities and Exchange Commission (collectively, the "Locked-Up Securities"), (b) effect any short sale or enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens or by establishing or increasing a put equivalent position or liquidating or decreasing a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any Locked-Up Securities, or publicly announce an intention to effect any such transaction, during the Lock-Up Period) any Locked-Up Securities, or (c) take any action that would make any representation or warranty of Securityholder contained herein untrue or incorrect or have the effect of preventing or disabling Securityholder from performing Securityholder's obligations under this Lock-Up Agreement. Notwithstanding the foregoing, and provided that transfers described in (a) through (e) of this sentence are not required to be reported in any public report or filing with the Securities and Exchange Commission (other than (i) a filing at any time on a Form 5 or (ii) a filing after the expiration of the Lock-Up Period on a Schedule 13D or Schedule 13G (or Schedule 13D/A or Schedule 13G/ A), Securityholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Lock-Up Agreement shall bind the transferee, (b) with respect to such Securityholder's PLx Options which expire on or prior to the Merger Effective Time, transfers, sales, or other dispositions of Locked-Up Securities to PLx as payment for the (i) exercise price of such Securityholder's PLx Options and (ii) taxes applicable to the exercise of such Securityholder's PLx Options, (c) if Securityholder is a partnership or limited liability company, a transfer to one or more partners or members of Securityholder or to an affiliated corporation, trust or other business entity under common control with Securityholder, or if Securityholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a lock-up agreement in substantially the form hereof, (d) any transfer to another holder of the capital stock of PLx that has signed a lock-up agreement in substantially the form hereof relating to the transferred PLx Securities and (e) transfers of shares acquired on the open market following the Closing Date. In the event that any securities held by any other holder (the "Other Holder") subject to a similar lock-up agreement in connection with the Merger are released from the restrictions of such lock-up agreement, the Locked-Up Securities held by the undersigned Securityholder shall likewise automatically be released from the restrictions contained herein in the same proportion as those securities released for the Other Holder.

- 2. As used in this Lock-Up Agreement, the term "Lock-Up Period" shall mean from and after the date hereof until the earlier to occur of (a) 120 days after the Closing Date or (b) such date and time as the Merger Agreement shall be terminated pursuant to Article VII thereof or otherwise. Upon termination or expiration of this Lock-Up Agreement, no party shall have any further obligations or liabilities under this Lock-Up Agreement; *provided*, *however*, such termination or expiration shall not relieve any party from liability for any willful breach of this Lock-Up Agreement or acts of bad faith prior to termination hereof.
- 3. Securityholder also agrees and consents to the entry of stop transfer instructions with DPRX's transfer agent and registrar against the transfer of Securityholders' Locked-Up Securities, except in compliance with this Lock-Up Agreement. In furtherance of the foregoing, DPRX and its transfer agent are hereby authorized to decline to make any transfer of securities if such transfer would constitute a violation or breach of this Lock-Up Agreement.
- 4. Securityholder understands that DPRX, PLx and AcquireCo will proceed with the Merger in reliance on this Lock-Up Agreement. Moreover, Securityholder understands and agrees that DPRX, AcquireCo and PLx are relying upon the accuracy, completeness, and truth of Securityholder's representations, warranties, agreements, and certifications contained in this Lock-Up Agreement.

[Remainder of Page has Intentionally Been Left Blank; Signature Page Follows]

Date:	, 2016	
		Very truly yours,
If an individual	l, please sign here:	
	Signatu	re:
	Print Na	me:
If a corporation	n, a limited partnership or other legal entity, plea	se sign here:
	Legal N	ame:
		By:
		Name: Title:
	[Signature Pag	e to Lock-Up Agreement]

Date:	, 2016	
		DIPEXIUM PHARMACEUTICALS, INC.
		By:
		Name:
		Title:
		PLX PHARMA INC.
		By:
		Name:
		Title:
		[Signature Page to PLx Lock-Up Agreement]

EXHIBIT D

Form of DPRX Lock-Up Agreement

LOCK-UP AGREEMENT

Dipexium Pharmaceuticals, Inc. 14 Wall Street, Suite 3D New York, New York 10005

PLx Pharma Inc. 8285 El Rio, Suite 130 Houston, Texas 77054

Ladies and Gentlemen:

In connection with the proposed acquisition of PLx Pharma Inc. (the "PLx") by Dipexium Pharmaceuticals, Inc. ("DPRX") whereby Dipexium Acquisition Corp. ("AcquireCo"), a wholly- owned subsidiary of DPRX, will merge with and into the PLx (the "Merger"), and in consideration of DPRX, AcquireCo and PLx entering into the Agreement and Plan of Merger and Reorganization dated on or about December 22, 2016 (the "Merger Agreement;" all capitalized terms used in this Lock-Up Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement), the receipt and sufficiency of such consideration being hereby acknowledged and accepted, and in order to induce DPRX and PLx each to close the Merger, the undersigned ("Securityholder"), a holder of DPRX Shares (the "DPRX Securities") hereby agrees with DPRX and PLx as follows:

During the Lock-Up Period (as defined below), Securityholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, or otherwise dispose of (whether by actual disposition or effective economic disposition due to cash settlement or otherwise. including, without limitation, by the creation of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("Liens") on) any (i) DPRX Securities and (ii) any securities convertible into, exchangeable for or that represent the right to receive DPRX Shares, in each case whether now owned or hereinafter acquired, owned directly by the Securityholder (including holding as a custodian) or with respect to which the Securityholder has beneficial ownership within the rules and regulations of the Securities and Exchange Commission (collectively, the "Locked-Up Securities"), (b) effect any short sale or enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens or by establishing or increasing a put equivalent position or liquidating or decreasing a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any Locked-Up Securities, or publicly announce an intention to effect any such transaction, during the Lock-Up Period) any Locked-Up Securities, or (c) take any action that would make any representation or warranty of Securityholder contained herein untrue or incorrect or have the effect of preventing or disabling Securityholder from performing Securityholder's obligations under this Lock-Up Agreement. Notwithstanding the foregoing, and provided that transfers described in (a) through (e) of this sentence are not required to be reported in any public report or filing with the Securities and Exchange Commission (other than (i) a filing at any time on a Form 5 or (ii) a filing after the expiration of the Lock-Up Period on a Schedule 13D or Schedule 13G (or Schedule 13D/A or Schedule 13G/A), Securityholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Lock-Up Agreement shall bind the transferee, (b) with respect to such Securityholder's DPRX Options which expire on or prior to the Merger Effective Time, transfers, sales, or other dispositions of Locked-Up Securities to DPRX as payment for the (i) exercise price of such Securityholder's DPRX Options and (ii) taxes applicable to the exercise of such Securityholder's DPRX Options, (c) if Securityholder is a partnership or limited liability company, a transfer to one or more partners or members of Securityholder or to an affiliated corporation, trust or other business entity under common control with Securityholder, or if Securityholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a lock-up agreement in substantially the form hereof, (d) any transfer to another holder of the capital stock of DPRX that has signed a lock-up agreement in substantially the form hereof relating to the transferred DPRX Securities and (e) transfers of shares acquired on the open market following the Closing Date. In the event that any securities held by any other holder (the "Other Holder") subject to a similar lock-up agreement in connection with the Merger are released from the restrictions of such lock-up agreement, the Locked-Up Securities held by the undersigned Securityholder shall likewise automatically be released from the restrictions contained herein in the same proportion as those securities released for the Other Holder.

- 2. As used in this Lock-Up Agreement, the term "Lock-Up Period" shall mean from and after the date hereof until the earlier to occur of (a) 120 days after the Closing Date or (b) such date and time as the Merger Agreement shall be terminated pursuant to Article VII thereof or otherwise. Upon termination or expiration of this Lock-Up Agreement, no party shall have any further obligations or liabilities under this Lock-Up Agreement; *provided*, *however*, such termination or expiration shall not relieve any party from liability for any willful breach of this Lock-Up Agreement or acts of bad faith prior to termination hereof.
- 3. Securityholder also agrees and consents to the entry of stop transfer instructions with DPRX's transfer agent and registrar against the transfer of Securityholders' Locked-Up Securities, except in compliance with this Lock-Up Agreement. In furtherance of the foregoing, DPRX and its transfer agent are hereby authorized to decline to make any transfer of securities if such transfer would constitute a violation or breach of this Lock-Up Agreement.
- 4. Securityholder understands that DPRX, PLx and AcquireCo will proceed with the Merger in reliance on this Lock-Up Agreement. Moreover, Securityholder understands and agrees that DPRX, AcquireCo and PLx are relying upon the accuracy, completeness, and truth of Securityholder's representations, warranties, agreements, and certifications contained in this Lock-Up Agreement.

[Remainder of Page has Intentionally Been Left Blank; Signature Page Follows]

Date:	, 2016	
		Very truly yours,
If an individual	, please sign here:	
	Signature:	
	Print Name:	
If a corporation	, a limited partnership or other legal entity, please sign	here:
	Legal Name:	
	Ву:	
	Name Title:	
	[Signature Page to DPRX	Lock-Up Agreement]

Date:	, 2016	
		DIPEXIUM PHARMACEUTICALS, INC.
		By:
		Name:
		Title:
		PLX PHARMA INC.
		Ву:
		Name:
		Title:
		[Signature Page to DPRX Lock-Up Agreement]

EXHIBIT E

Form of Promissory Note

SECURED PROMISSORY NOTE

\$2,000,000.00 January [], 2017

PLX PHARMA INC., a Delaware corporation ("Maker"), promises to pay to the order of DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation ("Payee"), TWO MILLION DOLLARS (\$2,000,000.00), together with all accrued and unpaid interest at the rate and calculated as hereinafter provided.

- 1. **Merger Agreement.** This note (this "*Note*") is being made and entered into in connection with that certain Agreement and Plan of Merger and Reorganization, dated as of December 22, 2016, by and among Maker, Payee, and Dipexium Acquisition Corp., a Delaware corporation (the "*Merger Agreement*").
- 2. **Interest.** Interest shall accrue on the unpaid principal balance of this Note beginning on the date hereof (the "*Effective Date*") at the per annum rate equal to 8.0%, calculated on the basis of a year of 365 or 366 days, as the case may be, for the actual number of days elapsed.
- 3. **Maturity.** The entire outstanding principal balance of this Note, together with any accrued and unpaid interest thereon and any other sums due hereunder, shall be due and payable in full on the later of (a) October 15, 2017, or (b) the date that is two hundred seventy (270) days following the termination of the Merger Agreement (in either case, the "*Maturity Date*"); provided, that, notwithstanding the foregoing, in the event that (i) the Merger Agreement is terminated pursuant to Section 7.1(d)(iii) thereof, then the Maturity Date and the entire outstanding principal balance of this Note, together with any accrued and unpaid interest thereon and any other sums due hereunder, shall be due and payable on the date that is ninety (90) days following the date of such termination or (ii) the Maker, on or prior to the dates set forth in clauses (a) or (b) above, completes a financing of at least \$10.0 million or conducts a reorganization, consolidation, or merger of Maker pursuant to which the holders of Maker's securities immediately prior to such transaction beneficially own less than 50% of the outstanding voting securities of the surviving entity after the transaction or the consummation of the sale, lease, transfer, conveyance or other disposition in one or a series of transactions, of all or substantially all of Maker's assets, or Maker and its subsidiaries, taken as a whole, to any person or entity, then the Maturity Date and the entire outstanding principal balance of this Note, together with any accrued and unpaid interest thereon and any other sums due hereunder, shall be due and payable on the closing date of any such financing or transaction.
- 4. **Prepayment.** This Note may be prepaid in part or in whole at any time and from time to time without premium or penalty; provided, however, that any such prepayment shall be credited first to interest and then to the principal due.
- 5. **Costs and Expenses.** In addition to all principal and interest on this Note, Maker agrees to pay (a) all reasonable costs and expenses incurred by Payee in collecting this Note through any reorganization, bankruptcy or any other proceeding and (b) reasonable attorneys' fees when and if this Note is placed in the hands of an attorney for collection after default (together, the "*Collection Costs*").

6. Security Interest

(a) As collateral for Maker's obligations and indebtedness to Payee hereunder, Maker hereby grants to Payee a first priority perfected security interest in and lien on all right, title and interest of Maker in and to any and all of the following property and all other property whether now or hereafter existing or acquired, wherever located, all substitutions, replacements, accessions, products or proceeds (including, without limitation, insurance proceeds) of such property, wherever located and in whatever form, and all books and records pertaining to such property (the "Collateral"):

All Accounts; all cash and currency; all Chattel Paper; all Commercial Tort Claims; all Copyrights; all Copyright Licenses; all Deposit Accounts; all Documents; all Domain Names; all Equipment; all Fixtures; all General Intangibles; all Goods; all Instruments; all Inventory; all Investment Property; all Letter-of-Credit Rights; all Other Intellectual Property; all Patents Licenses; all Payment Intangibles; all Pledged Equity; all Proprietary Databases; all Proprietary Software; all Securities Accounts; all Software; all Supporting Obligations; all Trademarks; all Trademark Licenses; all Trade Secrets; all Websites and to the extent not otherwise included, all Accessions and all Proceeds of any and all of the foregoing.

Capitalized terms used in the above paragraph and not defined shall have the meanings set forth in the Uniform Commercial Code in effect from time to time in the State of New York (the "*UCC*").

Maker hereby authorizes Payee to file at any time financing statements, continuation statements, and amendments thereto that (i) either specifically describe the Collateral or describe the Collateral as all assets of Maker of the kind pledged hereunder, and (ii) contain any other information required by the UCC for the sufficiency of filing office acceptance of any financing statement, continuation statement, or amendment, including whether Maker is an organization, the type of organization and any organizational identification number issued to Maker, if applicable. Any such financing statements may be filed by Payee at any time in any jurisdiction whether or not Article 9 of the UCC is then in effect in that jurisdiction. Maker shall from time to time endorse and deliver to Payee, at the request of Payee, all documents that Payee may request, in form reasonably satisfactory to Payee, to perfect and continue perfection of Payee's security interests in the Collateral and in order to fully consummate all of the transactions contemplated hereby. Maker shall have possession of the Collateral, except where expressly otherwise provided in this Note, or, subject to the terms hereof, where Payee chooses to perfect its security interest by possession in addition to the filing of a financing statement. Where Collateral exceeding \$500,000 in value is in possession of a third party bailee or maintained at any leased location, Maker shall take such steps as Payee reasonably requests for Payee to (i) obtain an acknowledgement, in form and substance reasonably satisfactory to Maker, of the bailee or landlord that the bailee or landlord holds such Collateral for the benefit of Payee, and otherwise grant access to such property to Payee. Maker shall use commercially reasonable efforts to take such steps as Payee requests to obtain "control" of any Collateral consisting of investment property, deposit accounts, securities accounts, letter-of-credit rights or electronic chattel paper (as such items and term "control" are defined in Article 9 of the UCC) by causing the securities intermediary or depositary institution or issuing bank to execute a control agreement in form and substance reasonably satisfactory to Payee. Maker will not create any chattel paper with a value in excess, individually or in the aggregate, of \$500,000 without placing a legend on the chattel paper reasonably acceptable to Payee indicating that Payee has a security interest in the chattel paper.

- (c) Upon occurrence of a Security Interest Termination Event (as defined below), (i) any security interest created by this Note shall immediately cease to be effective, (ii) Payee shall take all actions reasonably necessary to terminate any financing statements filed by or on behalf of Payee in connection with the Collateral, and (iii) Payee authorizes Maker to take on its behalf all actions reasonably necessary to effect the foregoing.
- (d) For purposes of this Section 6, a "Security Interest Termination Event" means a termination of the Merger Agreement other than a termination pursuant to (i) Section 7.1(a), (ii) Section 7.1(b)(ii), (iii) Section 7.1(b)(iii), (iv) Section 7.1(b)(iv), (v) Section 7.1(d)(ii) and (vi) Section 7.1(d)(iii) thereof, which in each case this Note shall continue to be secured in favor of Payee in accordance with the terms of this Section 6.
- 7. **Default Rate.** All past due amounts under this Note shall bear interest at a rate (the "**Default Interest Rate**") equal to 10.0% per annum. During the continuance of an Event of Default (hereinafter defined) that continues beyond the cure period set forth in Section 9, the entire unpaid balance of principal shall automatically bear interest at the Default Interest Rate. It is the specific intent of the Maker and Payee that this Note bear a lawful rate of interest. If any court of competent jurisdiction should determine that the interest rates, or any other charges or costs set forth in this Note which may be deemed to be interest, exceed the amount(s) which is (are) statutorily permitted for the type of transaction evidenced hereby, the interest rates (and other applicable charges or costs) shall be reduced to the highest rate permitted by applicable law, with any excess interest theretofore collected being applied against principal or, if such principal has been fully repaid, returned to Maker on demand.
- 8. **Use of Proceeds**. Maker shall not permit more than \$500,000 of the proceeds of this Note to be used for purposes other than the following:
- (a) Aspertec development actions including, without limitation, the following: (i) execute a contract with Contract Manufacturing Organization and initiate efforts under such contract; (ii) begin HPLC method development; (iii) formalize key manufacturing and supplier agreements (including without limitation any contract packaging agreement) and pay legal fees related to the agreements; (iv) purchase long lead time tooling for packaging and equipment purchases; (v) acquire dearator; (vi) consulting support for various regulatory and manufacturing items; (vii) conduct supplier audits; and (viii) travel expenses reasonably required to execute the foregoing; and

- (b) Other actions or items as may be determined in advance by the mutual written agreement of Maker and Payee.
- 9. **Default.** If any of the following events (each an "Event of Default") occur then Payee may (a) by written notice to Maker, declare the outstanding principal balance of this Note, together with any accrued interest and other amounts hereunder to be due and payable and (b) pursue any remedies, in equity or under applicable law, including, without limitation, the right to collect any amounts equal to the unpaid amount of this Note, including all accrued and unpaid interest and other amounts hereunder, plus interest hereon at the Default Rate, and Collection Costs and exercising all rights and remedies available to it as a secured creditor, including, without limitation, the rights and remedies of secured creditors under the UCC:

(ii)

- (i) Maker fails to pay, when due, any payment of principal or interest on or with respect to the indebtedness evidenced by this Note;
 - Maker (i) applies for or consents to the appointment of a receiver, trustee, or liquidator of Maker or all or a substantial part of Maker's assets, or files a voluntary petition commencing a case under Title 11 of the United States Code, seeking liquidation, reorganization, or rearrangement or taking advantage of any bankruptcy, insolvency, debtor's relief, or other similar law of the United States, the State of Delaware, or any other jurisdiction (any of the foregoing an "*Insolvency Proceeding*"), (ii) makes a general assignment for the benefit of creditors, (iii) admits in writing its inability to pay its debts generally as they become due, or (iv) files an answer admitting the material allegations of a petition filed against it in any Insolvency Proceeding; or
- Any order, judgment or decree is entered against Maker by any court of competent jurisdiction or by any other duly authorized authority, on the petition of a creditor or otherwise, granting relief in any Insolvency Proceeding or approving a petition seeking reorganization or an arrangement of Maker's debts or appointing a receiver, trustee, conservator, custodian, or liquidator of Maker or all or any substantial part of Maker's assets, and such order, judgment or decree shall not be dismissed or stayed within 30 days.
- Waivers. No delay or omission on the part of Payee in exercising any right hereunder shall operate as a waiver of any such right or of any other right of Payee, nor shall any delay, omission or waiver on any one occasion be deemed a bar to or waiver of the same or any other right on any future occasion. Maker hereby waives presentment, demand, protest and notice and all other demands and notices of every kind in connection with the delivery, acceptance, performance, default or enforcement hereof and consents that as to time and terms of payment hereunder this Note may be extended or postponed and that no such extension, postponement or any other indulgence, and no substitution, exchange, release or surrender of collateral, if any, and no discharge or release of any other party primarily or secondarily liable hereon, shall discharge or otherwise affect the liability of Maker.

Notices. All notices, waivers and other communications required by, or permitted under, this Note shall be in writing and shall be deemed given when (i) delivered by hand (with written confirmation of receipt); (ii) sent by electronic mail (with confirmation of receipt), provided that a copy is mailed by registered mail, return receipt requested; or (iii) received by the addressee, if sent by a nationally recognized overnight delivery service (receipt requested); in each case to the appropriate addresses and facsimile numbers set forth below (or to such other addresses and facsimile numbers as a party may designate by written notice to the other party):

if to Maker:

PLx Pharma Inc. 8285 El Rio Street, Ste. 130 Houston, TX 77054 Attention: President and Chief Executive Officer E-mail: ngiordano@plxpharma.com

with a copy (which will not constitute notice) to:

Jackson Walker L.L.P. 2323 Ross Ave, Ste. 600 Dallas, TX 75202 Attention: Michael Laussade E-mail: mlaussade@jw.com

if to Payee:

Dipexium Pharmaceuticals, Inc.
14 Wall Street, Suite 3D
New York, New York 10005
Attention: President and Chief Executive Officer
E-mail: davidluci@dipexium.com

with a copy (which will not constitute notice) to: Mintz, Levin, Cohn, Ferris, Glovsky and Popeo, P.C. 666 Third Avenue New York, New York 10017 Attention: Ivan K. Blumenthal, Esq. E-mail: ikblumenthal@mintz.com

12. **Governing Law; Miscellaneous.** This Note shall be governed by and construed in accordance with the laws of the State of New York except to the extent that federal law preempts the laws of the State of New York. All sums referred to herein shall be calculated by reference to and payable in the lawful currency of the United States. This Note has been reviewed and negotiated by Maker and Payee at arms' length with the benefit of or opportunity to seek the assistance of legal counsel and shall not be construed against any such party. Any reference to Payee in this Note shall include any successor to or assignee of Payee.

13.	Waiver of Ju-	ry Trial. MAKER	HEREBY IRRE	EVOCABLY A	ND UNCONDI	TIONALLY	WAIVES A	LL RIGHTS
TO TRIAL BY	JURY IN ANY I	LEGAL ACTION, I	PROCEEDING	OR COUNTER	CLAIM BROU	GHT BY PAY	YEE AGAIN	IST MAKER
ON ANY MA	TTER WHATSOF	EVER ARISING O	UT OF, IN CON	NECTION WI	TH OR RELAT	ED TO THIS	NOTE.	

	14.	Successors and Assig	ns . This Note shall be	binding upon and	d shall inure to t	he benefit of the par	ties hereto a	nd their
respectiv	e succes	sors and permitted assi	gns. Maker shall not as	ssign its rights or	obligations her	eunder without the p	rior written	consent
of Payee,	which o	consent may be withhel	d, conditioned or delay	ed in Payee's sol	e discretion.			

[Signature on following page.]

	MAKER:
	PLX PHARMA INC.
	By: Name: Title:
AGREED TO AND ACKNOWLEDGED:	
DIPEXIUM PHARMACEUTICALS, INC.	
By: Name: Title:	-

VOTING AGREEMENT

among:

DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation;

PLX PHARMA INC., a Delaware corporation; and

the undersigned Stockholder

Dated as of December 22, 2016

TABLE OF CONTENTS

1.	Agreement to Vote Shares	1
2.	Expiration Date	2
3.	Additional Purchases	2
4.	Agreement to Retain Shares	2
5.	Representations and Warranties of Stockholder	3
6.	Irrevocable Proxy	3
7.	No Solicitation	4
8.	Release; No Legal Actions	4
9.	Other Remedies; Specific Performance	5
10.	Directors and Officers	5
11.	No Ownership Interest	6
12.	Termination	6
13.	Further Assurances	6
14.	Disclosure	6
15.	Notice	6
16.	Severability	6
17.	Assignability	7
18.	No Waivers	7
19.	Applicable Law; Jurisdiction	7
20.	Waiver of Jury Trial	7
21.	No Agreement Until Executed	7
22.	Entire Agreement; Counterparts; Exchanges by Facsimile	7
23.	Amendment	8
24.	Definition of Merger Agreement	8
25.	Construction	8
	:	

VOTING AGREEMENT

THIS VOTING AGREEMENT ("Agreement"), dated as of December 22, 2016, is made by and among Dipexium Pharmaceuticals, Inc., a Delaware corporation ("<u>DPRX</u>"), PLx Pharma Inc., a Delaware corporation ("<u>PLx</u>"), and the undersigned holder ("<u>Stockholder</u>") of shares of capital stock (the "<u>Shares</u>") of DPRX.

WHEREAS, DPRX, Dipexium Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of DPRX ("<u>AcquireCo</u>"), and PLx, have entered into an Agreement and Plan of Merger and Reorganization, dated of even date herewith (in the form in effect on the date hereof and attached hereto as <u>Exhibit A</u> or as amended pursuant to <u>Section 24</u> hereof, the "<u>Merger Agreement</u>"), providing for the merger of AcquireCo with and into PLx (the "<u>Merger</u>");

WHEREAS, Stockholder beneficially owns and has sole or shared voting power with respect to the number of Shares, and holds stock options or other rights to acquire the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of DPRX, AcquireCo and PLx to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, DPRX's, AcquireCo's and PLx's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the expenses incurred and to be incurred by them in connection therewith, Stockholder, DPRX and PLx agree as follows:

- 1. <u>Agreement to Vote Shares.</u> Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of DPRX or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of DPRX, with respect to approval of the Merger as contemplated by the Merger Agreement and adoption of the Merger Agreement or any DPRX Acquisition Proposal, Stockholder shall:
- (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat for purposes of calculating a quorum; and
- (b) vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that such Stockholder shall be entitled to so vote: (i) in favor of (A) the adoption of the Merger Agreement and the approval of the Merger, including without limitation the issuance of the DPRX Shares by virtue of the Merger as contemplated by the Merger Agreement, and (B) the DPRX Charter Amendment; (ii) against any action, proposal, transaction or agreement that, to the knowledge of Stockholder, would reasonably be expected to result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of DPRX under the Merger Agreement or that would reasonably be expected to result in any of the conditions to DPRX's, AcquireCo's or PLx's obligations under the Merger Agreement not being fulfilled; and (iii) against any DPRX Acquisition Proposal, or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement. The Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.

- 2. <u>Expiration Date</u>. As used in this Agreement, the term "<u>Expiration Date</u>" shall mean the earlier to occur of (a) the Merger Effective Time, (b) such date and time as the Merger Agreement shall be terminated pursuant to <u>Article VII</u> thereof or otherwise, or (c) upon mutual written agreement of the parties to terminate this Agreement. Upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; *provided, however*, such termination or expiration shall not relieve any party from liability for any willful breach of this Agreement or acts of bad faith prior to termination hereof.
- 3. <u>Additional Purchases</u>. Stockholder agrees that any shares of capital stock or other equity securities of DPRX that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any stock options or otherwise ("New Shares"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Agreement to Retain Shares. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below) on) any Shares, (b) deposit any Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens on) any Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing. Stockholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee and transferee shall sign a voting agreement in substantially the form hereof, (b) with respect to such Stockholder's DPRX Options which expire on or prior to the Expiration Date, transfers, sales, or other dispositions of Shares to DPRX as payment for the (i) exercise price of such Stockholder's DPRX Options and (ii) taxes applicable to the exercise of such Stockholder's DPRX Options, (c) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an affiliated corporation, trust or other business entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (d) any transfer to another holder of the capital stock of PLx that has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (e) any transfer to a person if, as a condition precedent to the transfer, such person executes and delivers to PLx an agreement containing voting and transfer provisions with respect to the Shares so transferred that are substantially identical in all material respects to those set forth in this Agreement, and (f) as PLx may otherwise agree in writing in its sole discretion.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to DPRX and PLx as follows:
- (a) Stockholder has the full power and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder;
- (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of PLx and DPRX, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of law or a court of equity and by bankruptcy, insolvency and similar laws affecting creditors' rights and remedies generally;
- (c) except as set forth on <u>Schedule 1</u>, Stockholder beneficially owns the number of Shares indicated opposite such Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("<u>Liens</u>"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares and none of the Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares, except as contemplated by this Agreement;
- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his or her obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares pursuant to, any agreement, instrument, note, bond, mortgage, contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other entity, any bylaw or other organizational document of Stockholder; and
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any governmental or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his or her obligations under this Agreement in any material respect.
- 6. <u>Irrevocable Proxy.</u> Subject to the penultimate sentence of this <u>Section 6</u>, by execution of this Agreement, Stockholder does hereby appoint PLx with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of the undersigned's rights with respect to the Shares, to vote, or give consent with respect to, each of such Shares solely with respect to the matters set forth in <u>Section 1</u> hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date and hereby revokes any proxy previously granted by Stockholder with respect to the Shares. Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date. The Stockholder hereby revokes any proxies previously granted and represents that none of such previously-granted proxies are irrevocable.

No Solicitation. From and after the date hereof until the Expiration Date, Stockholder shall not (a) initiate, solicit, seek or knowingly encourage or support any inquiries, proposals or offers that constitute or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (b) engage or participate in, or knowingly facilitate, any discussions or negotiations regarding any inquiries, proposals or offers that constitute, or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (c) furnish to any Person other than PLx any non-public information that could reasonably be expected to be used for the purposes of formulating any DPRX Acquisition Proposal, (d) enter into any letter of intent, agreement in principle or other similar type of agreement relating to a DPRX Acquisition Proposal, or enter into any agreement or agreement in principle requiring DPRX to abandon, terminate or fail to consummate the transactions contemplated hereby, (e) initiate a stockholders' vote or action by consent of DPRX's stockholders with respect to a DPRX Acquisition Proposal, (f) except by reason of this Agreement, become a member of a "group" (as such term is defined in Section 13(d) of the Exchange Act) with respect to any voting securities of DPRX that takes any action in support of a DPRX Acquisition Proposal or (g) propose or agree to do any of the foregoing. In the event that Stockholder is a corporation, partnership, trust or other entity, it shall not permit any of its Subsidiaries or Affiliates to, nor shall it authorize any officer, director or representative of Stockholder, or any of its Subsidiaries or Affiliates to, undertake any of the actions contemplated by this Section 7.

8. Release; No Legal Actions.

The undersigned Stockholder acknowledges that the release of certain claims by stockholders of DPRX against PLx, DPRX, AcquireCo and their respective affiliates constitutes a material inducement for the completion of the transactions contemplated by the Merger Agreement and that PLx, DPRX and AcquireCo would not enter into the Merger Agreement without being released from such claims by the undersigned Stockholder. The undersigned Stockholder, and, to the extent within the undersigned's control, each of the undersigned's equity holders and each of their respective subsidiaries, affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns (each, a "Releasor"), hereby completely releases, acquits and forever discharges, to the fullest extent permitted by law, PLx, DPRX, AcquireCo, the Surviving Company and their respective affiliates and each of their respective current, former and future officers, directors, employees, agents, advisors, successors and assigns (each, a "Releasee"), from any and all losses, liabilities, suits, actions, debts or rights, whether fixed or contingent, known or unknown, matured or unmatured, arising out of, relating to, or in any manner connected with any facts, events or circumstances, or any actions taken, at or prior to the Merger Effective Time that any Releasor ever had or now has against the Releasees ("Released Matters"), excluding any rights of the Releasor under the Merger Agreement. Notwithstanding anything to the contrary in this Agreement, nothing herein shall release PLx or any of its Affiliates of obligations to the undersigned Stockholder with respect to (A) any employment or consulting agreement, (B) any other employment-related obligations of PLx or any of its Affiliates, (C) vested retirement benefits, (D) any rights that cannot be waived as a matter of law, (E) any indemnification obligations to the undersigned Stockholder under PLx's or any of its Affiliates' bylaws, certificate of incorporation, or other organizational documents, or under Delaware law or otherwise, or (F) any rights relating to the undersigned's relationship with PLx or any of its Affiliates (other than as a stockholder). The undersigned hereby waives the provisions of section 1542 of the California Civil Code, or any successor thereto, which currently states: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." Effective as of the Merger Effective Time, the undersigned Stockholder shall not, and, to the extent within the undersigned's control, shall not cause or permit its equity holders or any of their respective Subsidiaries, Affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns, to assert any claims against the Releasees in respect of any Released Matters. The undersigned Stockholder acknowledges that it would be difficult to fully compensate DPRX or any of its Affiliates (including the Surviving Company) for damages resulting from any breach by him/her/it of the provisions of this release. Accordingly, in the event of any actual or threatened breach of such provisions, DPRX and its Affiliates (including the Surviving Company) shall (in addition to any other remedies which it may have) be entitled to seek temporary and/or permanent injunctive relief to enforce such provisions and recover attorneys' fees and costs for same. The undersigned Stockholder further acknowledges that this release constitutes a material inducement to DPRX to complete the transactions contemplated by the Merger Agreement and DPRX will be relying on the enforceability of this release in completing such transactions contemplated by the Merger Agreement.

- (b) The Stockholder will not in his, her or its capacity as a stockholder of DPRX bring, commence, institute, maintain, prosecute or voluntarily aid any legal proceeding which (i) challenges the validity or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this agreement by the Stockholder, either alone or together with the other voting agreements and proxies to be delivered in connection with the execution of the Merger Agreement, or the approval of the Merger Agreement by the DPRX Board of Directors, constitutes a breach of any fiduciary duty of the DPRX Board of Directors or any member thereof.
- 9. Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at Law or in equity.
- 10. <u>Directors and Officers</u>. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of DPRX and/or holder of options to purchase DPRX Shares and not in such Stockholder's capacity as a director, officer or employee of DPRX or any of its Subsidiaries or in such Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require Stockholder to attempt to) prohibit, limit, prevent, preclude or restrict a director and/or officer of DPRX in the exercise of his or her fiduciary duties consistent with the terms of the Merger Agreement as a director and/or officer of DPRX or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director and/or officer of DPRX or any trustee or fiduciary of any employee benefit plan or trust from taking any action in his or her capacity as such director, officer, trustee and/or fiduciary.

- 11. <u>No Ownership Interest</u>. Nothing contained in this Agreement shall be deemed to vest in PLx any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Stockholder, and PLx does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of DPRX or exercise any power or authority to direct Stockholder in the voting of any of the Shares, except as otherwise provided herein.
- 12. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, nothing set forth in this <u>Section 12</u> or elsewhere in this Agreement shall relieve either party hereto from any liability, or otherwise limit the liability of either party from any liability for any intentional breach of any obligation or other provision contained in this Agreement.
- 13. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as PLx or DPRX may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Merger Agreement.
- 14. <u>Disclosure</u>. Stockholder hereby agrees that DPRX and PLx may publish and disclose in the Joint Proxy Statement, any prospectus filed with any regulatory authority in connection with the Merger and any related documents filed with such regulatory authority and as otherwise required by Law, such Stockholder's identity and ownership of Shares and the nature of such Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to the Joint Proxy Statement or prospectus or in any other filing made by DPRX or PLx as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Merger.
- 15. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) or by facsimile transmission (providing confirmation of transmission) to PLx or DPRX, as the case may be, in accordance with <u>Section 9.1</u> of the Merger Agreement and to each Stockholder at its address set forth on Schedule 1 attached hereto (or at such other address for a party as shall be specified by like notice).
- 16. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- Assignability. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided, however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 18. No Waivers. Except as set forth in Section 23, no waivers of any breach of this Agreement extended by PLx or DPRX to Stockholder shall be construed as a waiver of any rights or remedies of PLx or DPRX, as applicable, with respect to any other stockholder of DPRX who has executed an agreement substantially in the form of this Agreement with respect to Shares held or subsequently held by such stockholder or with respect to any subsequent breach of the Stockholder or any other such stockholder of DPRX. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- 19. Applicable Law; Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (i) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (ii) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (i) of this Section 19, (iii) waives any objection to laying venue in any such action or proceeding in such courts, (iv) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (v) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 15 of this Agreement.
- 20. <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to trial by jury with respect to any action or proceeding related to or arising out of this Agreement, any document executed in connection herewith and the matters contemplated hereby and thereby.
- 21. <u>No Agreement Until Executed.</u> Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until
- (a) the DPRX Board of Directors has approved, for purposes of any applicable anti-takeover laws and regulations and any applicable provision of the DPRX certificate of incorporation, the transactions contemplated by the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.
- 22. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by facsimile or electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.

- 23. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 24. <u>Definition of Merger Agreement</u>. For purposes of this Agreement, the term "<u>Merger Agreement</u>" may include such agreement as amended or modified as long as such amendments or modifications (a) do not (i) change the form of consideration or (ii) change the Equity Exchange Ratio in a manner adverse to Stockholder, or (b) have been agreed to in writing by Stockholder.

25. Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page has Intentionally Been Left Blank]

STOCKHOLDER

By: /s/ Robert J. DeLuccia

Name: Robert J . DeLuccia

Title: Trustee

DIPEXIUM PHARMACEUTICALS, INC.

By: /s/ David P. Luci

Name: David P. Luci

Title: President and Chief Executive Officer

PLX PHARMA INC.

By: /s/ Natasha Giordano

Name: Natasha Giordano

Title: President and Chief Executive Officer

SCHEDULE 1

Name and Address of Stockholder	Shares	Options
ROBERT J. DELUCCIA 2016 ANNUITY TRUST, DATED SEPTEMBER 16, 2016, ROBERT J. DELUCCIA, TRUSTEE	500,000	0

among:

DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation;

PLX PHARMA INC., a Delaware corporation; and

the undersigned Stockholder

Dated as of December 22, 2016

TABLE OF CONTENTS

1.	Agreement to Vote Shares	1
2.	Expiration Date	2
3.	Additional Purchases	2
4.	Agreement to Retain Shares	2
5.	Representations and Warranties of Stockholder	3
6.	Irrevocable Proxy	3
7.	No Solicitation	4
8.	Release; No Legal Actions	4
9.	Other Remedies; Specific Performance	5
10.	Directors and Officers	5
11.	No Ownership Interest	6
12.	Termination	6
13.	Further Assurances	6
14.	Disclosure	6
15.	Notice	6
16.	Severability	6
17.	Assignability	7
18.	No Waivers	7
19.	Applicable Law; Jurisdiction	7
20.	Waiver of Jury Trial	7
21.	No Agreement Until Executed	7
22.	Entire Agreement; Counterparts; Exchanges by Facsimile	7
23.	Amendment	8
24.	Definition of Merger Agreement	8
25.	Construction	8
	i	

THIS VOTING AGREEMENT ("Agreement"), dated as of December 22, 2016, is made by and among Dipexium Pharmaceuticals, Inc., a Delaware corporation ("<u>DPRX</u>"), PLx Pharma Inc., a Delaware corporation ("<u>PLx</u>"), and the undersigned holder ("<u>Stockholder</u>") of shares of capital stock (the "<u>Shares</u>") of DPRX.

WHEREAS, DPRX, Dipexium Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of DPRX ("<u>AcquireCo</u>"), and PLx, have entered into an Agreement and Plan of Merger and Reorganization, dated of even date herewith (in the form in effect on the date hereof and attached hereto as <u>Exhibit A</u> or as amended pursuant to <u>Section 24</u> hereof, the "<u>Merger Agreement</u>"), providing for the merger of AcquireCo with and into PLx (the "<u>Merger</u>");

WHEREAS, Stockholder beneficially owns and has sole or shared voting power with respect to the number of Shares, and holds stock options or other rights to acquire the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of DPRX, AcquireCo and PLx to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, DPRX's, AcquireCo's and PLx's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the expenses incurred and to be incurred by them in connection therewith, Stockholder, DPRX and PLx agree as follows:

- 1. <u>Agreement to Vote Shares.</u> Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of DPRX or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of DPRX, with respect to approval of the Merger as contemplated by the Merger Agreement and adoption of the Merger Agreement or any DPRX Acquisition Proposal, Stockholder shall:
- (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat for purposes of calculating a quorum; and
- (b) vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that such Stockholder shall be entitled to so vote: (i) in favor of (A) the adoption of the Merger Agreement and the approval of the Merger, including without limitation the issuance of the DPRX Shares by virtue of the Merger as contemplated by the Merger Agreement, and (B) the DPRX Charter Amendment; (ii) against any action, proposal, transaction or agreement that, to the knowledge of Stockholder, would reasonably be expected to result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of DPRX under the Merger Agreement or that would reasonably be expected to result in any of the conditions to DPRX's, AcquireCo's or PLx's obligations under the Merger Agreement not being fulfilled; and (iii) against any DPRX Acquisition Proposal, or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement. The Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.

- 2. <u>Expiration Date</u>. As used in this Agreement, the term "<u>Expiration Date</u>" shall mean the earlier to occur of (a) the Merger Effective Time, (b) such date and time as the Merger Agreement shall be terminated pursuant to <u>Article VII</u> thereof or otherwise, or (c) upon mutual written agreement of the parties to terminate this Agreement. Upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; *provided, however*, such termination or expiration shall not relieve any party from liability for any willful breach of this Agreement or acts of bad faith prior to termination hereof.
- 3. <u>Additional Purchases</u>. Stockholder agrees that any shares of capital stock or other equity securities of DPRX that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any stock options or otherwise ("New Shares"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Agreement to Retain Shares. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below) on) any Shares, (b) deposit any Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens on) any Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing, Stockholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee and transferee shall sign a voting agreement in substantially the form hereof, (b) with respect to such Stockholder's DPRX Options which expire on or prior to the Expiration Date, transfers, sales, or other dispositions of Shares to DPRX as payment for the (i) exercise price of such Stockholder's DPRX Options and (ii) taxes applicable to the exercise of such Stockholder's DPRX Options, (c) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an affiliated corporation, trust or other business entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (d) any transfer to another holder of the capital stock of PLx that has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (e) any transfer to a person if, as a condition precedent to the transfer, such person executes and delivers to PLx an agreement containing voting and transfer provisions with respect to the Shares so transferred that are substantially identical in all material respects to those set forth in this Agreement, and (f) as PLx may otherwise agree in writing in its sole discretion.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to DPRX and PLx as follows:
- (a) Stockholder has the full power and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder;
- (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of PLx and DPRX, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of law or a court of equity and by bankruptcy, insolvency and similar laws affecting creditors' rights and remedies generally;
- (c) except as set forth on <u>Schedule 1</u>, Stockholder beneficially owns the number of Shares indicated opposite such Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("<u>Liens</u>"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares and none of the Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares, except as contemplated by this Agreement;
- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his or her obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares pursuant to, any agreement, instrument, note, bond, mortgage, contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other entity, any bylaw or other organizational document of Stockholder; and
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any governmental or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his or her obligations under this Agreement in any material respect.
- 6. <u>Irrevocable Proxy.</u> Subject to the penultimate sentence of this <u>Section 6</u>, by execution of this Agreement, Stockholder does hereby appoint PLx with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of the undersigned's rights with respect to the Shares, to vote, or give consent with respect to, each of such Shares solely with respect to the matters set forth in <u>Section 1</u> hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date and hereby revokes any proxy previously granted by Stockholder with respect to the Shares. Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date. The Stockholder hereby revokes any proxies previously granted and represents that none of such previously-granted proxies are irrevocable.

7. No Solicitation. From and after the date hereof until the Expiration Date, Stockholder shall not (a) initiate, solicit, seek or knowingly encourage or support any inquiries, proposals or offers that constitute or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (b) engage or participate in, or knowingly facilitate, any discussions or negotiations regarding any inquiries, proposals or offers that constitute, or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (c) furnish to any Person other than PLx any non-public information that could reasonably be expected to be used for the purposes of formulating any DPRX Acquisition Proposal, (d) enter into any letter of intent, agreement in principle or other similar type of agreement relating to a DPRX Acquisition Proposal, or enter into any agreement or agreement in principle requiring DPRX to abandon, terminate or fail to consummate the transactions contemplated hereby, (e) initiate a stockholders' vote or action by consent of DPRX's stockholders with respect to a DPRX Acquisition Proposal, (f) except by reason of this Agreement, become a member of a "group" (as such term is defined in Section 13(d) of the Exchange Act) with respect to any voting securities of DPRX that takes any action in support of a DPRX Acquisition Proposal or (g) propose or agree to do any of the foregoing. In the event that Stockholder is a corporation, partnership, trust or other entity, it shall not permit any of its Subsidiaries or Affiliates to, undertake any of the actions contemplated by this Section 7.

8. Release; No Legal Actions.

The undersigned Stockholder acknowledges that the release of certain claims by stockholders of DPRX against PLx, DPRX, AcquireCo and their respective affiliates constitutes a material inducement for the completion of the transactions contemplated by the Merger Agreement and that PLx, DPRX and AcquireCo would not enter into the Merger Agreement without being released from such claims by the undersigned Stockholder. The undersigned Stockholder, and, to the extent within the undersigned's control, each of the undersigned's equity holders and each of their respective subsidiaries, affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns (each, a "Releasor"), hereby completely releases, acquits and forever discharges, to the fullest extent permitted by law, PLx, DPRX, AcquireCo, the Surviving Company and their respective affiliates and each of their respective current, former and future officers, directors, employees, agents, advisors, successors and assigns (each, a "Releasee"), from any and all losses, liabilities, suits, actions, debts or rights, whether fixed or contingent, known or unknown, matured or unmatured, arising out of, relating to, or in any manner connected with any facts, events or circumstances, or any actions taken, at or prior to the Merger Effective Time that any Releasor ever had or now has against the Releasees ("Released Matters"), excluding any rights of the Releasor under the Merger Agreement. Notwithstanding anything to the contrary in this Agreement, nothing herein shall release PLx or any of its Affiliates of obligations to the undersigned Stockholder with respect to (A) any employment or consulting agreement, (B) any other employment-related obligations of PLx or any of its Affiliates, (C) vested retirement benefits, (D) any rights that cannot be waived as a matter of law, (E) any indemnification obligations to the undersigned Stockholder under PLx's or any of its Affiliates' bylaws, certificate of incorporation, or other organizational documents, or under Delaware law or otherwise, or (F) any rights relating to the undersigned's relationship with PLx or any of its Affiliates (other than as a stockholder). The undersigned hereby waives the provisions of section 1542 of the California Civil Code, or any successor thereto, which currently states: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." Effective as of the Merger Effective Time, the undersigned Stockholder shall not, and, to the extent within the undersigned's control, shall not cause or permit its equity holders or any of their respective Subsidiaries, Affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns, to assert any claims against the Releasees in respect of any Released Matters. The undersigned Stockholder acknowledges that it would be difficult to fully compensate DPRX or any of its Affiliates (including the Surviving Company) for damages resulting from any breach by him/her/it of the provisions of this release. Accordingly, in the event of any actual or threatened breach of such provisions, DPRX and its Affiliates (including the Surviving Company) shall (in addition to any other remedies which it may have) be entitled to seek temporary and/or permanent injunctive relief to enforce such provisions and recover attorneys' fees and costs for same. The undersigned Stockholder further acknowledges that this release constitutes a material inducement to DPRX to complete the transactions contemplated by the Merger Agreement and DPRX will be relying on the enforceability of this release in completing such transactions contemplated by the Merger Agreement.

- (b) The Stockholder will not in his, her or its capacity as a stockholder of DPRX bring, commence, institute, maintain, prosecute or voluntarily aid any legal proceeding which (i) challenges the validity or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this agreement by the Stockholder, either alone or together with the other voting agreements and proxies to be delivered in connection with the execution of the Merger Agreement, or the approval of the Merger Agreement by the DPRX Board of Directors, constitutes a breach of any fiduciary duty of the DPRX Board of Directors or any member thereof.
- 9. Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at Law or in equity.
- Directors and Officers. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of DPRX and/or holder of options to purchase DPRX Shares and not in such Stockholder's capacity as a director, officer or employee of DPRX or any of its Subsidiaries or in such Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require Stockholder to attempt to) prohibit, limit, prevent, preclude or restrict a director and/or officer of DPRX in the exercise of his or her fiduciary duties consistent with the terms of the Merger Agreement as a director and/or officer of DPRX or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director and/or officer, trustee or fiduciary of any employee benefit plan or trust from taking any action in his or her capacity as such director, officer, trustee and/or fiduciary.

- 11. <u>No Ownership Interest</u>. Nothing contained in this Agreement shall be deemed to vest in PLx any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Stockholder, and PLx does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of DPRX or exercise any power or authority to direct Stockholder in the voting of any of the Shares, except as otherwise provided herein.
- 12. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, nothing set forth in this <u>Section 12</u> or elsewhere in this Agreement shall relieve either party hereto from any liability, or otherwise limit the liability of either party from any liability for any intentional breach of any obligation or other provision contained in this Agreement.
- 13. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as PLx or DPRX may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Merger Agreement.
- 14. <u>Disclosure</u>. Stockholder hereby agrees that DPRX and PLx may publish and disclose in the Joint Proxy Statement, any prospectus filed with any regulatory authority in connection with the Merger and any related documents filed with such regulatory authority and as otherwise required by Law, such Stockholder's identity and ownership of Shares and the nature of such Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to the Joint Proxy Statement or prospectus or in any other filing made by DPRX or PLx as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Merger.
- 15. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) or by facsimile transmission (providing confirmation of transmission) to PLx or DPRX, as the case may be, in accordance with <u>Section 9.1</u> of the Merger Agreement and to each Stockholder at its address set forth on Schedule 1 attached hereto (or at such other address for a party as shall be specified by like notice).
- 16. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- 17. <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided*, *however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 18. No Waivers. Except as set forth in Section 23, no waivers of any breach of this Agreement extended by PLx or DPRX to Stockholder shall be construed as a waiver of any rights or remedies of PLx or DPRX, as applicable, with respect to any other stockholder of DPRX who has executed an agreement substantially in the form of this Agreement with respect to Shares held or subsequently held by such stockholder or with respect to any subsequent breach of the Stockholder or any other such stockholder of DPRX. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- 19. Applicable Law; Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (i) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (ii) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (i) of this Section 19, (iii) waives any objection to laying venue in any such action or proceeding in such courts, (iv) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (v) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 15 of this Agreement.
- 20. <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to trial by jury with respect to any action or proceeding related to or arising out of this Agreement, any document executed in connection herewith and the matters contemplated hereby and thereby.
- 21. <u>No Agreement Until Executed.</u> Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until
- (a) the DPRX Board of Directors has approved, for purposes of any applicable anti-takeover laws and regulations and any applicable provision of the DPRX certificate of incorporation, the transactions contemplated by the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.
- 22. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by facsimile or electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.

- 23. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 24. <u>Definition of Merger Agreement</u>. For purposes of this Agreement, the term "<u>Merger Agreement</u>" may include such agreement as amended or modified as long as such amendments or modifications (a) do not (i) change the form of consideration or (ii) change the Equity Exchange Ratio in a manner adverse to Stockholder, or (b) have been agreed to in writing by Stockholder.

25. Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page has Intentionally Been Left Blank]

EXECUTED	as	of the	date	first	ahove	writter
	45	OI IIIC	uaic	11151	anuve	willici

STOCKHOLDER:

/s/ Robert J. DeLuccia

Robert J. DeLuccia

DIPEXIUM PHARMACEUTICALS, INC.

By: /s/ David P. Luci

Name: David P. Luci

Title: President and Chief Executive Officer

PLX PHARMA INC.

By: /s/ Natasha Giordano

Name: Natasha Giordano

Title: President and Chief Executive Officer

SCHEDULE 1

Name and Address

of Stockholder	Shares	Options
Robert J. DeLuccia	1,336,909	368,826

among:

DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation;

PLX PHARMA INC., a Delaware corporation; and

the undersigned Stockholder

Dated as of December 22, 2016

TABLE OF CONTENTS

1.	Agreement to Vote Shares	1
2.	Expiration Date	2
3.	Additional Purchases	2
4.	Agreement to Retain Shares	2
5.	Representations and Warranties of Stockholder	3
6.	Irrevocable Proxy	3
7.	No Solicitation	4
8.	Release; No Legal Actions	4
9.	Other Remedies; Specific Performance	5
10.	Directors and Officers	5
11.	No Ownership Interest	6
12.	Termination	6
13.	Further Assurances	6
14.	Disclosure	6
15.	Notice	6
16.	Severability	6
17.	Assignability	7
18.	No Waivers	7
19.	Applicable Law; Jurisdiction	7
20.	Waiver of Jury Trial	7
21.	No Agreement Until Executed	7
22.	Entire Agreement; Counterparts; Exchanges by Facsimile	7
23.	Amendment	8
24.	Definition of Merger Agreement	8
25.	Construction	8
	i	

THIS VOTING AGREEMENT ("Agreement"), dated as of December 22, 2016, is made by and among Dipexium Pharmaceuticals, Inc., a Delaware corporation ("<u>DPRX</u>"), PLx Pharma Inc., a Delaware corporation ("<u>PLx</u>"), and the undersigned holder ("<u>Stockholder</u>") of shares of capital stock (the "<u>Shares</u>") of DPRX.

WHEREAS, DPRX, Dipexium Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of DPRX ("<u>AcquireCo</u>"), and PLx, have entered into an Agreement and Plan of Merger and Reorganization, dated of even date herewith (in the form in effect on the date hereof and attached hereto as <u>Exhibit A</u> or as amended pursuant to <u>Section 24</u> hereof, the "<u>Merger Agreement</u>"), providing for the merger of AcquireCo with and into PLx (the "<u>Merger</u>");

WHEREAS, Stockholder beneficially owns and has sole or shared voting power with respect to the number of Shares, and holds stock options or other rights to acquire the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of DPRX, AcquireCo and PLx to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, DPRX's, AcquireCo's and PLx's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the expenses incurred and to be incurred by them in connection therewith, Stockholder, DPRX and PLx agree as follows:

- 1. <u>Agreement to Vote Shares</u>. Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of DPRX or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of DPRX, with respect to approval of the Merger as contemplated by the Merger Agreement and adoption of the Merger Agreement or any DPRX Acquisition Proposal, Stockholder shall:
- (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat for purposes of calculating a quorum; and
- (b) vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that such Stockholder shall be entitled to so vote: (i) in favor of (A) the adoption of the Merger Agreement and the approval of the Merger, including without limitation the issuance of the DPRX Shares by virtue of the Merger as contemplated by the Merger Agreement, and (B) the DPRX Charter Amendment; (ii) against any action, proposal, transaction or agreement that, to the knowledge of Stockholder, would reasonably be expected to result in any material respect of any covenant, representation or warranty or any other obligation or agreement of DPRX under the Merger Agreement or that would reasonably be expected to result in any of the conditions to DPRX's, AcquireCo's or PLx's obligations under the

Merger Agreement not being fulfilled; and (iii) against any DPRX Acquisition Proposal, or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement. The Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.

- 2. <u>Expiration Date</u>. As used in this Agreement, the term "<u>Expiration Date</u>" shall mean the earlier to occur of (a) the Merger Effective Time, (b) such date and time as the Merger Agreement shall be terminated pursuant to <u>Article VII</u> thereof or otherwise, or (c) upon mutual written agreement of the parties to terminate this Agreement. Upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; *provided, however*, such termination or expiration shall not relieve any party from liability for any willful breach of this Agreement or acts of bad faith prior to termination hereof.
- 3. <u>Additional Purchases</u>. Stockholder agrees that any shares of capital stock or other equity securities of DPRX that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any stock options or otherwise ("New Shares"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Agreement to Retain Shares. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below) on) any Shares, (b) deposit any Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens on) any Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing, Stockholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee and transferee shall sign a voting agreement in substantially the form hereof, (b) with respect to such Stockholder's DPRX Options which expire on or prior to the Expiration Date, transfers, sales, or other dispositions of Shares to DPRX as payment for the (i) exercise price of such Stockholder's DPRX Options and (ii) taxes applicable to the exercise of such Stockholder's DPRX Options, (c) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an affiliated corporation, trust or other business entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (d) any transfer to another holder of the capital stock of PLx that has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (e) any transfer to a person if, as a condition precedent to the transfer, such person executes and delivers to PLx an agreement containing voting and transfer provisions with respect to the Shares so transferred that are substantially identical in all material respects to those set forth in this Agreement, and (f) as PLx may otherwise agree in writing in its sole discretion.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to DPRX and PLx as follows:
- (a) Stockholder has the full power and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder;
- (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of PLx and DPRX, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of law or a court of equity and by bankruptcy, insolvency and similar laws affecting creditors' rights and remedies generally;
- (c) except as set forth on <u>Schedule 1</u>, Stockholder beneficially owns the number of Shares indicated opposite such Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("<u>Liens</u>"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares and none of the Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares, except as contemplated by this Agreement;
- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his or her obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares pursuant to, any agreement, instrument, note, bond, mortgage, contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other entity, any bylaw or other organizational document of Stockholder; and
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any governmental or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his or her obligations under this Agreement in any material respect.
- 6. <u>Irrevocable Proxy.</u> Subject to the penultimate sentence of this <u>Section 6</u>, by execution of this Agreement, Stockholder does hereby appoint PLx with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of the undersigned's rights with respect to the Shares, to vote, or give consent with respect to, each of such Shares solely with respect to the matters set forth in <u>Section 1</u> hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date and hereby revokes any proxy previously granted by Stockholder with respect to the Shares. Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date. The Stockholder hereby revokes any proxies previously granted and represents that none of such previously-granted proxies are irrevocable.

7. No Solicitation. From and after the date hereof until the Expiration Date, Stockholder shall not (a) initiate, solicit, seek or knowingly encourage or support any inquiries, proposals or offers that constitute or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (b) engage or participate in, or knowingly facilitate, any discussions or negotiations regarding any inquiries, proposals or offers that constitute, or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (c) furnish to any Person other than PLx any non-public information that could reasonably be expected to be used for the purposes of formulating any DPRX Acquisition Proposal, (d) enter into any letter of intent, agreement in principle or other similar type of agreement relating to a DPRX Acquisition Proposal, or enter into any agreement or agreement in principle requiring DPRX to abandon, terminate or fail to consummate the transactions contemplated hereby, (e) initiate a stockholders' vote or action by consent of DPRX's stockholders with respect to a DPRX Acquisition Proposal, (f) except by reason of this Agreement, become a member of a "group" (as such term is defined in Section 13(d) of the Exchange Act) with respect to any voting securities of DPRX that takes any action in support of a DPRX Acquisition Proposal or (g) propose or agree to do any of the foregoing. In the event that Stockholder is a corporation, partnership, trust or other entity, it shall not permit any of its Subsidiaries or Affiliates to, undertake any of the actions contemplated by this Section 7.

8. <u>Release; No Legal Actions</u>.

The undersigned Stockholder acknowledges that the release of certain claims by stockholders of DPRX against PLx, DPRX, AcquireCo and their respective affiliates constitutes a material inducement for the completion of the transactions contemplated by the Merger Agreement and that PLx, DPRX and AcquireCo would not enter into the Merger Agreement without being released from such claims by the undersigned Stockholder. The undersigned Stockholder, and, to the extent within the undersigned's control, each of the undersigned's equity holders and each of their respective subsidiaries, affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns (each, a "Releasor"), hereby completely releases, acquits and forever discharges, to the fullest extent permitted by law, PLx, DPRX, AcquireCo, the Surviving Company and their respective affiliates and each of their respective current, former and future officers, directors, employees, agents, advisors, successors and assigns (each, a "Releasee"), from any and all losses, liabilities, suits, actions, debts or rights, whether fixed or contingent, known or unknown, matured or unmatured, arising out of, relating to, or in any manner connected with any facts, events or circumstances, or any actions taken, at or prior to the Merger Effective Time that any Releasor ever had or now has against the Releasees ("Released Matters"), excluding any rights of the Releasor under the Merger Agreement. Notwithstanding anything to the contrary in this Agreement, nothing herein shall release PLx or any of its Affiliates of obligations to the undersigned Stockholder with respect to (A) any employment or consulting agreement, (B) any other employment-related obligations of PLx or any of its Affiliates, (C) vested retirement benefits, (D) any rights that cannot be waived as a matter of law, (E) any indemnification obligations to the undersigned Stockholder under PLx's or any of its Affiliates' bylaws, certificate of incorporation, or other organizational documents, or under Delaware law or otherwise, or (F) any rights relating to the undersigned's relationship with PLx or any of its Affiliates (other than as a stockholder). The undersigned hereby waives the provisions of section 1542 of the California Civil Code, or any successor thereto, which currently states: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." Effective as of the Merger Effective Time, the undersigned Stockholder shall not, and, to the extent within the undersigned's control, shall not cause or permit its equity holders or any of their respective Subsidiaries, Affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns, to assert any claims against the Releasees in respect of any Released Matters. The undersigned Stockholder acknowledges that it would be difficult to fully compensate DPRX or any of its Affiliates (including the Surviving Company) for damages resulting from any breach by him/her/it of the provisions of this release. Accordingly, in the event of any actual or threatened breach of such provisions, DPRX and its Affiliates (including the Surviving Company) shall (in addition to any other remedies which it may have) be entitled to seek temporary and/or permanent injunctive relief to enforce such provisions and recover attorneys' fees and costs for same. The undersigned Stockholder further acknowledges that this release constitutes a material inducement to DPRX to complete the transactions contemplated by the Merger Agreement and DPRX will be relying on the enforceability of this release in completing such transactions contemplated by the Merger Agreement.

- (b) The Stockholder will not in his, her or its capacity as a stockholder of DPRX bring, commence, institute, maintain, prosecute or voluntarily aid any legal proceeding which (i) challenges the validity or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this agreement by the Stockholder, either alone or together with the other voting agreements and proxies to be delivered in connection with the execution of the Merger Agreement, or the approval of the Merger Agreement by the DPRX Board of Directors, constitutes a breach of any fiduciary duty of the DPRX Board of Directors or any member thereof.
- 9. Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at Law or in equity.
- 10. <u>Directors and Officers</u>. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of DPRX and/or holder of options to purchase DPRX Shares and not in such Stockholder's capacity as a director, officer or employee of DPRX or any of its Subsidiaries or in such Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require Stockholder to attempt to) prohibit, limit, prevent, preclude or restrict a director and/or officer of DPRX in the exercise of his or her fiduciary duties consistent with the terms of the Merger Agreement as a director and/or officer of DPRX or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director and/or officer, trustee or fiduciary of any employee benefit plan or trust from taking any action in his or her capacity as such director, officer, trustee and/or fiduciary.

- 11. No Ownership Interest. Nothing contained in this Agreement shall be deemed to vest in PLx any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Stockholder, and PLx does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of DPRX or exercise any power or authority to direct Stockholder in the voting of any of the Shares, except as otherwise provided herein.
- 12. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, nothing set forth in this <u>Section 12</u> or elsewhere in this Agreement shall relieve either party hereto from any liability, or otherwise limit the liability of either party from any liability for any intentional breach of any obligation or other provision contained in this Agreement.
- 13. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as PLx or DPRX may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Merger Agreement.
- 14. <u>Disclosure</u>. Stockholder hereby agrees that DPRX and PLx may publish and disclose in the Joint Proxy Statement, any prospectus filed with any regulatory authority in connection with the Merger and any related documents filed with such regulatory authority and as otherwise required by Law, such Stockholder's identity and ownership of Shares and the nature of such Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to the Joint Proxy Statement or prospectus or in any other filing made by DPRX or PLx as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Merger.
- 15. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) or by facsimile transmission (providing confirmation of transmission) to PLx or DPRX, as the case may be, in accordance with <u>Section 9.1</u> of the Merger Agreement and to each Stockholder at its address set forth on Schedule 1 attached hereto (or at such other address for a party as shall be specified by like notice).
- 16. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- Assignability. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided, however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 18. No Waivers. Except as set forth in Section 23, no waivers of any breach of this Agreement extended by PLx or DPRX to Stockholder shall be construed as a waiver of any rights or remedies of PLx or DPRX, as applicable, with respect to any other stockholder of DPRX who has executed an agreement substantially in the form of this Agreement with respect to Shares held or subsequently held by such stockholder or with respect to any subsequent breach of the Stockholder or any other such stockholder of DPRX. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- 19. Applicable Law; Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (i) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (ii) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (i) of this Section 19, (iii) waives any objection to laying venue in any such action or proceeding in such courts, (iv) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (v) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 15 of this Agreement.
- 20. <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to trial by jury with respect to any action or proceeding related to or arising out of this Agreement, any document executed in connection herewith and the matters contemplated hereby and thereby.
- 21. <u>No Agreement Until Executed.</u> Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until
- (a) the DPRX Board of Directors has approved, for purposes of any applicable anti-takeover laws and regulations and any applicable provision of the DPRX certificate of incorporation, the transactions contemplated by the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.
- 22. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by facsimile or electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.

- 23. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 24. <u>Definition of Merger Agreement</u>. For purposes of this Agreement, the term "<u>Merger Agreement</u>" may include such agreement as amended or modified as long as such amendments or modifications (a) do not (i) change the form of consideration or (ii) change the Equity Exchange Ratio in a manner adverse to Stockholder, or (b) have been agreed to in writing by Stockholder.

25. Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page has Intentionally Been Left Blank]

EXECUTED	as of the	date first	ahove	written

STOCKHOLDER

By: /s/ David P. Luci

Name: David P. Luci

Title: Trustee

DIPEXIUM PHARMACEUTICALS, INC.

By: /s/ David P. Luci

Name: David P. Luci

Title: President and Chief Executive Officer

PLX PHARMA INC.

By: /s/ Natasha Giordano

Name: Natasha Giordano

Title: President and Chief Executive Officer

SCHEDULE 1

Name and Address of Stockholder	Shares	Options
DAVID P. LUCI 2016 ANNUITY TRUST	300,000	0

among:

DIPEXIUM PHARMACEUTICALS, INC., a Delaware corporation;

PLX PHARMA INC., a Delaware corporation; and

the undersigned Stockholder

Dated as of December 22, 2016

TABLE OF CONTENTS

1.	Agreement to Vote Shares	1
2.	Expiration Date	2
3.	Additional Purchases	2
4.	Agreement to Retain Shares	2
5.	Representations and Warranties of Stockholder	3
6.	Irrevocable Proxy	3
7.	No Solicitation	4
8.	Release; No Legal Actions	4
9.	Other Remedies; Specific Performance	5
10.	Directors and Officers	5
11.	No Ownership Interest	6
12.	Termination	6
13.	Further Assurances	6
14.	Disclosure	6
15.	Notice	6
16.	Severability	6
17.	Assignability	7
18.	No Waivers	7
19.	Applicable Law; Jurisdiction	7
20.	Waiver of Jury Trial	7
21.	No Agreement Until Executed	7
22.	Entire Agreement; Counterparts; Exchanges by Facsimile	7
23.	Amendment	8
24.	Definition of Merger Agreement	8
25.	Construction	8
	i	

THIS VOTING AGREEMENT ("Agreement"), dated as of December 22, 2016, is made by and among Dipexium Pharmaceuticals, Inc., a Delaware corporation ("<u>DPRX</u>"), PLx Pharma Inc., a Delaware corporation ("<u>PLx</u>"), and the undersigned holder ("<u>Stockholder</u>") of shares of capital stock (the "<u>Shares</u>") of DPRX.

WHEREAS, DPRX, Dipexium Acquisition Corp., a Delaware corporation and a wholly owned subsidiary of DPRX ("<u>AcquireCo</u>"), and PLx, have entered into an Agreement and Plan of Merger and Reorganization, dated of even date herewith (in the form in effect on the date hereof and attached hereto as <u>Exhibit A</u> or as amended pursuant to <u>Section 24</u> hereof, the "<u>Merger Agreement</u>"), providing for the merger of AcquireCo with and into PLx (the "<u>Merger</u>");

WHEREAS, Stockholder beneficially owns and has sole or shared voting power with respect to the number of Shares, and holds stock options or other rights to acquire the number of Shares indicated opposite Stockholder's name on <u>Schedule 1</u> attached hereto;

WHEREAS, as an inducement and a condition to the willingness of DPRX, AcquireCo and PLx to enter into the Merger Agreement, and in consideration of the substantial expenses incurred and to be incurred by them in connection therewith, Stockholder has agreed to enter into and perform this Agreement; and

WHEREAS, all capitalized terms used in this Agreement without definition herein shall have the meanings ascribed to them in the Merger Agreement.

NOW, THEREFORE, in consideration of, and as a condition to, DPRX's, AcquireCo's and PLx's entering into the Merger Agreement and proceeding with the transactions contemplated thereby, and in consideration of the expenses incurred and to be incurred by them in connection therewith, Stockholder, DPRX and PLx agree as follows:

- 1. <u>Agreement to Vote Shares</u>. Stockholder agrees that, prior to the Expiration Date (as defined in <u>Section 2</u> below), at any meeting of the stockholders of DPRX or any adjournment or postponement thereof, or in connection with any written consent of the stockholders of DPRX, with respect to approval of the Merger as contemplated by the Merger Agreement and adoption of the Merger Agreement or any DPRX Acquisition Proposal, Stockholder shall:
- (a) appear at such meeting or otherwise cause the Shares and any New Shares (as defined in <u>Section 3</u> below) to be counted as present thereat for purposes of calculating a quorum; and
- (b) vote (or cause to be voted), or deliver a written consent (or cause a written consent to be delivered) covering all of the Shares and any New Shares that such Stockholder shall be entitled to so vote: (i) in favor of (A) the adoption of the Merger Agreement and the approval of the Merger, including without limitation the issuance of the DPRX Shares by virtue of the Merger as contemplated by the Merger Agreement, and (B) the DPRX Charter Amendment; (ii) against any action, proposal, transaction or agreement that, to the knowledge of Stockholder, would reasonably be expected to result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of DPRX under the Merger Agreement or that would reasonably be expected to result in any of the conditions to DPRX's, AcquireCo's or PLx's obligations under the Merger Agreement not being fulfilled; and (iii) against any DPRX Acquisition Proposal, or any agreement, transaction or other matter that is intended to, or would reasonably be expected to, impede, interfere with, delay, postpone, discourage or materially and adversely affect the consummation of the Merger and all other transactions contemplated by the Merger Agreement. The Stockholder shall not take or commit or agree to take any action inconsistent with the foregoing.

- 2. <u>Expiration Date</u>. As used in this Agreement, the term "<u>Expiration Date</u>" shall mean the earlier to occur of (a) the Merger Effective Time, (b) such date and time as the Merger Agreement shall be terminated pursuant to <u>Article VII</u> thereof or otherwise, or (c) upon mutual written agreement of the parties to terminate this Agreement. Upon termination or expiration of this Agreement, no party shall have any further obligations or liabilities under this Agreement; *provided, however*, such termination or expiration shall not relieve any party from liability for any willful breach of this Agreement or acts of bad faith prior to termination hereof.
- 3. <u>Additional Purchases</u>. Stockholder agrees that any shares of capital stock or other equity securities of DPRX that Stockholder purchases or with respect to which Stockholder otherwise acquires sole or shared voting power after the execution of this Agreement and prior to the Expiration Date, whether by the exercise of any stock options or otherwise ("New Shares"), shall be subject to the terms and conditions of this Agreement to the same extent as if they constituted the Shares.
- Agreement to Retain Shares. From and after the date hereof until the Expiration Date, Stockholder shall not, directly or indirectly, (a) sell, assign, transfer, tender, offer, exchange, assign, pledge or otherwise dispose of (including, without limitation, by the creation of any Liens (as defined in Section 5(c) below) on) any Shares, (b) deposit any Shares into a voting trust or enter into a voting agreement or similar arrangement with respect to such Shares or grant any proxy or power of attorney with respect thereto (other than this Agreement), (c) enter into any contract, option, commitment or other arrangement or understanding with respect to the direct or indirect sale, transfer, assignment or other disposition of (including, without limitation, by the creation of any Liens on) any Shares, or (d) take any action that would make any representation or warranty of Stockholder contained herein untrue or incorrect or have the effect of preventing or disabling Stockholder from performing Stockholder's obligations under this Agreement. Notwithstanding the foregoing, Stockholder may make (a) transfers by will or by operation of law or other transfers for estate-planning purposes, in which case this Agreement shall bind the transferee and transferee shall sign a voting agreement in substantially the form hereof, (b) with respect to such Stockholder's DPRX Options which expire on or prior to the Expiration Date, transfers, sales, or other dispositions of Shares to DPRX as payment for the (i) exercise price of such Stockholder's DPRX Options and (ii) taxes applicable to the exercise of such Stockholder's DPRX Options, (c) if Stockholder is a partnership or limited liability company, a transfer to one or more partners or members of Stockholder or to an affiliated corporation, trust or other business entity under common control with Stockholder, or if Stockholder is a trust, a transfer to a beneficiary, provided that in each such case the applicable transferee has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (d) any transfer to another holder of the capital stock of PLx that has signed a voting agreement in substantially the form hereof relating to the transferred Shares, (e) any transfer to a person if, as a condition precedent to the transfer, such person executes and delivers to PLx an agreement containing voting and transfer provisions with respect to the Shares so transferred that are substantially identical in all material respects to those set forth in this Agreement, and (f) as PLx may otherwise agree in writing in its sole discretion.

- 5. <u>Representations and Warranties of Stockholder</u>. Stockholder hereby represents and warrants to DPRX and PLx as follows:
- (a) Stockholder has the full power and authority to execute and deliver this Agreement and to perform Stockholder's obligations hereunder;
- (b) this Agreement has been duly executed and delivered by or on behalf of Stockholder and, assuming this Agreement constitutes a valid and binding agreement of PLx and DPRX, constitutes a valid and binding agreement with respect to Stockholder, enforceable against Stockholder in accordance with its terms, except as enforcement may be limited by general principles of equity whether applied in a court of law or a court of equity and by bankruptcy, insolvency and similar laws affecting creditors' rights and remedies generally;
- (c) except as set forth on <u>Schedule 1</u>, Stockholder beneficially owns the number of Shares indicated opposite such Stockholder's name on <u>Schedule 1</u>, and will own any New Shares, free and clear of any liens, claims, charges or other encumbrances or restrictions of any kind whatsoever ("<u>Liens</u>"), and has sole or shared, and otherwise unrestricted, voting power with respect to such Shares and none of the Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Shares, except as contemplated by this Agreement;
- (d) the execution and delivery of this Agreement by Stockholder does not, and the performance by Stockholder of his or her obligations hereunder and the compliance by Stockholder with any provisions hereof will not, violate or conflict with, result in a material breach of or constitute a default (or an event that with notice or lapse of time or both would become a material default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of any Liens on any Shares pursuant to, any agreement, instrument, note, bond, mortgage, contract, lease, license, permit or other obligation or any order, arbitration award, judgment or decree to which Stockholder is a party or by which Stockholder is bound, or any law, statute, rule or regulation to which Stockholder is subject or, in the event that Stockholder is a corporation, partnership, trust or other entity, any bylaw or other organizational document of Stockholder; and
- (e) the execution and delivery of this Agreement by Stockholder does not, and the performance of this Agreement by Stockholder does not and will not, require any consent, approval, authorization or permit of, or filing with or notification to, any governmental or regulatory authority by Stockholder except for applicable requirements, if any, of the Exchange Act, and except where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not prevent or delay the performance by Stockholder of his or her obligations under this Agreement in any material respect.
- 6. <u>Irrevocable Proxy</u>. Subject to the penultimate sentence of this <u>Section 6</u>, by execution of this Agreement, Stockholder does hereby appoint PLx with full power of substitution and resubstitution, as Stockholder's true and lawful attorney and irrevocable proxy, to the fullest extent of the undersigned's rights with respect to the Shares, to vote, or give consent with respect to, each of such Shares solely with respect to the matters set forth in <u>Section 1</u> hereof. Stockholder intends this proxy to be irrevocable and coupled with an interest hereunder until the Expiration Date and hereby revokes any proxy previously granted by Stockholder with respect to the Shares. Stockholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. Notwithstanding anything contained herein to the contrary, this irrevocable proxy shall automatically terminate upon the Expiration Date. The Stockholder hereby revokes any proxies previously granted and represents that none of such previously-granted proxies are irrevocable.

7. No Solicitation. From and after the date hereof until the Expiration Date, Stockholder shall not (a) initiate, solicit, seek or knowingly encourage or support any inquiries, proposals or offers that constitute or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (b) engage or participate in, or knowingly facilitate, any discussions or negotiations regarding any inquiries, proposals or offers that constitute, or may reasonably be expected to lead to, a DPRX Acquisition Proposal, (c) furnish to any Person other than PLx any non-public information that could reasonably be expected to be used for the purposes of formulating any DPRX Acquisition Proposal, (d) enter into any letter of intent, agreement in principle or other similar type of agreement relating to a DPRX Acquisition Proposal, or enter into any agreement or agreement in principle requiring DPRX to abandon, terminate or fail to consummate the transactions contemplated hereby, (e) initiate a stockholders' vote or action by consent of DPRX's stockholders with respect to a DPRX Acquisition Proposal, (f) except by reason of this Agreement, become a member of a "group" (as such term is defined in Section 13(d) of the Exchange Act) with respect to any voting securities of DPRX that takes any action in support of a DPRX Acquisition Proposal or (g) propose or agree to do any of the foregoing. In the event that Stockholder is a corporation, partnership, trust or other entity, it shall not permit any of its Subsidiaries or Affiliates to, undertake any of the actions contemplated by this Section 7.

8. Release; No Legal Actions.

The undersigned Stockholder acknowledges that the release of certain claims by stockholders of DPRX against PLx, DPRX, AcquireCo and their respective affiliates constitutes a material inducement for the completion of the transactions contemplated by the Merger Agreement and that PLx, DPRX and AcquireCo would not enter into the Merger Agreement without being released from such claims by the undersigned Stockholder. The undersigned Stockholder, and, to the extent within the undersigned's control, each of the undersigned's equity holders and each of their respective subsidiaries, affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns (each, a "Releasor"), hereby completely releases, acquits and forever discharges, to the fullest extent permitted by law, PLx, DPRX, AcquireCo, the Surviving Company and their respective affiliates and each of their respective current, former and future officers, directors, employees, agents, advisors, successors and assigns (each, a "Releasee"), from any and all losses, liabilities, suits, actions, debts or rights, whether fixed or contingent, known or unknown, matured or unmatured, arising out of, relating to, or in any manner connected with any facts, events or circumstances, or any actions taken, at or prior to the Merger Effective Time that any Releasor ever had or now has against the Releasees ("Released Matters"), excluding any rights of the Releasor under the Merger Agreement. Notwithstanding anything to the contrary in this Agreement, nothing herein shall release PLx or any of its Affiliates of obligations to the undersigned Stockholder with respect to (A) any employment or consulting agreement, (B) any other employment-related obligations of PLx or any of its Affiliates, (C) vested retirement benefits, (D) any rights that cannot be waived as a matter of law, (E) any indemnification obligations to the undersigned Stockholder under PLx's or any of its Affiliates' bylaws, certificate of incorporation, or other organizational documents, or under Delaware law or otherwise, or (F) any rights relating to the undersigned's relationship with PLx or any of its Affiliates (other than as a stockholder). The undersigned hereby waives the provisions of section 1542 of the California Civil Code, or any successor thereto, which currently states: "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor." Effective as of the Merger Effective Time, the undersigned Stockholder shall not, and, to the extent within the undersigned's control, shall not cause or permit its equity holders or any of their respective Subsidiaries, Affiliates, employees, agents, advisors, heirs, legal representatives, successors and assigns, to assert any claims against the Releasees in respect of any Released Matters. The undersigned Stockholder acknowledges that it would be difficult to fully compensate DPRX or any of its Affiliates (including the Surviving Company) for damages resulting from any breach by him/her/it of the provisions of this release. Accordingly, in the event of any actual or threatened breach of such provisions, DPRX and its Affiliates (including the Surviving Company) shall (in addition to any other remedies which it may have) be entitled to seek temporary and/or permanent injunctive relief to enforce such provisions and recover attorneys' fees and costs for same. The undersigned Stockholder further acknowledges that this release constitutes a material inducement to DPRX to complete the transactions contemplated by the Merger Agreement and DPRX will be relying on the enforceability of this release in completing such transactions contemplated by the Merger Agreement.

- (b) The Stockholder will not in his, her or its capacity as a stockholder of DPRX bring, commence, institute, maintain, prosecute or voluntarily aid any legal proceeding which (i) challenges the validity or seeks to enjoin the operation of any provision of this Agreement or (ii) alleges that the execution and delivery of this agreement by the Stockholder, either alone or together with the other voting agreements and proxies to be delivered in connection with the execution of the Merger Agreement, or the approval of the Merger Agreement by the DPRX Board of Directors, constitutes a breach of any fiduciary duty of the DPRX Board of Directors or any member thereof.
- 9. Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with, and not exclusive of, any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at Law or in equity.
- 10. <u>Directors and Officers</u>. This Agreement shall apply to Stockholder solely in Stockholder's capacity as a stockholder of DPRX and/or holder of options to purchase DPRX Shares and not in such Stockholder's capacity as a director, officer or employee of DPRX or any of its Subsidiaries or in such Stockholder's capacity as a trustee or fiduciary of any employee benefit plan or trust. Notwithstanding any provision of this Agreement to the contrary, nothing in this Agreement shall (or require Stockholder to attempt to) prohibit, limit, prevent, preclude or restrict a director and/or officer of DPRX in the exercise of his or her fiduciary duties consistent with the terms of the Merger Agreement as a director and/or officer of DPRX or in his or her capacity as a trustee or fiduciary of any employee benefit plan or trust or prevent or be construed to create any obligation on the part of any director and/or officer of DPRX or any trustee or fiduciary of any employee benefit plan or trust from taking any action in his or her capacity as such director, officer, trustee and/or fiduciary.

- 11. No Ownership Interest. Nothing contained in this Agreement shall be deemed to vest in PLx any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Stockholder, and PLx does not have authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of DPRX or exercise any power or authority to direct Stockholder in the voting of any of the Shares, except as otherwise provided herein.
- 12. <u>Termination</u>. This Agreement shall terminate and shall have no further force or effect as of the Expiration Date. Notwithstanding the foregoing, nothing set forth in this <u>Section 12</u> or elsewhere in this Agreement shall relieve either party hereto from any liability, or otherwise limit the liability of either party from any liability for any intentional breach of any obligation or other provision contained in this Agreement.
- 13. <u>Further Assurances</u>. Stockholder shall, from time to time, execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as PLx or DPRX may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement and the Merger Agreement.
- 14. <u>Disclosure</u>. Stockholder hereby agrees that DPRX and PLx may publish and disclose in the Joint Proxy Statement, any prospectus filed with any regulatory authority in connection with the Merger and any related documents filed with such regulatory authority and as otherwise required by Law, such Stockholder's identity and ownership of Shares and the nature of such Stockholder's commitments, arrangements and understandings under this Agreement and may further file this Agreement as an exhibit to the Joint Proxy Statement or prospectus or in any other filing made by DPRX or PLx as required by Law or the terms of the Merger Agreement, including with the SEC or other regulatory authority, relating to the Merger.
- 15. <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) or by facsimile transmission (providing confirmation of transmission) to PLx or DPRX, as the case may be, in accordance with <u>Section 9.1</u> of the Merger Agreement and to each Stockholder at its address set forth on Schedule 1 attached hereto (or at such other address for a party as shall be specified by like notice).
- 16. Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit such term or provision, to delete specific words or phrases or to replace such term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be valid and enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term or provision.

- 17. <u>Assignability</u>. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and assigns; *provided*, *however*, that neither this Agreement nor any of a party's rights or obligations hereunder may be assigned or delegated by such party without the prior written consent of the other parties hereto, and any attempted assignment or delegation of this Agreement or any of such rights or obligations by such party without the other party's prior written consent shall be void and of no effect. Nothing in this Agreement, express or implied, is intended to or shall confer upon any Person (other than the parties hereto) any right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.
- 18. No Waivers. Except as set forth in Section 23, no waivers of any breach of this Agreement extended by PLx or DPRX to Stockholder shall be construed as a waiver of any rights or remedies of PLx or DPRX, as applicable, with respect to any other stockholder of DPRX who has executed an agreement substantially in the form of this Agreement with respect to Shares held or subsequently held by such stockholder or with respect to any subsequent breach of the Stockholder or any other such stockholder of DPRX. No waiver of any provisions hereof by any party shall be deemed a waiver of any other provisions hereof by any such party, nor shall any such waiver be deemed a continuing waiver of any provision hereof by such party.
- 19. Applicable Law; Jurisdiction. This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware, regardless of the Laws that might otherwise govern under applicable principles of conflicts of Laws. In any action or proceeding between any of the parties arising out of or relating to this Agreement, each of the parties: (i) irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware or to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware, (ii) agrees that all claims in respect of such action or proceeding shall be heard and determined exclusively in accordance with clause (i) of this Section 19, (iii) waives any objection to laying venue in any such action or proceeding in such courts, (iv) waives any objection that such courts are an inconvenient forum or do not have jurisdiction over any party, and (v) agrees that service of process upon such party in any such action or proceeding shall be effective if notice is given in accordance with Section 15 of this Agreement.
- 20. <u>Waiver of Jury Trial</u>. The parties hereto hereby waive any right to trial by jury with respect to any action or proceeding related to or arising out of this Agreement, any document executed in connection herewith and the matters contemplated hereby and thereby.
- 21. No Agreement Until Executed. Irrespective of negotiations among the parties or the exchanging of drafts of this Agreement, this Agreement shall not constitute or be deemed to evidence a contract, agreement, arrangement or understanding between the parties hereto unless and until (a) the DPRX Board of Directors has approved, for purposes of any applicable anti-takeover laws and regulations and any applicable provision of the DPRX certificate of incorporation, the transactions contemplated by the Merger Agreement, (b) the Merger Agreement is executed by all parties thereto, and (c) this Agreement is executed by all parties hereto.
- 22. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the parties with respect to the subject matter hereof and thereof. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all parties by facsimile or electronic transmission via ".pdf" shall be sufficient to bind the parties to the terms and conditions of this Agreement.

- 23. <u>Amendment</u>. This Agreement may not be amended, supplemented or modified, and no provisions hereof may be modified or waived, except by an instrument in writing signed on behalf of each party hereto.
- 24. <u>Definition of Merger Agreement</u>. For purposes of this Agreement, the term "<u>Merger Agreement</u>" may include such agreement as amended or modified as long as such amendments or modifications (a) do not (i) change the form of consideration or (ii) change the Equity Exchange Ratio in a manner adverse to Stockholder, or (b) have been agreed to in writing by Stockholder.

25. Construction.

- (a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.
- (b) The parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement.
- (c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."
- (d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement, respectively.
- (e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[Remainder of Page has Intentionally Been Left Blank]

STOCKHOLDER

By:	/s/ David P. Luci

Name: David P. Luci

DIPEXIUM PHARMACEUTICALS, INC.

By: /s/ David P. Luci

Name: David P. Luci

Title: President and Chief Executive Officer

PLX PHARMA INC.

By: /s/ Natasha Giordano

Name:Natasha Giordano

Title: President and Chief Executive Officer

SCHEDULE 1

	d Address kholder	Shares	Options
David P. Luci		1,539,709	368,826